

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Darmody, Merike; McGinnity, Frances; Russell, Helen

Research Report Children of migrants in Ireland: How are they faring?

Research Series, No. 134

Provided in Cooperation with: The Economic and Social Research Institute (ESRI), Dublin

Suggested Citation: Darmody, Merike; McGinnity, Frances; Russell, Helen (2022) : Children of migrants in Ireland: How are they faring?, Research Series, No. 134, The Economic and Social Research Institute (ESRI), Dublin, https://doi.org/10.26504/rs134

This Version is available at: https://hdl.handle.net/10419/268071

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



RESEARCH SERIES NUMBER 134 March 2022

CHILDREN OF MIGRANTS IN IRELAND: HOW ARE THEY FARING?

MERIKE DARMODY, FRANCES MCGINNITY AND HELEN RUSSELL





CHILDREN OF MIGRANTS IN IRELAND: HOW ARE THEY FARING?

Merike Darmody

Frances McGinnity

Helen Russell

March 2022

RESEARCH SERIES

NUMBER 134

Available to download from www.esri.ie © The Economic and Social Research Institute Whitaker Square, Sir John Rogerson's Quay, Dublin 2

https://doi.org/10.26504/rs134



This Open Access work is licensed under a Creative Commons Attribution 4.0 International License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly credited.

ABOUT THE ESRI

The mission of the Economic and Social Research Institute is to advance evidencebased policymaking that supports economic sustainability and social progress in Ireland. ESRI researchers apply the highest standards of academic excellence to challenges facing policymakers, focusing on 12 areas of critical importance to 21st century Ireland.

The Institute was founded in 1960 by a group of senior civil servants led by Dr T.K. Whitaker, who identified the need for independent and in-depth research analysis to provide a robust evidence base for policymaking in Ireland.

Since then, the Institute has remained committed to independent research and its work is free of any expressed ideology or political position. The Institute publishes all research reaching the appropriate academic standard, irrespective of its findings or who funds the research.

The quality of its research output is guaranteed by a rigorous peer review process. ESRI researchers are experts in their fields and are committed to producing work that meets the highest academic standards and practices.

The work of the Institute is disseminated widely in books, journal articles and reports. ESRI publications are available to download, free of charge, from its website. Additionally, ESRI staff communicate research findings at regular conferences and seminars.

The ESRI is a company limited by guarantee, answerable to its members and governed by a Council, comprising 14 members who represent a cross-section of ESRI members from academia, civil services, state agencies, businesses and civil society. The Institute receives an annual grant-in-aid from the Department of Public Expenditure and Reform to support the scientific and public interest elements of the Institute's activities; the grant accounted for an average of 30 per cent of the Institute's income over the lifetime of the last Research Strategy. The remaining funding comes from research programmes supported by government departments and agencies, public bodies and competitive research programmes.

Further information is available at www.esri.ie.

THE AUTHORS

Helen Russell is a Research Professor at the Economic and Social Research Institute (ESRI) and an Adjunct Professor at Trinity College Dublin (TCD). Fran McGinnity is an Associate Professor at the ESRI and an Adjunct Professor at TCD. Merike Darmody is a Research Officer at the ESRI and an Adjunct Assistant Professor at Trinity College Dublin (TCD).

ACKNOWLEDGEMENTS

This publication was funded by the Department of Children, Equality, Disability Integration and Youth (DCEDIY) through a joint research partnership with the ESRI. The views, opinions, findings and conclusions and/or recommendations expressed here are strictly those of the authors. They do not necessarily reflect the views of the DCEDIY, which takes no responsibility for any errors or omissions in, or for the accuracy of, the information contained in this publication. It is presented to inform and stimulate wider debate among the policy community and among academics and practitioners in the field.

The authors are grateful to members of the programme steering group, ESRI colleagues and an external reviewer for comments on an earlier draft of the report. We would also like to acknowledge the assistance of Aidan Sloyan while working on a research internship at the ESRI.

This research programme has received funding through WHAT WORKS. This is an initiative led by the DCEDIY, with funding from the Dormant Accounts Fund, to support a move towards evidence-informed prevention and early intervention services for children, young people and their families.

Growing Up in Ireland (GUI) is funded by the Department of Children, Equality, Disability, Integration and Youth (DCEDIY). It is managed by DCEDIY in association with the Central Statistics Office (CSO). Results in this report are based on analyses of data from Research Microdata Files provided by the Central Statistics Office (CSO). Neither the CSO nor DCEDIY take any responsibility for the views expressed or the outputs generated from these analyses.

This report has been accepted for publication by the Institute, which does not itself take institutional policy positions. All ESRI Research Series reports are peer reviewed prior to publication. The author(s) are solely responsible for the content and the views expressed.

TABLE OF CONTENTS

ABBREVIATIO	ONS		i	
EXECUTIVE S	UMMA	ARY	iii	
CHAPTER 1	Introduction			
	1.1 1.2 1.3 1.4	Migration policy and migration flows to Ireland Defining second-generation migrants Education policy context Report structure	2 6 9 13	
CHAPTER 2	Theoretical framework and literature review			
	2.1 2.2 2.3	Introduction Theoretical framework Literature review	15 15 19	
CHAPTER 3	Meth	nodology	29	
	3.1 3.2 3.3	About <i>Growing Up in Ireland</i> (Cohort '08) Outcome measures Measurement of migrant background	29 30 31	
	3.4 3.5	Measurement of explanatory variables	33	
CHAPTER 4	Profile of the second generation			
	4.1 4.2 4.3 4.4	Birthplace and linguistic and ethnic background of parents Early childcare/education and starting school School DEIS status and migrant concentration at age 9 Summary	37 40 43 45	
CHAPTER 5	English language and reading levels among second-generation migrant children		47	
	5.1 5.2 5.3 5.4 5.5 5.6 5.7	Introduction Child's first language English vocabulary and reading achievement at ages 5 and 9 Trajectories of skill development among migrant children (AGE 3 to 9) Models of vocabulary at age 5 Models of English reading at age 9 Summary	47 50 52 54 62	
CHAPTER 6	Child self-concept		71	
	6.16.26.36.4	Introduction Self-concept scores Participation in sport, social and cultural activities Models	71 72 74 76	
	6.5	Summary	81	

CHAPTER 7	Conclusion		83
	7.1	Summary of findings	83
	7.2	Policy implications	86
	7.3	Limitations of study and future research	
REFERENCES		93	
Appendix 1			

LIST OF FIGURES

Figure 1.1	Immigration, emigration and net migration, 1987–2020
Figure 1.2	Nationality of immigration flows, 2000–2019
Figure 3.1	Timeline of data collection for Cohort '08
Figure 4.1	Proportion of mothers born abroad who are Irish citizens
Figure 4.2	Ethnicity of mothers
Figure 4.3	Proportion of parents with English as native language40
Figure 4.4	Proportion of children in centre-based care at age 3 by maternal country of birth 41
Figure 4.5	Proportion of children who availed of ECCE FPSY by parental country of birth42
Figure 4.6	Proportion of children in school at age 5 by parental country of birth43
Figure 4.7	Proportion of children in a disadvantaged (DEIS) school by parental country of birth.44
Figure 4.8	Proportion of children in primary schools by concentration of migrants45
Figure 5.1	Proportion of children whose first language is English by age and parental origin and linguistic background
Figure 5.2	Mean naming vocabulary scores of 5 year olds by country of birth, ethnicity and language background of parents
Figure 5.3	Mean English reading scores of 9 year olds by country of origin, ethnicity and linguistic background of parents
Figure 5.4	Proportion of children in the lowest quintile of English vocabulary/reading by parental country of birth*
Figure 5.5	Proportion of children in the lowest quintile of vocabulary/reading by parental linguistic background
Figure 6.1	Mean total self-concept scores of 9 year olds by country of birth, ethnicity and language background of parents
Figure 6.2	Self-concept subscale scores of 9 year olds by parental country of birth74
Figure 6.3	Participation in sports at age 9 by migrant Background (parents born abroad)75
Figure 6.4	Mean number of social/cultural activities at age 9 by maternal country of birth and parents born abroad
Figure A.1	Proportion of migrant children in school at 9 years by parental country of birth 109

LIST OF TABLES

Table 4.1	Proportion of respondents with Irish and non-Irish parents
Table 4.2	Parental country of birth (%)
Table 5.1	OLS regression of vocabulary at age 5 and change over time (3-5 years)57
Table 5.2	Logistic regression models of lowest quintile of vocabulary at 5 years (bottom quintile) and change over time (age 3-5) (odds ratios)61
Table 5.3	OLS regression models of reading scores at age 9 and change over time between ages 5 and 963
Table 5.4	Logistic regression models of being in the lowest quintile in reading scores at 9 years (bottom quintile) and change over time (5-9) (odds ratios)67
Table 6.3	OLS regression of self-concept scores at age 979

ABBREVIATIONS

ADHD	Attention deficit hyperactivity disorder
BAME	Black, Asian and minority ethnic
BAS	British Ability Scales
СоВ	Country of birth
CSO	Central Statistics Office
DEIS	Delivering Equality of Opportunity in Schools
EAL	English as an additional language
ECCE	Early Childhood Care and Education Scheme
FPSY	Free pre-school year
GUI	Growing Up in Ireland
GAM	General Allocation Model
OECD	Organisation for Economic Co-operation and Development
OLS	Ordinary least squares
PISA	Programme for International Student Assessment
PDST	Professional Development Service for Teachers
SEN	Special educational need
SES	Socio-economic status
SDQ	Strengths and Difficulties Questionnaire
RMF	Randomised microfile

EXECUTIVE SUMMARY

There has been an increasing focus in policy development internationally and in Ireland on the importance of migrant integration. Consequently, a growing body of research has focused on the experiences of adult migrants and their children. Earlier studies on the experiences and outcomes of migrant-origin children in Ireland have examined children at age nine and older, many of whom moved to Ireland as children but were not born here. Much of this research has documented the social and academic challenges these children encounter. However, there has been relatively little research on the factors influencing child cognitive and socio-emotional development drawing on the experiences of second-generation children – children born in Ireland who have at least one migrant parent. This study looks at a variety of factors that shape these outcomes in early childhood, a crucial period for skill development, using rich data on children from Cohort '08 of the *Growing Up in Ireland* (GUI) study, who were born in 2008 and were nine years old in 2017.

The study investigates whether migrant-origin children differ from children with two Irish-born parents in terms of their English language development at three, five and nine years of age, and their self-concept at nine years. Language development is measured using tests of English vocabulary carried out at ages three and five, and of reading ability at nine years. Child self-concept is captured at nine years using a shortened version of the Piers-Harris Self-concept Scale, which the children completed themselves. English-language ability is important for children's school achievement and later integration into the labour market; it is also important for peer relationships. Self-concept is a key indicator of child socioemotional wellbeing.

English vocabulary outcomes and self-concept are analysed through a series of statistical models that firstly explore differences by parental country of birth, linguistic background and ethnic background. Previous research on child development highlights the importance of family, social and economic resources and child characteristics (such as gender and special educational needs (SEN), as well as institutional factors (for example, childcare and school characteristics). The models therefore also explore the role of these factors in understanding differences between migrant-origin children and children whose parents are born in Ireland.

The study draws out implications for policy at early childhood and primary school level for migrant-origin children, in order to inform future policy development, including actions in the successor to the *Migrant Integration Strategy 2017–2021* and the *National Action Plan Against Racism* currently being developed (Department of Justice and Equality, 2017; Anti-Racism Committee, 2021).

MAIN FINDINGS

- One-third of children in the '08 Cohort of GUI have at least one parent born abroad, though these children are diverse in terms of parental country of birth, linguistic background and ethnicity.
- In total, 14 per cent of children had one parent born abroad and one Irish-born parent. A further 19 per cent had both parents, or a lone parent, born outside the Republic of Ireland.
- Eleven per cent of children were in families where neither parent (or a lone parent) was a native English speaker.
- Six per cent of children had a mother from a Black, Asian or Other ethnic minority background. Ethnicity was self-reported using the Census categories.
- Children with a migrant background were somewhat less likely than those with Irish-born parents to have attended centre-based childcare at age three, prior to enrolment in the Early Childhood Care and Education Scheme (ECCE) scheme, but this varied widely by maternal country of birth.
- Children with both parents born abroad started school at an earlier age than those with one or both Irish parents and were much more likely to attend a Delivering Equality of Opportunity in Schools (DEIS) school: 28 per cent compared to 20 per cent of children when both parents are Irish-born and 17 per cent for those with one Irish-born parent and one parent born abroad. Note this cohort started school prior to the recent introduction of the Education (Admissions to Schools) Act, which is intended to provide greater equality in school access.
- Regarding English language achievement, children whose mothers come from the UK or other English-speaking countries (the US, Australia, Canada and New Zealand) do not differ from the children of Irish mothers in English vocabulary scores at age five. However, children with mothers from eastern Europe have significantly lower English vocabulary scores, on average, than students of Irish origin. Further analysis shows that having two parents that are non-native English speakers is the main distinction for vocabulary scores, regardless of parents' country of origin.
- At age three, 60 per cent of migrant-origin children were in the bottom quintile on the English vocabulary test (including those who had insufficient language skills to take part in the test). By age five, this had declined to 52 per cent and by age nine, 26 per cent were in the bottom quintile for reading. This suggests considerable progress in English-language development between ages three and nine for migrant-origin children living in Ireland.
- While the differential in English language achievement for second-generation migrant-origin children is much reduced at age nine, children of parents who are both non-native English speakers still have lower mean reading scores at age nine. Children who have one parent who is a native English speaker do not differ from children where both parents are native English speakers. A similar pattern is found for those who have one Irish-born parent and one foreignborn parent.
- This study has shown that, for the most part, second-generation migrant-origin children do not differ from Irish-origin children in terms of wellbeing at age

nine, as measured by their total self-concept score. This measure incorporates children's sense of their intellectual, educational, physical, emotional and social characteristics. Some migrant groups (for example, eastern Europeans) were found to have lower mean self-concept scores, which was mediated through child characteristics (including lower participation in sports and social/cultural activities). Lower mean self-concept scores are observed among children with parents from western Europe and the United States (US)/Canada/Australia compared to children with Irish parents, despite accounting for a range of other factors linked to child wellbeing.

- Having no native English-speaking parent in the household was associated with a lower self-concept score and this was mediated through fewer socioeconomic resources.
- Participation in team sports is associated with higher self-concept among children, regardless of their national, ethnic or cultural background.

IMPLICATIONS FOR POLICY AND PRACTICE

The study's findings indicate that parental region of origin and linguistic background can make a difference to migrant-origin children's cognitive and socioemotional development. The findings highlight the importance of facilitating access to quality early learning and care for migrant-origin children, as participation in early learning and care at age three is lower for them than for Irishorigin children, despite earlier research showing that children from non-Englishspeaking backgrounds benefit more from participation in centre-based care than their Irish-origin peers (McGinnity et al, 2015a). The findings also point to the need for support for migrant-origin students at primary level, including: enhanced support for DEIS Urban Band 1 schools; and provision of continuous Englishlanguage support, with ongoing assessment of the effectiveness of this language support in meeting children's needs. In addition, enhancing the English language skills of migrant parents would facilitate both their children's language development and learning, as well as migrant parents' liaison with schools. Previous research has highlighted the lack of a coordinated approach to English language provision for adult learners in Ireland, and the lack of awareness around provision that does exist (McGinnity et al., 2020b). Opportunities for migrant-origin children to engage in social activities, particularly engagement in team sports, would also be beneficial as our findings show this fosters a positive self-image among children.

How children view themselves and their linguistic proficiency in the language of instruction is likely to influence their experiences in the Irish education system and beyond, and to ultimately shape their integration into Irish society. Given recent changes to school admissions policy, as well as to access and affordability of early learning and care in Ireland, further research is needed to establish whether these patterns observed for this cohort born in 2008 persist for later cohorts of children.

CHAPTER 1

Introduction

The past 30 years have seen an increase in migration to Ireland, and this has led to a corresponding increase in research on migrant integration (McGinnity et al., 2012, 2020a; Gilmartin and Dagg, 2020; Kelly et al., 2016), as well as in policy (Department of Justice and Equality, 2017; 2019). The number of secondgeneration children in Ireland – those born here to at least one migrant parent – has also grown in recent years. The integration of the second migrant generation provides the real 'litmus test' for integration, as many of the skills and experience of the first generation will have been acquired abroad (OECD, 2018). Indeed, some have argued that 'the outcomes of the native-born offspring of immigrants are (thus) a better measurement for integration than the outcomes of the foreignborn' (OECD, 2018, p. 18).

The aim of this study is to consider the integration of the second generation by using a high-quality representative cohort study, *Growing Up in Ireland* (GUI), which allows us to compare migrant-origin children with Irish-origin children and to follow all children (and their families) to track their development from infancy to nine years of age.

The first wave of the GUI study was carried out in 2008, when the children were nine months old, following a rapid increase in immigration to Ireland. The second wave was undertaken in 2011, when the children were aged three and the third wave in 2013 when they were aged five. The children were also surveyed in 2017, at age nine. The report considers: the region of origin of parents; the parents' linguistic background and their ethnicity; and how the latter are associated with child English language development and self-concept.

This report seeks to answer the following key questions:

- Are differences in English language development evident in early childhood?
- Can differences between second-generation migrant-origin and native Irish children can be explained by differences in socio-economic background?
- Are differences between second-generation migrant-origin groups and children with Irish-born parents stable over time or do they increase or decrease as the child develops?

The study takes a longitudinal approach, exploring English language outcomes of second-generation migrant-origin children at three, five and nine years of age. Experiences of social integration (participation in sport and cultural activities) and self-concept are examined at age nine, based on the children's own reports. Under

this approach, the report combines a focus on English language outcomes as well as the wellbeing and self-concept of migrant children.

This chapter provides context for the report in terms of: migration flows to Ireland and how we define second-generation migrant-origin children; and policy concerning the care and education of children at preschool and primary school level, both in general and with specific regard to the needs of migrant-origin children.

1.1 MIGRATION POLICY AND MIGRATION FLOWS TO IRELAND

The landscape of migration in Ireland has changed significantly over the last 30 years, with large shifts in the migratory patterns linked to changing economic conditions, as well as the expansion of the EU. Up until the 1990s, Ireland had a longstanding history of net emigration (see Figure 1.1, dotted line). This changed due to a period of economic growth, which led immigrants from other countries, as well as returning Irish emigrants, to move to Ireland in search of employment. In the period 2004–2007, the years preceding the first wave of the GUI study, the country saw a large increase in net migration (the number of immigrants minus the number of emigrants) due to the expansion of the EU and the continuing economic boom (see Figure 1.1). The sharp recession in 2008, together with the rapid increase in unemployment, led to a large fall in immigration and increased emigration, and in 2010 Ireland entered a period of net emigration. As immigration recovered and emigration fell, by 2015 net migration was positive once again and has remained so.¹

¹ By 2020, net migration was just under 30,000, meaning that around 30,000 more people moved to Ireland to live than emigrated.



FIGURE 1.1 IMMIGRATION, EMIGRATION AND NET MIGRATION, 1987–2020

According to figures from Eurostat,² 16 per cent of the population in Ireland in 2008 were born abroad, of which 74 per cent were born in the EU and 26 per cent were born in non-EU countries.³ While EU nationals are free to work and live in Ireland, non-EU nationals are required to obtain residence permits in order to live in Ireland.⁴ Residence permits in Ireland are issued for a number of reasons, including education, work, family reasons and protection factors, such as asylum seeker status. Since 2007, work permits are typically issued for highly skilled jobs with an annual salary over €30,000 (see McGinnity et al., 2020a).⁵ Most non-EU migrants come to either study or work, with a smaller number seeking international protection or coming for other reasons, such as family reunification.

The immigrant population in Ireland is diverse in terms of country of origin, nationality, ethnic background and language proficiency. Figure 1.2 shows the

Source:
 CSO 'Population and migration estimates', various releases.

 Note:
 The red vertical line is the year of the first wave of the GUI study.

² Eurostat data explorer table [migr_pop3ctb].

³ Prior to Brexit in January 2020, migrants from the UK were counted as EU migrants. Post-Brexit, British nationals enjoy similar work and residence rights to EU nationals.

⁴ A small number of European Economic Area (EEA) migrants who are not EU citizens have equivalent rights to EU citizens in Ireland. The term EU is used here as it is more widely understood. See McGinnity et al. (2020a) for a discussion.

⁵ In January 2007, a new employment permit system was adopted with the aim of attracting highly skilled non-EEA migrants and restricting low-skilled migration from outside the EEA. Since 2007, there are a very restricted number of occupations with salaries less than €30,000 for which work permits can be issued (see McGinnity et al., 2012).

nationality of immigration flows between 2000 and 2008. Immigration increased gradually between 2000 and 2002. After the enlargement of the EU in 2004, immigration to Ireland from EU countries rapidly increased. Immigration from EU15 and EU28 countries made up most of the immigration flows from 2005 to 2008, accounting for 48 per cent of immigration in 2008.





Source: CSO, 'Population and migration estimates', up to end April of reference year.

Notes: *Rest of EU15 excluding Ireland and the UK: countries before enlargement on 1 May 2004 (Austria, Belgium, Denmark, Finland, France, Germany, Greece, Italy, Luxembourg, Netherlands, Spain, Sweden and Portugal). **EU15 to EU28: Ten countries that joined the EU on 1 May 2004 (Cyprus, Czech Republic, Estonia, Hungary, Latvia, Lithuania, Malta, Poland, Slovakia and Slovenia), along with Bulgaria and Romania, which joined on 1 January 2007, and Croatia, which joined on 1 July 2013. Prior to 2005, EU15 and EU28 migrants were included in the non-EU group.

Immigration peaked around the time of the GUI Cohort '08 infant interviews in 2007–2008 and then fell dramatically as a result of the economic recession (see also discussion of Figure 1.1). Immigration started to slowly increase from 2011, though by 2019 was still far from the 2006–2008 peak, particularly from EU28.

A distinctive feature of migrants in Ireland is that they are, in general, highly educated: this is particularly true for those of west European or Asian origin (McGinnity et al., 2020a). Eastern Europeans are the only group with somewhat lower levels of educational attainment (based on the highest qualifications of 25–34 year olds in 2019, see McGinnity et al., 2020a). Another distinctive feature of migrants in Ireland is that a significant proportion of non-EU nationals have

acquired Irish citizenship through naturalisation, which usually signals an intention to stay in the country on a permanent basis.

Immigration has also brought increased ethnic diversity to Ireland, though as most immigrants are of European origin, Black, Asian or Other ethnic groups represent a small proportion of the population.⁶ McGinnity et al. (2018), using Census data, show how the number and proportion of people who indicated an ethnicity other than 'White Irish' increased significantly between 2006 and 2016 in Ireland.⁷ Of the population who indicated an ethnicity other than White Irish, 'other White' remains the largest group, at 9.5 per cent of the population in 2016.⁸ That same census year, Asian or Asian Irish accounted for just over two per cent of the population, while Black or Black Irish made up 1.4 per cent of the total population. While ethnicity, nationality and place of birth overlap, they are not the same: 38 per cent of the Black ethnic group were born in Ireland and 62 per cent were born abroad, mostly in Africa. Fifty per cent of the Black ethnic group hold Irish citizenship (McGinnity et al., 2020a). The 'mixed' or 'Other' ethnicity category accounted for 1.5 per cent of the population, including those from a mixedethnicity background, and others who did not feel they fitted into any of the other categories.9

Migrants residing in Ireland are also diverse in terms of languages spoken at home.¹⁰ Non-Irish nationals who spoke a language other than English or Irish at home amounted to 363,715 persons in 2016 (CSO, 2016).¹¹ Eighty per cent of the non-Irish nationals who arrived in Ireland in 1996 or before indicated that they spoke English very well, while among the later arrivals (2015), only 44.4 per cent spoke English very well, with nearly one in five (19.1 per cent) stating they could not speak English well or at all (*ibid*.).¹² McGinnity et al. (2020b) analyse the self-reported English language skills of adult migrants from particular countries of birth using Census microdata. Not surprisingly, a high proportion of migrants from countries where English is widely spoken, such as the UK, the US, and many former British colonies, report that they speak English very well, as do many from western

⁶ See Chapter 3 for the definition of ethnic groups used in this report.

⁷ Information on ethnicity was collected for the first time in the Irish Census in 2006. Census respondents were asked: 'What is your ethnic or cultural background?' See https://www.cso.ie/en/media/csoie/census/census2016/2016censusforms/65995_English_Household_2016_New_V ersion_Do_Not_Complete.pdf, question 11 for detailed answer categories. A number of commentators have highlighted the limitations of this measure (see, for example, King O'Riain, 2007), but it is the only measure currently available in Ireland.

⁸ This group comprises a mixture of regions of origin, with east European nationals dominating, followed by UK nationals, west European nationals, and some White non-EU (McGinnity et al., 2018).

⁹ McGinnity et al. (2018) note this category is very diverse and includes people from Poland, Romania, Brazil, and the UK, among other countries. We cannot rule out the possibility of some respondents having difficulties answering the question. Note this is separate to 'not stated', which refers to those who did not respond to this question.

¹⁰ Figure derived from PxStat tables E7047 and EY025.

¹¹ See: https://www.cso.ie/en/releasesandpublications/ep/p-cp7md/p7dgs/.

¹² Speaking English capacity varied by national background, with Brazilian, Latvian, Lithuanian, Polish and Romanian respondents reporting low English language proficiency (CSO, 2016).

Europe. Origin countries where only one-third of migrants report they speak English well are scattered around the globe, but include Afghanistan, Brazil, China, Latvia, Lithuania, Moldova, Poland and Romania (*ibid*.). English language proficiency of parents is likely to impact on employment chances and household resources, but also the English language proficiency and academic achievement of migrant children. Chapter 3 considers in more detail the region of origin, language background and ethnicity of the parents of migrant-origin children in the GUI cohort study.

1.2 DEFINING SECOND-GENERATION MIGRANTS

The individual circumstances and experiences of both first- and second-generation migrant children and young people are likely to vary. For this reason, it is important to use clear, precise definitions in this area of research. Over the years, several definitions have been put forward as to how to distinguish between different migrant groups. Anderson and Blinder (2019) point towards a broad differentiation in migration literature and public discourses between 'immigrants' (individuals aiming to settle in the host country) and 'migrants' (individuals more likely to be temporary residents). However, the two terms are often used interchangeably by various interest groups. Moreover, available large-scale datasets have utilised different factors in defining foreign-born individuals, such as country of birth, nationality and/or length of stay. This ambiguity in definitions used is likely to result in inconsistency regarding studies seeking to quantify immigrant/migrant populations and those that explore the impact of migration on host countries, which in turn is likely to confuse policy debates on the topic (Anderson and Blinder, 2019).

Research to date has also differentiated between different migrant generations. First-generation generally refers to individuals born abroad who have migrated to host countries for a variety of reasons. These individuals are, arguably, the most affected by migration-related challenges and socio-cultural adaptation.

Much of the earlier research and theoretical work regarding second-generation migrants originates from the US and Canada, though European research on this topic has also gathered pace. In immigration research, the term 'second generation' generally refers to children born in the host country to at least one foreign-born parent (Eurostat, 2014).¹³ Some authors also include in this category children born abroad but brought to a host country before primary school (King et al., 2006). Molcho et al. (2006) note that these children are, at least to some extent, affected by the migration experience of their parents. In this report, we use the

¹³ There has been some criticism of using the term 'second-generation migrant' as it is seen to perpetuate immigrant status across generations (OECD, 2018, p. 18).

term 'second-generation migrant-origin children to refer to Irish children born to at least one migrant parent.

Several authors have put forward additional categories of generation. 'Generation 1.5' refers to children of those individuals who were born abroad and brought into the host country at a very young age (Röder and Ward, 2014; Rumbaut, 2004). Their identity encompasses a combination of 'new' and 'old' culture and tradition (Asher, 2011). While children belonging to '1.5G' spend their formative years in a new country, they may still be influenced by the traditions of their country of origin (Rumbaut, 2004). Further differentiation has been made between '1.75G' – children arriving to a host country under the age of five – and '1.25G' – young people between the ages of 13 and 17 years – to reflect the extent of children's experiences and socialisation in the host country (Rumbaut, 2004).

To date, few studies of second-generation migrant-origin children in Ireland have been conducted. This reflects migration trends and the age of the population concerned. The number of such studies is growing, however, and it seems reasonable to imagine that future research will explore the outcomes of other generations.

Much of the existing migration research in the US and Europe has conceptualised 'second generation' as one group. However, some authors have cautioned against the use of such a 'pan-ethnic' term, as it refers to a very heterogenous group (Vathi, 2015; Portes and Zhou, 1993). Thus, the category of second-generation migrants should not be considered a homogeneous one; rather, one in which subgroups experience outcomes that are better or worse than, or equal to, those of the comparable native population (Schurer, 2008). Eckstein (2002) argues that migrant generations are influenced by the different social conditions they experience in their pre- and post-migration times. To address the heterogeneity in the secondgeneration migrant group in Ireland, Röder et al. (2014) developed and utilised the following broad country groupings: IRL (indigenous families); UK; EU accession states; EU13; Africa; Asia; and other (Oceania, North America, Americas, non-EU European and unspecified nationalities). Disaggregating migrant data further is important, as several studies conducted in different European countries have found that second-generation young people from some countries fare less well in host countries compared to others (Crul and Vermeulen, 2003). Ethnic background may also play a role in understanding the outcomes of second-generation migrant children (Jonsson et al., 2018).¹⁴ Linguistic background is important too, given the key role of the host country language in the integration of children (*ibid*.). This is especially true in a country like Ireland, as English is widely spoken in many parts of the world. In researching second-generation migrant outcomes there are

¹⁴ Disaggregating by ethnicity is also consistent with plans, under the National Action Plan against Racism, to introduce an ethnic identifier more broadly in data in Ireland (Anti-Racism Committee, 2021).

additional factors to consider. For example, Röder et al. (2014) note that infants with two migrant parents may have different outcomes from those with one migrant and one native-born parent.

To date, research on second-generation migrants in Ireland has been limited. This is due to the relatively low levels of inward migration to Ireland up to the 2000s, and partly due to lack of data regarding parents' country of birth in available datasets. An important exception to this in terms of quantitative, representative data is the Growing up in Ireland (GUI) longitudinal study. The data available in GUI enable the exploration of outcomes for subgroups of children in greater detail, and the representative nature of the data makes it possible to generalise the findings to the wider population of this age cohort. This dataset enables a detailed exploration of different dimensions of child outcomes, including English language, wellbeing and self-concept. To date, little is known about how young migrants perform in the language of instruction relative to their native peers and whether gaps exist between groups. Furthermore, areas such as wellbeing and self-concept among migrant children have not received much attention in Irish research, despite the fact that wellbeing can tap into processes of adaptation and acculturation that are not captured by indicators of 'structural integration', such as employment or income (Jonsson and Mood, 2018). Measures of wellbeing have been widely used as an indicator of migrant integration in the host country, both for adults and young children (for example, Safi, 2010; Johnsson and Mood, 2018). Self-concept as part of a sense of belonging has also been considered important in migrant integration (Wu et al., 2012). Understanding how different groups of migrant-origin children fare in Ireland helps to pinpoint areas that need greater policy intervention. In this report, we define second-generation migrant-origin children as those born in Ireland where at least one parent was born abroad. Based on the information available on the survey, the basic grouping includes the following categories:

- both parents (or lone parent) born abroad;
- one (of the two parents) born abroad;
- both parents (or lone parent) born in Ireland.¹⁵

A more detailed classification is based on parental country or region of birth: Ireland; UK; eastern Europe; western Europe; Africa; Asia; the US/Canada/New Zealand/Australia; and other (mainly South and Central America). An additional grouping is based on the study child's linguistic background, distinguishing whether one or both parents is a native English speaker. Finally, we distinguish the second generation according to whether: both parents are from an ethnic minority group (Black, Asian or Other); both parents are of White ethnicity; or the study child is

15

⁵ In the GUI survey, in almost all cases, the primary caregiver was the child's mother (99.7%) and the secondary caregiver (where present) was the child's father. Throughout this report, we refer to primary caregivers as mothers and secondary caregivers as fathers.

from a mixed-ethnic background (one parent White, one parent ethnic minority). Chapter 3 outlines these classifications in more detail. Descriptive statistics present outcomes for these groupings separately, but as they are clearly related and overlap, the statistical modelling allows us to identify the association of each of these characteristics with child outcomes. Throughout the report we use the terms migrant-origin children, children from a migrant background or second-generation migrant-origin children interchangeably.

1.3 EDUCATION POLICY CONTEXT

1.3.1 Early childhood education

This section considers the educational policy context in Ireland for the integration of migrants, both second-generation Irish children and those who arrived in the country as children. Educational policy in Ireland is constantly evolving but this section only considers policy relevant to the GUI '08 cohort (from 2008 to 2018), discussing first pre-school and then primary school.

According to OECD (2021), participation in early care and education plays a key role in enabling migrant children to reach their full potential. Barriers include low availability of such services in general, as well as limited access specific to immigrant children. While issues of affordability are relevant to all, awareness of services can be lower among migrant-origin families.

In Ireland, non-parental childcare for children aged three years and younger is largely market-based and very expensive – one of the most expensive in the EU when the GUI children were of pre-school age (2008–2013) (Russell et al., 2018). In that period, there was limited availability of lower-cost childcare in community crèches, but considerable reliance on informal childcare such as grandparents, particularly for infants (McGinnity et al., 2013). This lack of investment and state support can be particularly challenging for migrant parents, many of whom do not have relatives living nearby and who may not be able to afford centre-based care.

Indeed, McGinnity et al. (2014) found that, with the exception of three year olds with west European mothers, all children of migrant origin were much less likely to be in non-parental childcare at age three, when compared with Irish three year olds. Linked to this, migrant mothers had considerably lower maternal employment rates than Irish mothers. Roeder et al. (2017) highlight, in particular, the situation of mothers from EU accession states, who, as labour migrants, had very high pre-birth employment rates, but low earnings and very low rates of non-parental childcare post-birth. Low participation of migrant children in non-parental childcare has also been found internationally (OECD, 2021). This is potentially of concern; in 2015, McGinnity et al. found a positive effect of attending centre-based childcare on vocabulary outcomes for children from non-English-speaking

backgrounds in Ireland, which was not found for children from English-speaking backgrounds.

The Early Childhood Care and Education Scheme (ECCE), introduced in 2010, was an important policy initiative for children from the '08 cohort of GUI. This universal scheme provides part-time education and care to pre-school aged children. Between 2010 and 2015, the scheme was available to children between the ages of three years, two months and four years, seven months and covered one school year (38 weeks) for 15 hours a week.¹⁶ While the scheme is free, parents and guardians could also pay for additional hours in the childcare facility if this option was available and they could afford it as, for many parents, 15 hours state-provided childcare would not be enough to cover working hours, especially when you factor in commuting time. This scheme had extremely high take-up from its introduction: McGinnity et al. (2015) found almost all (96 per cent) of the GUI '08 cohort had participated in the ECCE scheme by age five. Participation in this scheme was high across all social class groups, but given very low participation in (very expensive) centre-based care at age three, the jump in participation rates was highest for more disadvantaged groups (see Figure 6.2 in Murray et al., 2016).¹⁷ In Chapter 4 we consider the participation of migrant-origin children in this scheme. Lack of participation among this group would tend to indicate lack of awareness on the part of migrant families, as the scheme is free to all parents.

The Department of Education also provides an early childhood programme known as Early Start in a small number of schools to meet the educational needs of children between the ages of three years, two months and four years, seven months from disadvantaged areas. The programme is a one-year fully funded scheme offered in certain designated disadvantaged areas. The aim of the scheme is to enhance the overall development of young children and prevent school failure and to 'counteract the effects of social disadvantage'.¹⁸ Early Start preschools are located in vacant primary school classrooms and are staffed by both trained teachers and childcare workers.¹⁹ Similar to ECCE, classes are not full-time, with children attending class for two and a half hours each day (Monday to Friday). Children cannot be enrolled in both ECCE and Early Start at the same time.

A second policy approach to supporting migrant children at pre-school age is the provision of language screening and, where necessary, language support through pre-school (OECD, 2021). Language screening involves assessing the language development of young children to assess those falling behind. Early language screening and early (pre-school) language stimulation is a feature of several OECD

¹⁶ In 2016, this scheme was extended to two years.

¹⁷ In most cases, children who did not attend the ECCE scheme did not do so because of their participation in other schemes, such as Early Start (see below), or because the child had SEN (*ibid*.).

¹⁸ See https://www.gov.ie/en/service/78ff8-early-start-programme/#what-the-early-start-programme-is.

¹⁹ Parental involvement is highly encouraged under the Early Start Programme and parents are encouraged to volunteer to attend class activities.

countries. For example, Denmark and the UK screen at age two; a number of other countries, such as Austria and Germany, screen between three and six years (OECD, 2021). In most of these countries, early language stimulation is provided via pre-school settings. One such course, in the Netherlands, which provides ten hours per week of language development as part of pre-school education, was evaluated using a large representative national sample (Leseman et al., 2017). Findings indicate that this approach reaps large benefits in terms of language proficiency (*ibid.*). However, there is currently no early language assessment or support provided to non-English language learners at pre-school in Ireland.

1.3.2 Primary school

Ireland provides state-funded education to both Irish and non-Irish nationals at both primary and secondary school levels. Children usually attend their local school. However, parents can apply for their child to attend any primary or secondary school in the country. Some commentators have argued that active school choice on the part of families is a key feature of Irish education policy that contributes to the social stratification of schools (Smyth, 2016), and results in some 'desirable' schools being oversubscribed. Previous research has found that finding places in schools has been difficult for some migrant families; this is due to some schools being oversubscribed, as well as admissions policies that favour settled communities, such as waiting lists and prioritisation of children of previous students (Smyth et al., 2009).²⁰ In Ireland, Byrne et al. (2010) found an absence of school segregation, at least at the level found in many other European countries, mainly due to the geographical dispersal of the migrant population (see also Fahey et al., 2019), and the wide variety of national groups represented. However, Smyth et al. (2009) found that immigrant-origin students are over-represented in larger schools, schools located in urban areas and those with a socio-economically disadvantaged intake. The authors attribute this to the interaction between geographical location, parental choice of schools and school admissions criteria that tend to favour settled communities (ibid.).

While every child in Ireland has equal access to primary school education, targeted supports, such as language supports, are needed in schools for migrant-origin children to ensure that they have the same or similar opportunities and outcomes to children who are not from migrant backgrounds. The Intercultural Education Strategy 2010–2015, which aimed to 'ensure that all education providers are assisted with ensuring that inclusion and integration within an intercultural learning environment become the norm' (Department of Education and Skills,

²⁰ This situation has changed somewhat with the introduction of the Education (Admission to Schools) Act 2018. While this represented a major reform of admissions policy, with its stated aim of making 'rules around admissions to schools more structured, fair, and transparent' (Department of Education, 2021), it was implemented after the GUI cohort of children were interviewed at nine years. This policy change may influence the transition to second level schools of this cohort. For details of changes, see https://www.education.ie/en/parents/information/school-enrolment.

2010, p. x), was a key policy initiative. Under this strategy, resources were allocated for English as a second language (EAL) in both primary and secondary schools. Budget cuts in 2007 and 2008 saw the number of EAL supports reduced to a maximum of two teachers per school. In the 2012/2013 academic year, EAL supports funding was merged with special needs funding. This change enabled schools to change teaching hours between EAL and special needs supports. As EAL now falls under the General Allocation Model (GAM) and schools are not required to report how GAM funds are spent and allocated, it is not possible to assess whether supports are meeting language learning needs.²¹

As Smyth (2016) notes, the Irish education system is characterised by a considerable degree of school-level discretion as to how resources and processes are organised. Policies regarding migrant children are no exception. For example, English language proficiency assessment kits are disseminated to primary schools, but for the most part no standardised system of language assessment is in place in primary schools. There is also likely to be considerable variation in terms of how schools assign resources to English language tuition and special educational needs (SEN).

There was no successor to the Intercultural Education Strategy, but the Migrant Integration Strategy (2017–2020) includes 12 education-related actions (Department of Justice and Equality, 2017). The mid-term progress report on the Migrant Integration Strategy Review (2019) reported on the implementation of these actions (Department of Justice and Equality, 2019). Action 26, to pass the Education (Admissions to Schools) Act, noted above, was fulfilled when the Act was enacted in 2018. Action 29, to monitor the number of non-English-speaking migrant children in schools has also been addressed; since 2016/2017 the Department of Education and Skills has collected additional data on migrant children in the annual primary school census (ibid.).²² Action 31, monitoring the effectiveness of training for teachers in managing diversity and tackling racism, delivered via the Professional Development Service for Teachers (PTSD), was reported in the interim review (*ibid.*) as having minor problems or delays. In terms of fostering and developing positive attitudes towards diversity and celebrating difference (Action 36), the anti-bullying procedures published in 2013-2014 include several specific measures that specifically deal with identity-based bullying.²³

An important policy initiative to combat educational disadvantage in Ireland is the DEIS programme (Delivering Equality of Opportunity in Schools), introduced in

²¹ Schools that can demonstrate that they have high concentrations of pupils requiring EAL can apply for additional funds (Department of Justice and Equality, 2019).

²² Preliminary findings from this school census database, the primary online database, are in Department of Education and Skills (2017). See also: https://www.education.ie/en/Publications/Statistics/Primary-Online-Database-POD-/Primary-Online-Database-POD-.html.

²³ For further details on the actions and their implementation, see MIS interim review (2019).

2006 and designed to target additional resources to schools serving disadvantaged populations. To address the needs of pupils from disadvantaged communities, the DEIS scheme allows for smaller class sizes and provides additional funding, access to literacy and numeracy programmes and assistance with school planning (Smyth et al., 2015). An evaluation of the scheme in 2015 found some improvements in attendance, retention and exam performance, though the authors point to the continuing concentration of disadvantage in DEIS schools, especially Urban Band 1 schools.²⁴ In line with Weir and Kavanagh (2018), they also point to the fact that a significant proportion of disadvantaged students attend non-DEIS schools (up to one-half), meaning that they are unable to avail of the additional supports that DEIS schools provide (*ibid*).

A number of resources are also available to all primary school children who are in need of extra supports. These include resource teachers, pastoral care and learning supports, and are available to migrant children, where appropriate.

1.4 REPORT STRUCTURE

This chapter has set out the aims and scope of this research study, as well as relevant contextual factors such as migration and education policy in Ireland. Chapter 2 considers the theoretical frameworks that have been developed to understand the experience of second-generation migrant children in their receiving or host countries. It reviews previous international literature and, where available, Irish literature, on the cognitive and socio-emotional outcomes of migrant-origin children, alongside the factors at individual and family level that influence these outcomes. It also reviews literature on how these children's preschool and school experience serves to facilitate or impede their development.

Chapter 3 describes the evidence base for this report: the GUI Cohort '08 at ages nine months, three years, five years and nine years. It details the measures used for vocabulary, reading and socio-emotional wellbeing, how second-generation migrant children are identified and how their family background and school situation are measured. Using these data, Chapter 4 presents a profile of the second-generation migrant population in Ireland, including parental country of origin, linguistic and ethnic background, and participation in pre-school and primary school in Ireland.

Chapter 5 considers a key measure of cognitive achievement – English language ability – measured from age three through to age nine.²⁵ In doing so it explores how migrant-origin children compare to Irish-origin children, and how this pattern changes over time. Chapter 6 compares child self-concept at age nine among

²⁴ Smyth et al. (2015) also note improvement in overall literacy and numeracy in DEIS schools over the period 2007–2014, though they find no narrowing of the gap with non-DEIS schools.

²⁵ It is beyond the scope of this research to consider ability in mathematics.

migrant-origin and Irish-origin children, and also looks at how this relates to parental country of origin and a range of other factors, including participation in sport and cultural activities. Chapter 7 summarises the key findings of the report and reflects on their implications for policy and future research.

CHAPTER 2

Theoretical framework and literature review

2.1 INTRODUCTION

Various theoretical and conceptual frameworks have been developed for understanding the experiences of migrants in their receiving countries. Theoretical work regarding the integration and experiences of second-generation migrants has, until recently, mainly been carried out by US scholars (Vathi, 2015). These theories can broadly be divided into culturalist (Portes and Zhou, 1993) and structuralist (Ogbu, 1983) perspectives, depending on whether the emphasis is placed on acculturation or socioeconomic mobility in the host society (Portes et al., 2016).

Research exploring the development and outcomes of migrant children in receiving countries has a long history, mostly with a focus on western, especially 'old' migrant-receiving countries. Much of this research targets migrant children's academic development and focuses on older children and adolescents (Cheah and Leung, 2011). Studies on socio-emotional outcomes among immigrant children are limited and have tended to focus on bullying and racial discrimination (Priest et al., 2019). In Ireland, a growing body of research focuses on the experiences of migrant-origin young people, mainly drawing on the first generation. Much of this research has focused on the intersection of education and migration, exploring the experiences of migrant children in Irish primary and secondary schools (Smyth et al., 2009; Darmody and Smyth, 2018; Devine, 2011; McGinnity and Darmody, 2019; Faas et al., 2019). Existing research in Ireland has highlighted challenges confronted by migrant children in both academic and social spheres, which are often underscored by ethnicity and proficiency in the English language (McGinnity et al., 2015; Darmody and Smyth, 2015; Smyth et al., 2009). To date, little is known about the educational and social outcomes of second-generation migrant-origin children (with notable exceptions of research by Röder et al., 2014; Frese et al., 2015; Sprong and Skopek, 2021). This chapter provides a brief overview of theories and research on the integration and experiences of migrants, providing important contextualisation for the analysis and findings of this study.

2.2 THEORETICAL FRAMEWORK

A wealth of theoretical concepts and analytical tools have been utilised to study migrant integration in Europe and elsewhere. This body of knowledge demonstrates the complex nature of migrant integration, which is shaped by origin-country influences (such as migrant selection, cultural distance, ethnic profile and language spoken), the experiences of migrants (such as attitudinal response and discrimination), as well as structural characteristics of the receiving country (such as policies for children, education systems, childcare and social welfare systems).

Initial frameworks that focus on the integration of migrants include 'assimilation theory', which suggests that the longer migrants stay in a host country the greater the likelihood of the second-generation identifying with the dominant group (Waters, 1990). However, in the US, Portes and Zhou (1993) differentiated between three alternative pathways for the second-generation: assimilation into the poor underclass; acculturation and integration into the native middle class; and rapid upward social advancement, relying on a strong ethnic community. The theory of 'segmented assimilation' attempts to explain the factors that determine into which segment of American society an immigrant group can assimilate (Zhou, 1997). Extrafamilial influences such as neighbourhood, school and peer effects are important in educational outcomes for migrants and natives alike (Luthra and Soehl, 2015). Crucially, however, important 'gaps' in this theory concern its lack of recognition of the differences that exist within ethnic groups in terms of education, cultural background, social class (Thomson and Crul, 2007), and of gendered patterns of integration (Waldinger and Perlmann, 1998). Furthermore, educational outcomes of migrants are also affected by additional factors, such as (parental) legal status, perceptions of the host country, discrimination and characteristics of the co-ethnic community (Luthra and Soehl, 2015). Some migrant groups may fail to attain middle class status, despite parental skill levels and are, thus, incorporated into the 'working-class' stratum (Luthra and Waldinger, 2013).

In Europe, migration scholars have found the theory of segmented assimilation to be not easily generalisable to the European context, due to notable differences between national contexts (Crul and Vermeulen, 2003). However, the theory is helpful in explaining differences in migrant experiences and in identifying the factors at the individual/background and contextual level associated with differential outcomes. Context matters because it influences how structure, culture, personal agency and outcomes influence migrants and their children (Thomson and Crul, 2007). Thus, according to comparative integration context theory, migrant integration is strongly influenced by differences in the contexts in which integration takes place – the receiving country's institutional arrangements regarding education, its labour market, its housing market and its legislation (Crul and Schneider, 2010).

The impact of culture – the set of values, symbols, beliefs, languages and norms that guide human behaviour (see Menipaz and Menipaz, 2011) – has been widely considered in sociological and education research. According to cultural reproduction theory, one needs to consider a broader range of factors than financial resources alone when exploring individual outcomes. Bourdieu's cultural reproduction theory originated as a framework for understanding social class differences in the transmission of advantage from one generation to the next.

Cultural capital of the dominant group acquired at home, in school, or both, can be converted into social and economic advantage in later life. For Bourdieu, cultural capital is a competence in society's high-status culture – its behaviour, habitus and attitudes. Thus, cultural capital plays an important part in reproducing educational and social hierarchies; the amount of cultural capital that students 'inherit' from their family of origin is a function of their socioeconomic status (Bourdieu, 1979). Equally important is social capital – aggregate of the actual or potential resources that are linked to a durable network of somewhat institutionalised relationships of mutual acquaintances and recognition. The inheritance of cultural capital, along with economic and social capital, enables members of the dominant classes to reproduce their socioeconomic position (Bourdieu, 1998).

While initially formulated in terms of social class, Bourdieu's theory of cultural capital has been expanded by subsequent theorists (see, for example, Lareau and Horvat, 1999) to take account of cultural diversity in terms of nationality and/or ethnicity. These authors suggest that policies and practices within the schools of a receiving country are likely to be more familiar to students from the dominant group – the 'insiders' who possess this information as part of their cultural capital. By contrast, members of migrant groups, possessing different cultural capital, norms and values, often find themselves in the position of 'outsiders' (Bourdieu, 1998). Perception of difference in values and dispositions by individuals or groups in relation to others may result in cultural distance and may have a direct influence on student attitudes and outcomes (see Triandis, 1998).

This 'mismatch' between home and school cultures may vary across nationalities or linguistic groups, as well as social class, depending on various types of capitals at their disposal. Converting one form of capital into another is not straightforward, but is achieved by complex processes (Devine, 2009). Some individuals have more capital and so are dominant over those with less capital; others may have equal but different compositions of capital at their disposal, which puts them in a different relationship to other individuals or institutions. For migrant children, the school is their primary source of contact with the majority culture, and thus an important site for acquiring knowledge of the *lingua franca* of the receiving country, as well as culturally relevant knowledge, skills and attitudes (Park-Taylor, et al., 2007). Bourdieu (1998) argues that the 'outsider' status of migrants may differ across groups since some migrant families are more likely than others to be in a position to access information on the education system of the receiving country and to internalise its cultural preferences (Weine et al., 2004).

With respect to the educational outcomes of migrant children, the level of cultural integration – the sense of belonging their parents feel towards the host country – can affect their investment in educational activities (Schüller, 2015). According to the family investment model (Conger et al., 2010) parents can 'invest' in the

development of their children and do so by providing a stimulating learning environment at home and through family activities.

The literature on international migration suggests that migrants are not drawn at random from their country of origin and that nor is the receiving country randomly chosen. Rather, migrants are negatively or positively selected depending on a series of factors. Examples include: unobservable characteristics, such as migrant abilities, ambition and resiliency; observable skills, such as migrant education or occupation; and returns from educational attainment in the country of origin as well as in the receiving country (Rooth, 2007; Feliciano, 2020).

Status attainment theory refers to the transmission of socio-economic advantage from one generation to the next (also called intergenerational social mobility); it seeks to shed light on how the characteristics of an individual's family/socio-economic background relate to their educational attainment and occupational status in society (Campbell, 1983). In migration research, parental occupational status and educational level influence the status attainment of their children (Bauer and Riphahn, 2007). Status attainment theory can also illuminate opportunities for migrants to 'catch up' with their native counterparts. A number of studies have referred to 'migrant optimism' when discussing the outcomes of migrants and their children, underscored by the drive to succeed in their host country (Fernandez-Reino, 2016; Feliciano and Lanuza, 2015). According to literature in this area, the educational attainment levels of migrant children is influenced by a positive disposition towards education and future orientation among parents, as well as proficiency in the language of the host country (Doepke and Zilibotti, 2015).

Despite the disadvantaged background of some migrant parents, authors have found that parents often have high expectations and aspirations for their children's education and greater school engagement (Minello and Barban, 2012). While expectations and aspirations have sometimes been used interchangeably in previous studies on migrant outcomes, the terms differ, with the former referring to more realistic beliefs, while the latter describes desired outcomes - individuals' aspirations are generally higher than their expectations (Fernandez-Reino, 2016). A study in the UK has identified a gap in student expectations among those aged 14 years between ethnic minority and White British groups, even controlling for student achievement and socio-economic background. Higher educational expectations are experienced among the former, mostly driven by parental expectations (Fernandez-Reino, 2016). Various studies refer to the immigrant paradox whereby the children of immigrants in some cases perform well or even outperform children with native parents, despite facing various challenges (Crosnoe, 2013). This is evident even when both groups come from similar socioeconomic backgrounds (ibid.). Feliciano and Lanuza (2017) argue that higher educational achievement of the 1.5 and second generations in the US could be explained by their higher socio-economic background, which may not be adequately captured by standard measures of socio-economic status (SES).

When considering child outcomes, social stratification literature has often drawn on the conceptual framework of primary and secondary effects. This framework is now increasingly used in exploring ethnic inequalities in education. The former considers the differentials in academic performance between migrants and the majority group, controlling for the family socioeconomic background of students. The latter refers to the differences in choices between the two groups at relevant transition points during the educational career of the young people (Fernandez-Reino, 2016). While primary effects in the case of migrant children tend to be negative, with many lagging behind their native peers, secondary effects have been found to be positive considering higher continuation rates among migrants in education (*ibid*.).

2.3 LITERATURE REVIEW

The successful integration of migrant children and young people has been among the foremost policy challenges for many western countries. Evidence from different jurisdictions indicates that, compared to their native peers, these children and young people tend to experience more limited outcomes in terms of educational attainment, school performance, opportunities and life chances (Heath and Brinbaum, 2007; Becker et al., 2013). This can partly be explained by various background and structural factors. Background factors that influence the outcomes of migrant children include: migrant status or country of origin (Flisi, et al., 2016); ethnicity (Crosnoe and Turley, 2011); generation (OECD, 2018); family structure and characteristics - children from lone parent families and mixed (native-migrant) families (Sprong and Skopek, 2021); minority linguistic background (Sierens and van Avermaet, 2015; Turney and Kao, 2009); and parental socio-economic position and educational attainment (Borjas, 1992; Crosnoe, 2007; Bradley et al., 2001). Structural and school-level factors include: access to schools (Smyth, et al., 2009); school size; socio-economic and ethnic school segregation (Levels and Dronkers, 2008);²⁶ student-teacher relationships (Suárez-Orozco et al., 2009); and attitude to school (Ismail, 2019).

Taken together, the opportunities and life chances of migrant children are shaped both by resources within migrant families, as well as by the opportunities provided by educational and social institutions in the destination countries (Crul, 2007). Migration is associated with a set of contextual risk factors, interaction of which may result in a high likelihood of more long-term negative consequences for the cognitive, socio-emotional and educational development of many migrant children

²⁶ Segregation at the neighbourhood level has been found to influence progress in (host) language acquisition of adult migrants, though the strength and direction of the effect varies across migrant groups (Liebig and Spielvogel, 2021).

and young people, despite the resilience of some migrants/migrant groups (Dimitrova et al., 2016).

The sections that follow consider these factors in greater detail. It is important, however, to note that migrant children comprise a very diverse group and differences exist between jurisdictions as well as between and within migrant groups. Furthermore, Baysu et al. (2018) demonstrate that school performance of migrant young people within one ethnic group can vary across different national school systems.

2.3.1 Background factors and child outcomes

Much of the research on the outcomes of migrant children and young people have compared migrants (either as one group or a disaggregated group) and natives. Some authors argue that in order to fully understand the experiences and outcomes of migrant children, one needs to compare their outcomes to those young people who did not migrate (migrant selection effect) (Hoffmann, 2021; Zuccotti, et al., 2017). Positive migrant selectivity – whereby individuals who have migrated tend to have higher educational attainment, skills and resilience compared to their counterparts who remained – may help to explain patterns of success among migrants and their children across different domains (Feliciano, 2020). Not considering the migrant selection effect may lead to misinterpretation of unequal outcomes among migrants and their children, especially when assuming that intrinsic cultural differences explain such outcomes (*ibid*.). However, research in this area is limited, due to the unavailability of detailed data on migrant selection.

Children of migrants now make up a considerable proportion of the population across the developed world. International research suggests that even at preschool age, children differ in a number of important ways according to family migration history, generation, ethnicity and national origins (De Feyter et al., 2009). The reasons for migration are many and varied, but often include work, study, family and humanitarian protection. First-generation migrant families face many acculturation demands in the host country that impact on their integration and socio-emotional wellbeing (Oppedal, 2020).

Language proficiency

Proficiency in the language of the receiving country has been considered one of the main factors affecting the integration of migrants, as well as the academic and socio-emotional outcomes of migrant children (Kristen et al., 2016; Isphording, et al., 2016). Results of the Programme for International Student Assessment (PISA) across OECD countries show that, on average, first-generation migrant students are less likely to attain academic proficiency at the same level as their native peers
(OECD, 2018).²⁷ The average gap in test scores of migrant children and their native peers tends to vary across countries and is strongly associated with achievement differences in the parent generation and language spoken in the home (Dustmann et al., 2012).

In Germany, a different mother tongue to that of the language of instruction has been shown to have an additional significant negative association with child achievement in mathematics, even when controlling for socio-economic characteristics of the family (Becker, 2013). In the United States (US), Magnuson et al. (2006) reported that migrant children whose mother uses a non-English language in communication with them tend to have lower mathematics test results than children of native-born parents. This was not the case for migrant children whose mother spoke English with them (Magnuson et al., 2006, p. 1255). In Ireland, Kavanagh and Weir (2018) showed that children in fifth and second class (primary school) whose families spoke a language other than English/Irish at home had lower average reading achievement than pupils who spoke only English/Irish at home. Little difference was found among second class children in the mean mathematics score, based on the language spoken in the home, whereas at all other grade levels, children who spoke a foreign language at home outperformed their peers who spoke English/Irish (*ibid*.).

There is a growing interest in developing language proficiency outside the classroom, with authors highlighting the role of sport in the development of cognitive and language skills (Garcia-Hermoso et al., 2019). Language proficiency is also crucial in the socio-emotional adjustment of migrant children, as difficulties in the host language and ensuing difficulties in forming friendships with native peers have both been associated with depressive symptoms among the immigrant youth (Oppedal et al., 2020). Having high English proficiency has been found to be associated with higher self-esteem and fewer depressive symptoms among Latino migrants (Rumbaut, 1994).

Delay in language acquisition may undermine equality of opportunities from childhood for migrant children (Cavallo and Russo, 2020). Acquiring academic language proficiency has also been found to be instrumental in terms of the educational pathways of migrant young people considering high-stakes testing in many high-income countries (Suárez-Orozco et al., 2015). Passing or failing high-stakes standardised tests that evaluate academic proficiency has direct

²⁷ PISA distinguishes between four types of immigration background: first-generation immigrant students are foreignborn students whose parents were also foreign-born; second-generation immigrant students are students who were born in the country of assessment, but whose parents are foreign-born; students with an immigrant background include both first- and second-generation immigrant students (also referred to as 'immigrant students'); and students without an immigrant background refers to those born in the country of assessment or who have at least one parent who was born in that country. See:

https://www.oecd.org/education/school/Definitions.pdf#:~:text=PISA%20distinguishes%20among%20four%20types %20of%20immigration%20background%3A,country%20of%20assessment%2C%20but%20whose%20parents%20are %20foreign-born.

consequences for students regarding the options available to them. Students with low levels of the language of instruction have been found to be at a disadvantage in high-stakes testing (Anjeh, et al., 2006).

Ethnicity and cultural context

Educational outcome patterns also vary by ethnicity (Bottia, 2019), though these tend to differ across migrant-receiving countries and are linked to parental education levels (ibid.). As with cognitive outcomes, there are mixed findings from international research on the link between ethnicity and socio-emotional factors among migrant children and young people, possibly reflecting the heterogeneity of migrant groups. For example, while some studies have found greater socioemotional difficulties among Asian and Hispanic young people in the US than their European counterparts (Zhou, et al., 2003), other studies on Asians in the US have found no such differences (Cheah and Leung, 2011), possibly reflecting heterogeneity among the broad ethnic groups. These inconsistent findings highlight the need for more research regarding the social development of migrant children. Cultural context and parenting practices also seem to matter in terms of the socio-emotional outcomes of migrants. For example, warmth, reasoning and autonomy-granting among Chinese mothers regarding their pre-school children predicted greater behavioural and attention regulation abilities in these children, which in turn predicted decreased teacher-rated child socio-emotional and behavioural difficulties (Cheah et al., 2009). In another study by Izzo et al. (2000), use of parental warmth and control was found to be positively associated with child socio-emotional adjustment. Authors have also found an association between highly integrated migrant mothers and child pro-social behaviours (Shin et al., 2010). Cultural conflict among migrant parents and their children, in cases where the latter has adopted the cultural norms of the receiving country, has been found to be a strong negative predictor of migrant children's psychological, social and emotional adjustment (Sam, 2006). In some migrant families, children often acquire language proficiency more quickly than their parents. The resulting 'language brokering' has been found to be associated with greater internalising behaviour (such as feeling withdrawn) among Chinese American and Korean American adolescents (Chao, 2006).

Socio-economic background and family structure

Background characteristics and family resources matter to child outcomes (White, 2018). Several studies document a strong relationship between socio-economic background, parental education, childcare, family structure and cognitive development in early childhood (Anger and Heineck, 2010; Evans, 2004; Stevens et al., 2009). Overall, children from higher socio-economic backgrounds and with better educated parents tend to achieve better results at school (Crul, 2007).

One-parent families have among the highest risks of both material deprivation and income poverty (Watson, et al., 2018). Literature on family structure and child

educational and cognitive outcomes shows that single-parent family status has a negative impact on children's academic outcomes in a number of countries (Pong, et al., 2003). Single-parent families are more likely to have a lower standard of living, and family income tends to be a good predictor of a child's academic achievement. However, the debate on the causality of family structure and child outcomes is ongoing (Amato, et al., 2015). Research on lone parenthood and immigration has shown that migrant pupils from single-mother families score four points lower on a maths test than migrant pupils who live with both parents (Dronkers and Kalmjin, 2013), even when controlling for maternal SES and migration history. In Iceland, Chen and Ragnarsdóttir (2014) found that in addition to the challenges common among many lone parents, in terms of less time and fewer sources of income compared to two-parent families, additional challenges centre around integration and social interactions.

Research to date presents a complex picture for children with one migrant and one native parent (Sørensen et al., 2016; Dennis, et al., 2016; Sprong and Skopek, 2021). Overall, cognitive outcomes tend to be similar for children in mixed families with one native-born parent to those of their native peers. In Ireland, using data from the *Growing Up in Ireland* (GUI) study, Sprong and Skopek (2021) found migration-related disparities in verbal skills, already at the age of five. Furthermore, these disparities vary across the migrant groups, with children of Polish parents exhibiting lower scores, and tend to be more marked for children with two non-native parents. The language spoken at home was found to be a powerful predictor for migrant disadvantage (*ibid*.).

2.3.2 Structural and school-level factors

Early childhood education settings

Researchers analysing early childhood education and care programmes cite evidence that high-quality centre-based childcare has positive impacts on child development, particularly for disadvantaged children (Kulic et al., 2019). In France, a recent study by Berger et al. (2021) found that crèche attendance has a positive impact on language skills, no impact on motor skills and a negative impact on behaviour. The positive impact on language skills is particularly concentrated among disadvantaged children, which implies that facilitating increased crèche access among disadvantaged families may hold potential for decreasing early socioeconomic disparities in this area. Nursery school or crèche enrolment is generally lower among disadvantaged families (Gambaro et al, 2014). In Ireland, analysis using data from the GUI study did not find clear positive effects of centrebased care on vocabulary at age five, for all children or those from a disadvantaged background, though there was no measure of quality of childcare in the study (McGinnity et al., 2015). The one group for whom there was a small positive effect of centre-based childcare on vocabulary in Ireland was children from non-Englishspeaking backgrounds (ibid.). Yet crèche participation among immigrant families is generally lower than that for native-born families (Fortuny, 2010); this has also been found in Ireland (McGinnity et al., 2014).

Access to quality pre-school childcare is important for the socio-emotional development of children and in terms of preparing them for school, both academically and socially. According to parents' reports, children availing of centre-based care in Ireland had fewer emotional and peer problems, although somewhat higher conduct issues (Russell, et al., 2016). The authors of this study also found that centre-based care was associated with small but significant improvements in pro-social behaviour for children in lone-parent families (*ibid*.). In many receiving countries, migrant families tend to be more socio-economically disadvantaged and less likely to access centre-based care, compared to their native counterparts, despite the benefits accruing to these families and their children, especially regarding language development (Karoly and Gonzalez, 2011).

School settings

School systems have an important role to play in the outcomes of migrant-origin young people (Baysu et al., 2018). Educational researchers have identified school composition as one of the key areas responsible for school differences in overall academic success (NCES, 2015). Schools catering for students from different socioeconomic background may differ in many ways, including teacher quality, staffing ratios, school climate and teacher expectations. Academic outcomes can be influenced by the fact that many migrant-origin children tend to be concentrated in schools serving disadvantaged areas (Byrne, et al., 2010).

Aspects of destination countries impact the cognitive outcomes of migrant-origin children, over and above differences among migrant groups, something that becomes evident when comparing the experiences and outcomes of the same migrant group across different countries. This suggests that practices adopted by individual destination countries either support or hinder the educational progress of migrant children (Crul, 2007). Access to schools, integration policies and resources available all play a part in the settling-in process of migrant children (Smyth, et al., 2009). School admission policies and the tendency for migrant populations to be more concentrated within particular neighbourhoods have resulted in higher concentrations of migrant-origin children in some schools, and an extensive literature on how this may impact educational outcomes. Much of the research on migrant concentration in schools comes from the US, and generally shows that academic results are poorer in schools where migrants are more highly represented (Bottia, 2019). In other jurisdictions, results from studies on the impact of migrant concentration on educational outcomes tend to be more mixed. For example, while small negative spill-over effects from migrants to natives was found in a cross-country study by Brunello and Rocco (2011), no such effects of non-native English speakers on native students in English schools was detected by Geay et al. (2012). In the Netherlands, a study by Ohinata and VanOurs (2012) found no adverse effects for native students and some negative effects on migrant students' test scores in reading. In Denmark, however, negative effects of a higher migrant concentration were found for both native and migrant secondary school students (Jensen and Wurtz-Rasmussen, 2011). In Austria, Schneeweis (2015) found that migrant students have lower grades in schools that have a higher share of migrant students, with very little spill-over effect for native students. The author also found that a higher share of migrants within a particular grade reduced the likelihood of migrant students attending 'high track' schools after primary school, with no significant effects for native students.²⁸ The research also showed that the impact for migrant students was stronger when the share of the students from the same migrant group was considered.

Once social and individual characteristics were controlled for, Cebolla-Boado and Medina (2011) found that the concentration of migrants had no effect on achievement in Spanish schools. The effect of concentration was only significant if migrants represented at least one-fifth of the student body. The population attending schools where migrant-origin students were more highly represented tended to be more disadvantaged than the rest of the student population.

Social integration

International and Irish research has highlighted the important role teachers play in the experiences of children (especially at a younger age) and the development of their socio-emotional skills. Some Irish research studies refer to a lack of understanding among teachers regarding the backgrounds of and challenges experienced by migrant-origin children and young people (Darmody et al., 2011), as well as a misrecognition of minority social and cultural capital (Kitching, 2011), which may result in transmitting negative social stereotypes. Different cultural norms of teachers and migrant parents has also been found to have an adverse impact on teacher rating, both in academic competence and behavioural problems even after controlling for student gender and ethnicity, parental education and school involvement, as seen in the US study by Sirin et al. (2009). Data from the GUI study show that children whose mother was born abroad have fewer socioemotional difficulties and higher pro-social scores (based on the mother's report) than the children of Irish mothers, although the teacher-rated scale showed that this group of children had lower pro-social scores (Russell et al., 2016). However, other research, also drawing on GUI data, has pointed towards comparable teacher ratings on the Strengths and Difficulties Questionnaire (SDQ) for migrant and native students upon entry into formal schooling and more favourable teacher ratings for minority language children with poor English vocabulary skills compared to native children with poor English vocabulary skills (McNally et al., 2019).

²⁸ Low track schools provide basic general education and prepare students for further vocational education, while high track schools offer an academically-oriented curriculum.

Irish research has found that social integration is very important for young people, defined as feeling a sense of belonging to a group, being able to engage in similar activities and generally spending time together inside and outside school - in other words, feeling like an 'insider' rather than an 'outsider' (see Darmody and McGinnity, 2019; Darmody and Smyth, 2017; Darmody and Smyth, 2015; Smyth, et al., 2009). However, the research evidence shows that some migrant children experience difficulties in social integration in schools (Smyth, et al., 2009), have fewer close friends, especially some ethnic groups (Darmody et al., 2016; Liu, 2013), experience greater social distance from native peers (Tormey and Gleeson, 2012), and are confronted by stereotypical views on migrants in general, especially in terms of their reasons for moving to Ireland (Meade and O'Connell, 2009; Darmody and Smyth, 2015). Social connectedness to their native-born peers may serve to enhance the opportunities migrant students have at school and beyond (Darmody, 2011). Social interaction through meaningful shared activities (for example, sport and cultural participation) may also mitigate against social prejudices and racism (Darmody and Smyth, 2015; Darmody, 2011).

Cumulative exposure to racial discrimination and bullying victimisation has been found to result in socio-emotional difficulties among ethnic minority young people (Priest et al., 2019). Earlier studies have shown that awareness of racial stereotypes and outgroup biases – dislike for people outside their own identity group – emerge in middle childhood, which is usually defined as between six and 12 years (Aboud and Amato, 2001). Children recognise discriminatory acts by age 10 (McKown and Weinstein, 2003), suggesting that at this stage children are able to differentiate between bullying and racial discrimination (Priest et al., 2019). Bullying victimisation is associated with a range of negative impacts, including childhood mental health issues, anxiety and depression (Wolke and Lereya, 2015; Ford et al., 2017). The extent and nature of racial bullying may vary between ethnic groups. For example, bullying based on the stereotypical view of Asian young people in the US as a 'model minority', characterised as being comprised of highly motivated high achievers was found to be positively associated with more depressive symptoms and lower self-esteem (Rivas-Drake et al., 2009).

Participation in leisure activities and peer group dynamics

Participation in sport has been found to have a positive impact on children's education and behaviour (Felfe et al., 2016). Sport has consistently been cited as having a positive impact on the social integration of migrant children and young people. Participation in sport has been found to raise self-esteem and confidence among migrant-origin children as well as enhancing their popularity with peers (Erkut and Tracy, 2002).

Analysis of two waves of the GUI Cohort '98 study looked at the levels of participation in out-of-school social activities among migrant children living in Ireland, and found generally lower levels of leisure participation compared to their

Irish peers at both ages nine and 13, although the gap between Irish and migrant children narrows over time (Darmody and Smyth, 2017). On cultural participation, affordability is a factor as well as English language proficiency. There were lower levels of sports involvement among African, eastern European and, in particular, Asian families; while the reasons for lower levels of participation are not clear, it is possible that the types of activities on offer may be unfamiliar to some immigrant-origin children as they are heavily focused on team sports, such as GAA games *(ibid.)*.

As a result of peer group dynamics and exclusionary practices, as well as linguistic challenges, some migrants may revert to the 'comfort zone' of friendships with other young people from their country of origin (Darmody, 2011; Jonsson et al., 2018). This approach can negatively impact a child's number of friends and the composition of their friendship groups, as well as their sense of belonging and wellbeing. Making and keeping friendships is important for individual wellbeing and for emotional and social support over the life trajectory. Having at least one close friend is associated with better social skills and fewer adjustment problems (Dunn, 2004).

Outcomes of second-generation children and young people

There is now a considerable body of literature on the educational attainment of second-generation migrant children, most of which is US-based, though the volume of European research in this area is increasing. Much of this literature argues that the initial disadvantage faced by migrants leads to persistent gaps in the outcomes of second-generation children and young people (Krause et al., 2011). However, outcomes of second- generation are generally better than those of first-generation migrant children (OECD, 2018). In Ireland, PISA results for 2015 show that second-generation students on the whole do not differ from native students in their test scores in reading and mathematics, compared to first-generation students, who tend to perform at a considerably lower level in these disciplines (Darmody and Smyth, 2018). However, in Germany, Lüdemann and Schwerdt (2010) found that second-generation boys are more likely than their native counterparts to attend the lowest secondary school track (Hauptschule) and that both second-generation migrant boys and girls receive worse than average school grades in reading and mathematics.

Irish research which compared the academic outcomes of migrant children to that of natives found that many migrant students underperform academically (McGinnity et al., 2015; Darmody et al., 2016). It has also been found that language proficiency has a significant impact on the academic achievement of migrant students here (Darmody and Smyth, 2018; Darmody et al., 2016). In addition, differences can be observed between national or ethnic groups, with the lowest level of reading achievement among nine-year-olds found among children of eastern European origin (McGinnity et al., 2015). Emerging Irish research on a younger cohort of migrant children has shown that some migration-related disparities in verbal skills exist at the start of primary school (at the age of five), and that they differ across regional groups (Sprong and Skopek, 2021). The authors have also found that verbal skills differ between children with two migrant parents and those with one migrant and one native Irish parent.

The following chapters of this report explore differences in outcomes in early childhood and whether the differences between native and migrant children change over time.

CHAPTER 3

Methodology

3.1 ABOUT GROWING UP IN IRELAND (COHORT '08)

Participants were members of the *Growing Up in Ireland* (GUI) Cohort '08 (formerly known as the Infant Cohort).²⁹ The cohort members were recruited, with their primary caregivers (usually the biological mother), when they were infants.³⁰ The sample was nationally representative and based on a stratified random sample (for further details see McNamara et al., 2019). Data were collected during household interviews, with the first one conducted when the members were nine months old (in 2008–2009). Subsequent interviews took place at ages three, five and nine years (with a short postal survey following at seven/eight years, not utilised here). In the first wave, at nine months, there were 11,134 child participants, of whom 8,032 subsequently took part at age nine. A detailed description of the design, instrumentation and procedures for each wave are available from the GUI website (www.growingup.ie). A reduced sample of just under 7,000 respondents (unweighted) took part in all four waves and provided data for all the variables used in this analysis.



FIGURE 3.1 TIMELINE OF DATA COLLECTION FOR COHORT '08

A second older cohort of children were included in the GUI study from age nine; this is known as the '98 Cohort.

³⁰ The GUI survey instruments use the terms primary caregiver and secondary caregiver. In almost all cases, the primary caregiver was the child's mother (99.7%) and the secondary caregiver (where present) was the child's father. Therefore, throughout the report we refer to them as mothers and fathers.

3.2 OUTCOME MEASURES

3.2.1 Vocabulary test at three and five years

The Naming Vocabulary Scale was used as the cognitive test in GUI at ages three (Wave 2) and five (Wave 3). Taken from the British Ability Scales (BAS; Elliott et al., 1996), this scale measures expressive vocabulary (saying the word) as opposed to receptive vocabulary (understanding the word).³¹ The tests were administered in the child's home by the interviewer. In the Naming Vocabulary Scale, the interviewer showed the child pictures (drawings rather than photographs) of everyday objects, like a chair, and the child was asked to name the object (in English). If the primary caregiver judged the child to have insufficient (English) language skills to undertake this test, it was not administered. This applied to children with a disability or special educational need (SEN), as well as those who did not have English as their main language. In total, five per cent of children did not participate in the test at age three. Children from migrant backgrounds were over-represented among the non-completers, comprising 75 per cent of this group. (In Chapter 5, we take a variety of approaches to include the non-completers in the analysis to avoid bias.) The same tests were completed at both ages three and five years, with scores adjusted for the child's age (Williams, et al., 2019). At five years of age, only one per cent of children did not undertake the vocabulary test; this figure was three per cent for children with a migrant mother. For more detailed discussion of these tests and their psychometrics, see McGinnity et al. (2015).

3.2.2 Drumcondra reading test (vocabulary) at nine years

The Drumcondra English Reading Test was developed for Irish schoolchildren and is linked to the national curriculum. The vocabulary element of the test was administered at age nine. For each nine year old, interviewers were instructed to administer the Drumcondra test level that corresponded to the child's last year of school. Prior to analysis, scores were adjusted according to class level and the child's age at administration, so that they are comparable across the different levels. Only the vocabulary part of the test was administered (for further details see McNamara et al., 2020).

3.2.3 Piers-Harris Self-Concept Scale

Child self-concept was captured at nine years of age using a shortened version of the Piers-Harris Self-concept Scale (Piers and Herzberg, 2002; McNamara et al., 2020). The child participants completed these questions in their home as part of a self-complete questionnaire used for more sensitive or personal questions.

³¹ One of the particularly beneficial features of the British Ability Scales is that the core sub-tests are individually interpretable: to assess the level of performance it is not necessary to complete all tests in the battery (Elliot et al., 1997). This makes it particularly suitable for collection in a time-restricted survey setting such as the GUI study.

The scale consists of 31 items, which can be also subdivided into the following six subscales. (Some of the individual items load onto more than one sub-scale.)

- Behavioural adjustment A sub-scale of nine items on problematic behaviours, which includes perception of getting into trouble, or behaving badly at home or at school.
- Intellectual and school status A sub-scale of eight items reflecting the nine year olds' assessment of their abilities with respect to intellectual and academic tasks. It also includes items relating to general satisfaction with school and perceptions of future achievements.
- Physical appearance and attributes A sub-scale of seven items about perceptions of physical appearance and other attributes such as leadership and ability to express ideas.
- Freedom from anxiety A sub-scale of eight items exploring a variety of feelings including fear, unhappiness, nervousness, shyness and feeling left out of things.
- Popularity A sub-scale of six items exploring nine year olds' evaluation of their social functioning. This includes items such as difficulty making friends, being picked on and feeling left out.
- Happiness and satisfaction A sub-scale of six items reflecting feelings of happiness and satisfaction with life; for example, feeling unhappy and liking or disliking the way they are.

In the analysis, we focus primarily on the total score across the 31 items. The score ranges from three to 31. The mean score was 27 with a standard deviation of 4.21 (McNamara et al., 2020). A slightly higher proportion of children with two non-native English-speaking parents do not complete the Piers-Harris Self-Concept Scale (11.3 per cent) than those with two native English-speaking parents (8.9 per cent). This difference could be related to language difficulties in completing the questions.

3.3 MEASUREMENT OF MIGRANT BACKGROUND

We use a variety of measures to differentiate those with a migrant background, country of birth of primary and secondary caregivers (mothers and fathers), linguistic background of parents and ethnicity of parents. This information is taken from Wave 1 of the study from both primary caregiver (usually the mother) and secondary caregiver (usually the father) questionnaires. All of the children were living in Ireland by nine months of age, and a very small number had been born outside Ireland.³²

³² Of the 11,134 children, 114 were born abroad but came to Ireland before they were nine months old. We include these with the second-generation migrants.

3.3.1 Parental country of birth

The parents of the GUI children come from over 120 different countries; these have been grouped into eight categories in order to provide sufficient cases for analysis:

- Ireland;
- UK;
- eastern Europe, including EU countries (Bulgaria, Croatia, Cyprus, the Czech Republic, Estonia, Hungary, Latvia, Lithuania, Poland, Romania, Slovakia and Slovenia) and non-EU countries (Albania, Armenia, Belarus, Bosnia, Chechnya, Kosovo, Moldova, Russia, Serbia and the Ukraine);
- western Europe, including EU countries (Austria, Belgium, Denmark, Germany, Greece, Italy, the Netherlands, Portugal, Spain and Sweden) and Norway and Switzerland;
- Africa;
- Asia;
- US/Canada/New Zealand/Australia; and
- 'other' South and Central America and other countries not classified elsewhere.

We created a simplified second-generation identifier with three categories:

- both parents (or lone parent) born in Ireland;
- one parent born abroad and one born in Ireland; and
- both parents (or lone parent) born abroad.³³

In this study, children in the first and second categories are classified as secondgeneration migrants.

Linguistic background is captured from questions to both the mother and father on whether English is their native language. The information on both parents is combined into the following categorisation:

- both parents (or lone parent) native English speaker;
- one parent (mother or father) native English speaker and one not; and
- both parents not native English speakers.

Ethnic background of parents is classified as:

- both parents (or lone parent) White Irish or other White background;
- one parent Black, Asian or Other ethnic minority; and

³³ There is no case whereby the country of birth of a child's mother is missing, though for fathers, 24 per cent of cases have no information on the secondary caregiver's country of birth (see Table 4.2 below). In some cases, this parent may be born abroad. For consistency, we also use a consistent definition of migrant-origin children throughout the study: a child with two parents born abroad at nine months remains in this category throughout. A disadvantage is that some children may change 'status' through parental separation and/or re-partnering.

• both parents Black, Asian or Other ethnic minority.

Traveller status is included as a separate ethnicity in the questionnaire; however, in the research microdata file (RMF) data, Travellers are included in the White Irish group, and so cannot be separately identified here.

3.4 MEASUREMENT OF EXPLANATORY VARIABLES

A range of explanatory variables are included in the analysis to capture family social and economic status, child characteristics, participation in early education and care, school start and school characteristics. These variables are selected to measure the family, environmental and institutional influence on children's outcomes, some of which may mediate or moderate the influence of migrant background and are primarily based either on the literature review in Chapter 2 or previous work on developmental outcomes of this cohort in the GUI study (see for example Williams et al., 2016; McGinnity et al., 2015; Smyth, 2018).

3.4.1 Maternal education

Educational attainment was based on the highest qualification obtained. These were grouped into lower secondary or less, upper secondary (e.g. Leaving Certificate), non-degree (e.g. certificate or diploma level) and degree-level or above.

3.4.2 Low household income

Mothers provided an exact figure or best-guess estimate of household income (net of tax, PRSI, etc). This figure was then equivalised depending on the number of adults and children in the household, and divided into quintiles, with the highest (fifth) quintile comprising the wealthiest families. A dummy variable was created to measure families in the bottom two quintiles of the equivalised income distribution. An additional dummy was created to cover cases where income information was not available.

3.4.3 Special educational need (SEN)

Teacher reports were used to measure SEN at five years.³⁴ Children were recorded as having a SEN if a teacher reported them as having any of the following conditions to the extent that it limited their activity in school: physical disability or visual or hearing impairment; speech impairment; autism spectrum disorders; mild general learning disability; moderate/severe/profound general learning disability; specific learning difficulties (e.g. dyslexia); emotional or behavioural problem, such as attention deficit hyperactivity disorder (ADHD); or other limitations including

³⁴ The pattern of results is the same if we use SEN measured in Wave 5 at nine years of age, so we opted to use the same measure in the models of outcomes at age nine.

dyspraxia/motor problems; or other medical or health problem. Additional education needs due to language background or family difficulties were excluded from the measure.

3.4.4 Centre-based care

To analyse the influence of participation in formal childcare, we draw on a variable that records whether or not the child was in regular centre-based care at the age of three (and hence likely to have started early education before the rest of the cohort). In Chapter 4, we also examine whether take-up of the free pre-school place under the Early Childhood Care and Education Scheme (ECCE) differed between children of a migrant background and those whose parents were Irish born. Mothers were provided with information about the scheme and asked, 'Did you avail of the free preschool year for the Study Child?'³⁵

3.4.5 School start

This variable captures whether or not children had started school at the time of the Wave 3 interview, at age five, which was the case for almost three-quarters (72 per cent) of children. This variable also captures length of exposure to school at age nine.

3.4.6 School characteristics

At Wave 3 and Wave 5, information about the characteristics of schools were collected from school principals. Unlike Cohort '98, the sample for Cohort '08 was not clustered at school level. In Wave 5, the 8,032 children in the study were attending 2,365 schools.³⁶ Therefore, in most cases only a small number of the study children were attending the same school. Of the schools included, 83 per cent (1,863) have five or fewer students in the sample, and only 3.3 per cent (78) have over ten students in the sample. For this reason, we do not conduct multilevel analysis; instead, we include school characteristics as a separate step within nested ordinary least squares (OLS) or logistic regression models (see Section 3.4 below).

The social composition of schools is measured by Delivering Equality of Opportunity in Schools (DEIS) status. There are three designated DEIS statuses: DEIS Urban Band 1, DEIS Urban Band 2 and DEIS Rural. DEIS schools are identified as having a socially disadvantaged intake and are entitled to additional resources and supports. Urban Band 1 schools are the most disadvantaged.

³⁵ The preamble to the question stated, 'Children aged between 3 years 3 months and 4 years 6 months on the 1st of September each year are entitled to free part-time preschool places funded by the Government. For these questions, I would like you to think about only those preschool places funded by the free preschool year'.

³⁶ At the time (2017), there was a total of 3,246 primary schools in Ireland (Department of Education and Skills, 2019).

The proportion of children with a migrant background in a school is used as a second school-level characteristic. This measure is based on school principal reported figures of the total school population and, within that, the number of students with a migrant background. This has been recoded into a categorical variable in order to include those with missing values in the models. It is measured at Wave 5. Where there is missing information at Wave 5, we use the information collected at Wave 3 to reduce the number of missing cases.

3.5 ANALYTIC STRATEGY

In the chapters that follow, we analyse the GUI data using statistical models. Our strategy is to construct a set of nested models. First, country of birth information for mothers and fathers is entered. We then examine how these coefficients are altered by the inclusion of linguistic background of parents (the combined measure noted above) and the ethnic background of parents. The nested model approach allows us to exploit the rich data on family background in the GUI to investigate which aspects of child background – such as language and ethnicity – lie behind the parental country-of-birth effect. Following this, we examine how all three indicators change when controls for family socio-economic status (SES), child characteristics and school characteristics are added in three additional steps. In addition to the nested models, we conduct a Gelbach decomposition analysis (Gelbach, 2016) to assess how much of the country-of-birth effects are accounted for by the different groups of explanatory variables, which ensures that interpretation of the results is not determined by the sequence in which they are added. In the case of vocabulary and reading outcomes, we estimate a model in which we add the children's scores from an earlier wave. These are known as lagged dependent variable models and show how child progression over time relates to the explanatory variables.

The main models are OLS regression models. The outcomes (dependent variables) are continuous – vocabulary scores, reading scores and self-concept scores – and the results can be interpreted as a one-unit change in the outcome variable; for example, reading scores being associated with a one-unit change in the predictor variable. Many of the predictor variables are categorical, and in this case the coefficient can be interpreted as the unit change in the outcome measure associated with being in a given category compared to the reference category. For example, taking mother's education, a coefficient of +5.5 for 'degree' means that in cases where a mother has a degree, the study child has a 5.5 point higher score than children in the reference group, where the mother has lower secondary education. The effect of each variable is net of all the other explanatory variables in the model.

We also include two logistic regression models, where the outcome is a binary variable: being in the lowest quintile of the vocabulary/reading score or not.³⁷ In these models, coefficients are presented in the form of odds ratios; values above one indicate that the explanatory variable is associated with a greater likelihood of being in the bottom quintile for vocabulary/reading achievement compared to the reference group, while values below one indicate that the factor is related to a lower likelihood of being in the bottom quintile.

³⁷ We are primarily interested in the factors that influence low reading ability rather than variation across the whole distribution. For this reason, we do not use a quintile regression.

CHAPTER 4

Profile of the second generation

In this chapter, we outline the profile of the second generation in Cohort '08 of the *Growing Up in Ireland* (GUI) study. Section 4.1 considers the country of origin of the mother and the father (where he is co-resident), as well as the child's linguistic background and ethnicity, as important background information for understanding their school achievement and wellbeing. In addition to family background, an important theme in the literature on the integration of migrant children is exposure to the host or receiving country (Jonsson et al., 2018). For children, key sources of 'exposure' in Ireland are their participation in early education and childcare and school. Section 4.2 examines participation in early education and childcare and school start among children with a migrant background. Relating to debates articulated in Chapter 2, Section 4.3 considers some salient characteristics of the schools attended by second-generation migrant children at age nine.

4.1 BIRTHPLACE AND LINGUISTIC AND ETHNIC BACKGROUND OF PARENTS

Looking at parental country of birth across all families in the GUI Cohort '08 (Table 4.1), we find that, for over two-thirds (67 per cent) of children, their parents were both born in Ireland (or one parent, in the case of lone parent families). Almost one-fifth of children had parents who were both born abroad, and 14 per cent had one parent who was born abroad and one Irish-born parent. At the Wave 1 interview, at nine months, a somewhat lower proportion of migrant mothers were lone parents (12.4 per cent) than was the case for Irish-born mothers (15.4 per cent). By the Wave 5 interview, when the children were nine years, migrant mothers were more likely to be lone parents (16.8 per cent) than Irish-born mothers (14.4 per cent). Previous research suggests that the experience of migration may contribute to relationship breakdown (Kamijin, 2018). Analysis of the Census 2016 shows that lone parenthood rates also differ across migrant groups in Ireland (McGinnity et al, forthcoming).

TABLE 4.1 PROPORTION OF RESPONDENTS WITH IRISH AND NON-IRISH PARENTS

	Weighted %	Unweighted N
Both parents Irish	67.1	7,971
One parent born abroad	13.9	1,464
Both parents born abroad	19.0	1,699
Total	100.0	11,134

Source: GUI Cohort '08, Wave 1.

Notes: Lone parents are assigned to either the 'Both parents Irish' or 'Both parents born abroad' categories, depending on their country of birth.

Table 4.2 examines the country of birth of mothers and fathers. It shows that, following Irish-born, the two largest groups are those born in the UK and those born in eastern Europe, with each group accounting for around seven per cent of mothers and fathers). African and Asian migrants (roughly three per cent) comprise the next largest group, followed by those from western Europe.

	Mother	Father
Ireland	77.8	78.3
UK	6.5	7.3
Eastern Europe	7.0	6.3
Western Europe	1.3	1.2
Africa	3.2	2.9
Asia	2.6	2.8
US/Canada/Aus/NZ	1.1	0.9
Other	0.4	0.4
Ν	11,134	8,424

TABLE 4.2PARENTAL COUNTRY OF BIRTH (%)

Source: GUI Cohort '08, Wave 1

Notes: Couple and lone parent households. A total of 2,707 cases were excluded from the analysis of fathers as country of birth information was missing (24 per cent).

Obtaining Irish citizenship is associated with additional rights for those born outside of the EU and is also a signal of a long-term attachment to Ireland. Some migrants may have Irish citizenship before they move to Ireland by virtue of Irish ancestry or because they were born in Northern Ireland (see McGinnity et al., 2020a, for a discussion of citizenship attainment among migrants). The proportion of parents born abroad who are Irish citizens varies by country of birth, and is highest among those born in the UK or in the category comprising those in the US, Canada, Australia and New Zealand. Almost three-quarters of mothers from the UK (72 per cent) and 57 per cent of mothers from the US, Canada, Australia and New Zealand were Irish citizens. In contrast, 17 per cent of African mothers and 12 per cent of Asian mothers were Irish citizens. Mothers from eastern Europe (six per cent) were the least likely to be Irish citizens. These patterns of citizenship are broadly consistent with overall patterns of citizenship in the foreign-born adult population (McGinnity et al., 2020a).



FIGURE 4.1 PROPORTION OF MOTHERS BORN ABROAD WHO ARE IRISH CITIZENS

Source: GUI Cohort '08, Wave 1.

Note: Wave 1 results for western Europe are not included due to small cell sizes.

Figure 4.2 shows the ethnicity of mothers within the GUI. The results are very similar for fathers (not shown). Overall, 'White Irish' made up the majority of (83 per cent), followed by 'other White' (11 per cent). 'Asian' (three per cent), 'Black/African' (two per cent) and 'other' (one per cent) made up a significantly smaller proportion of ethnicity. The ethnic makeup of the GUI mothers reflects that of the population in the 2011 Census (see McGinnity et al., 2018).



FIGURE 4.2 ETHNICITY OF MOTHERS

Figure 4.3 presents the proportion of parents with English as a native language. A high proportion (86 per cent) of both the mother and the father (where present) were English native speakers. In terms of family context, 83 per cent of children had two native English-speaking parents and 11 per cent of children were living in families where neither parent was a native English speaker. A minority of children (five per cent) lived in households where one parent had English as a native language and the other parent was not a native English speaker.





Notes: Measured at Wave 1. Cross-sectional sample. N=11,134. Percentages do not sum to 100 due to rounding. Lone parents are assigned to the 'Both parents' category.

4.2 EARLY CHILDCARE/EDUCATION AND STARTING SCHOOL

As noted in Chapter 1, access to formal care early in Ireland is restricted by high costs, especially for young infants. Lower income may therefore act as a barrier for some migrant groups. Lack of knowledge about childcare services and entitlements may also reduce access to parents who are not familiar with the Irish system. In particular, market-based provision for children aged three and under may be difficult to navigate for newcomers. Limited provision of services results in extensive waiting lists, which can put newcomers at a disadvantage. At the same time, demand for formal care may be higher among migrant parents because they do not have ready access to family networks to provide informal childcare. Participation in formal childcare is also potentially influential for the acquisition of English language skills. McGinnity et al. (2015) find a small positive effect of participation in centre-based care compared to full-time parental care, at age three, on cognitive skills at five years for children from non-English-speaking backgrounds, but not for those from English-speaking backgrounds.

Previous GUI-based research showed that families with a non-Irish mother were less likely to use non-parental childcare for their child at age three (Murray et al., 2016). At the three-year interview, the study children were not yet eligible to participate in the Early Childhood Care and Education (ECCE) scheme.³⁸ When they did use non-parental care, migrant families were more likely to use centre-based care than home-based care – care by relatives or by childminders (*ibid*.). Figure 4.4 indicates that the use of centre-based care varies widely by the regional background of the mother. The highest level of use is observed among mothers of western European origin, followed by mothers from the US/Canada/Australia/NZ. In contrast, mothers from Asia and eastern Europe were significantly less likely to use formal centre-based care than Irish mothers. These differences are likely to partly reflect variation in the employment rates of mothers (see also McGinnity et al., 2014). As Roeder et al. (2017) note, using data from this cohort, eastern European mothers in Ireland have very high employment rates prior to having children, but this is then followed by low rates of employment.



FIGURE 4.4 PROPORTION OF CHILDREN IN CENTRE-BASED CARE AT AGE 3 BY MATERNAL COUNTRY OF BIRTH

Source: GUI Cohort '08 Wave 2, cross-sectional weight. N = 9,793.

Note: CoB stands for country of birth.

The study children were one of the first cohorts to be entitled to a free pre-school year (FPSY) with the introduction of the ECCE scheme in 2010. While that year saw very high take up overall, there is a noticeable difference in the proportion of families in which both parents were born abroad who availed of the service when compared to the rest of the population (see Figure 4.5). This is likely to reflect a lack of knowledge about the scheme: as it is free for parents, cost is unlikely to be a barrier (see also OECD, 2021). As by now, the scheme is much more established, this gap in awareness levels may no longer exist. We cannot differentiate further

³⁸ At the time of interviews, eligibility began at three years and three months. See pages 9-10 above.

by country of origin due to the small number who did not participate, though we do know that 90 per cent of children with both parents born abroad participated in the ECCE scheme.



FIGURE 4.5 PROPORTION OF CHILDREN WHO AVAILED OF ECCE FPSY BY PARENTAL COUNTRY OF BIRTH

Source:GUI Cohort '08 Wave 3 weighted by cross-sectional weight. N = 8,993.Notes:'Both' includes lone-parent families. CoB stands for country of birth.

While they are not legally obliged to start school until they are six years old, the majority of children in Ireland start school by or at five years of age (Murray et al., 2016). In many other European countries, children aged under six years are in preschool settings and, in fact, OECD statistics on 'early years' spending typically include spending for four- and five-year-olds in infant classes in primary school (ibid.). Compulsory school starting age also differs in sending countries. For example, in Britain it is five years, in Poland and Lithuania it is seven years, while in most of the EU it is six years (Sharp, 2002). Figure 4.6 shows that there is a significant difference in the age of school start by migrant background. Children with two migrant parents were significantly more likely to have started school by age five than those with at least one or both parents born in Ireland. If primary school in Ireland is considered a form of early care and education (albeit very different from that in preschool settings), this runs counter to patterns in many European countries, where participation of migrant-origin children in early care and education at the ages of five or six is typically lower than that of native-born children (Kulic et al., 2019). It is also in contrast to the participation rates of migrant-origin children in preschool at age three (Figure 4.4.) These two processes are likely to be related; if migrant parents are not accessing the ECCE they are more likely to start school early. School start is again potentially important for (English)

vocabulary development as it will influence the exposure of the child to formal learning through English.



FIGURE 4.6 PROPORTION OF CHILDREN IN SCHOOL AT AGE 5 BY PARENTAL COUNTRY OF BIRTH

Source: GUI Cohort '08 Wave 3, weighted by cross-sectional weight. N = 8,993. Notes: 'Both' includes lone parent families. CoB stands for country of birth.

4.3 SCHOOL DEIS STATUS AND MIGRANT CONCENTRATION AT AGE 9

This section considers evidence on two important features of the schools that migrant-origin children attend in Ireland, both of which are relevant to their development and integration. These features are: whether migrant-origin children are more likely to attend a disadvantaged school; and whether they are concentrated in particular primary schools and thus likely to be in schools with many other migrant children.

Previous research in Ireland showed that schools with a disadvantaged intake differ in many ways from non-disadvantaged schools, including teacher quality, school climate and teachers' expectations (McCoy et al., 2014; Smyth et al., 2015). Based on responses to a representative survey of principals in 2007, Byrne et al. (2010) found that, at both primary and secondary level, migrant-origin children were more likely to be concentrated in schools serving disadvantaged areas. The authors interpreted this as an interaction between residential segregation, parental choice and school admission criteria (*ibid*.). Was this still the case when this cohort of children were interviewed at age nine in 2017? Figure 4.7 presents the proportion of children by parental country of origin according to whether they were in any category of Delivering Equality of Opportunity in Schools (DEIS) school (see Section 3.3 for details of DEIS categorisation). Whereas around one-fifth (20 per cent) of children whose parents were both Irish were attending DEIS primary schools, this was the case for 28 per cent of children whose parents were both born abroad. The proportion of children in DEIS schools is highest for those who have a mother from Asia (37 per cent) or Africa (30 per cent), compared to 20 per cent for those with an Irish mother. The figure falls to 23–24 per cent for those with a mother from the UK or eastern Europe, and to 19 per cent for migrant-origin children whose mother is from any other county.

This pattern of over-representation is consistent with the findings of Byrne et al. (2010), and also reflects international trends. Interestingly though, children who have one parent born abroad are somewhat less likely to attend DEIS schools (17 per cent), suggesting that this group is distinct from children with both parents born abroad.

FIGURE 4.7 PROPORTION OF CHILDREN IN A DISADVANTAGED (DEIS) SCHOOL BY PARENTAL COUNTRY OF BIRTH



Source: GUI Cohort '08, Wave 5, weighted by cross-sectional weight. N = 7,250.

Notes: 'Both' includes single parent families. CoB stands for country of birth.

To what extent are migrant-origin children clustered in schools? Figure 4.8 presents the overall proportion of children in primary schools at age nine (in 2017) by the concentration of migrants in school, as reported by the school principal. (See Chapter 3 for details of measurement; this does not distinguish between first and second-generation migrants.) Almost 30 per cent of children in Cohort '08 of the GUI were found to attend schools with no migrant-origin children. A significant proportion (one-quarter) attended schools with a small proportion of migrants (less than five per cent). Around 27 per cent attended schools with between five and 19 per cent of migrants, while 20 per cent of children were found to attend schools in which over one-fifth of the school body is of migrant origin. Further analysis shows that children with both parents born outside Ireland were much more likely to be in schools with a higher concentration of migrant students: 43 per cent of this group were in schools where one-fifth or more of students were from a migrant background, compared to 19 per cent of those with one parent

born abroad and one born in Ireland and 16 per cent of children with only Irishborn parents (see appendix figure A.1).

Previous research (Fahey et al., 2019) found that the first-generation migrant population is widely distributed across the country measured at local electoral district level with some concentration within city centres. The pattern for school attendance suggests somewhat greater clustering at the school level. In Chapters 5 and 6, we consider whether the proportion of migrants in a school is associated with either cognitive achievement or socio-emotional wellbeing, both for all pupils attending the school, and for migrant-origin children specifically.



FIGURE 4.8 PROPORTION OF CHILDREN IN PRIMARY SCHOOLS BY CONCENTRATION OF MIGRANTS

Source: GUI Cohort '08, Wave 5, weighted by cross-sectional weight. N = 7,059. Excludes missing values (12 per cent of the sample).

4.4 SUMMARY

This chapter presented a profile of migrant-origin children from Cohort '08 of the GUI study, in terms of their backgrounds and their exposure to preschool and school in Ireland. Analysis of Wave 1 shows that for one-third of children in these data, either both parents (19 per cent) or one parent (14 per cent) was born abroad. This is using the 'broadest' definition of migrant-origin children, including parents born in the UK as born abroad, and recording those where one parent is born in Ireland, and one born abroad.

Reflecting the patterns of immigration to Ireland, half of the migrant-origin children had parents of either UK or eastern European origin, though a significant

minority had parents born in Africa or Asia. Fewer migrant-origin children had parents born in western Europe or the US. Overall, in this cohort, where mothers were not White Irish, they were mainly of 'other White' ethnicity (11 per cent of the sample). Children of Black mothers made up two per cent of the overall sample, and Asian mothers three per cent. In terms of linguistic background, 11 per cent of children lived in families where neither parent was a native-English speaker, with a further five per cent living in families where one parent was a native English speaker. This highlights that while one-third of children are of migrant origin, most of these have one native English-speaking parent. It will be interesting to investigate how the five per cent of children with one native English-speaking parent fare in terms of their English-language development.

Participation in preschool education is potentially beneficial for cognitive development, in particular for migrant children or children from a different language background. In 2011, around 30 per cent of children in this cohort were attending the (typically expensive) centre-based care at age three. A much higher proportion of children of west European and North American/Australasian mothers were found to attend centre-based care than those of eastern European, Asian and African mothers. Attendance of the part-time free preschool year, between ages three and five, at 96 per cent of children, was found to be much higher than participation in centre-based care at three years. Children with two parents born abroad have slightly lower participation rates than other children, though over 90 per cent of them still participate in this scheme. Most children in this cohort were in school by age five and, in contrast to preschool, children with both parents born abroad are more likely to be in school at age five (81 per cent) than children with two Irish parents (71 per cent). Being in school at age five may affect achievement scores at age five, but also will affect the duration of 'exposure' to school at age nine.

All the children in the cohort were attending primary school at age nine, but children with both parents born abroad were more likely to attend disadvantaged primary schools (28 per cent) than children with two Irish parents (20 per cent).

In terms of how migrants are concentrated in schools, in many schools in Ireland a small proportion of the student body is of migrant origin, as reported by principals. However, around one-fifth of all children in the '08 cohort were attending schools where over 20 per cent of students were of migrant background. Second-generation migrants were more likely to attend schools with a high share of migrant students. Chapters 5 and 6 will consider the effect, if any, of both family background and exposure to preschool and school on the cognitive achievement and socio-emotional adjustment of children in Cohort '08 of the GUI study. The next chapter considers child achievement in English and how this evolves over time

CHAPTER 5

English language and reading levels among second-generation migrant children

5.1 INTRODUCTION

In this chapter, we investigate the cognitive outcomes of second-generation migrant children in early to middle childhood. We focus on English vocabulary and reading outcomes;³⁹ this is because these outcomes are more likely to be related to migrant status and because competency in English is crucial for educational achievement in Ireland (Darmody and Smyth, 2018). Children whose first language is English tend to have better vocabulary and reading levels, compared to those for whom English is a second language (ibid, 2018). More generally, research has shown that competency in the language of the destination country is an important measure of migrant integration, something that remains relevant for the second generation (Jonsson et al., 2018). OECD (2018b) notes that language barriers and socio-economic disadvantage are two of the largest obstacles to the successful integration of young people with an immigrant background and that they play a significant role in gaps in educational attainment.

Children's expressive English language vocabulary was tested directly at age three and at age five using age-appropriate tests (see Chapter 3 for details of measures). At nine years of age, the children completed the Drumcondra Primary Reading Test, which is a set of curriculum-based standardised assessments of reading achievement for primary school pupils in Ireland. At age three, a minority of the child respondents did not undertake the English vocabulary test as their primary caregiver believed they would be unable to do so. Non-completion was higher among children of migrants (over 26 per cent of the children with both parents born abroad did not complete the test, compared to five per cent of the overall sample); therefore, we provide details on the scores at age five years because this is a more inclusive sample. In an analysis of children in the lowest quintile of reading scores at age three, those who did not complete the test because of insufficient English language skills are included.⁴⁰

It is important to note that the measure used in the *Growing Up in Ireland* (GUI) study is a measure of proficiency in English, rather than language ability *per se*. Children from a non-English-speaking background are likely to speak another language, and language proficiency is influenced by bilingualism. Studies have indicated that compared with monolinguals, young bilinguals have a smaller

³⁹ English is the language of instruction in most Irish primary schools (with the exception of 290 Gaelscoileanna). The Department of Education allows students to apply to be excused from studying Irish in English-medium primary and secondary schools. If a student has lived abroad or does not speak English, they may be exempted.

⁴⁰ Children with both parents born abroad made up just over 70 per cent of the non-completers at age three.

vocabulary if only one language is taken into account (Bialystok, 2006). However, when bilinguals' two languages were taken into account, their vocabulary was found to be similar to that of monolinguals (Poulin-Dubois et al., 2012). A more detailed study focusing on the linguistic ability of bilingual children would be required, and this was not explored in GUI. Of course, the first language of children may also change over time as they develop: this is the subject of the next section.

5.2 CHILD'S FIRST LANGUAGE

Given the importance of language for cognitive outcomes, we begin with parental reports of children's first language, data that were collected at each wave of the survey. What is interesting to note is that these reports change according to the child's age.⁴¹ At age three, almost all children with at least one Irish-born parent spoke English or Irish as their first language (97 per cent). Among children with two migrant parents, just over half spoke English as their first language at age three, a figure that rose to 60 per cent at age five and 73 per cent at age nine. A similar pattern emerges regarding parental linguistic background. A large majority (90 per cent) of children with at least one English-speaking parent had English as their first language at age three, rising to 96 per cent at both age five and age nine. Just over one-third of children with two parents (or a lone parent) with a non-English linguistic background spoke English as their first language at age three. But this rose to 62 per cent at nine years of age, suggesting a high level of English language adoption for those children with parents who both have a different linguistic background. This may be partly due to encouragement from parents, considering the importance parents attach to English language fluency in terms of educational attainment and options for the future.

⁴¹ The term 'first language' is used in the questionnaires. It is understood as the main language the child uses rather than the language that the child first learned. As the reported first language changes over time, it is clear that parents also interpret it in this way, rather than in a chronological sense.





Source: GUI Cohort '08 Wave 2, Wave 3 and Wave 5, weighted by longitudinal weight. Parental response to the question, 'What is the study child's first language?'

Notes: Children with two parents born in Ireland or where both parents are native English speakers are not presented. 'Both' includes single parent families. CoB stands for country of birth.

This 'linguistic integration' or adoption of the host-country language by children from a different language background is likely to have implications for both academic and social integration in Ireland.

5.3 ENGLISH VOCABULARY AND READING ACHIEVEMENT AT AGES 5 AND 9

5.3.1 Vocabulary achievement at age 5

How did the English vocabulary of migrant-origin children differ from that of Irishorigin children at age five? The vocabulary test at age five years has an average score of 55 and, as shown in Figure 5.2, the vocabulary scores of the second generation with one Irish and one non-Irish parent are no different from those with two Irish-born parents. Where both parents are born abroad, the children have a mean score of 45 compared to 55 for children of parents born in Ireland. There are significant differences by parental region of origin. In families where the mother was born in eastern Europe, the children had a mean score of 39; this rose to 47 for those born in Africa. Unsurprisingly, scores for children with a mother from the Anglophone countries in the 'US/Canada /Australia/New Zealand' category did not differ from those with Irish parents. Self-rated English-language competence of parents is not measured directly in the GUI study, but these patterns are consistent with self-reported English language proficiency by adults from these regions using 2016 Census data. Self-rated English-language skills vary considerably across country-of-origin group, but tend to be lower among eastern Europeans and higher for migrants who come from a country where English is widely spoken (McGinnity et al., 2020b, Chapter 3).





Source: GUI Cohort '08, Wave 3, weighted by longitudinal weight.

Notes: 'Both' includes single parent families. CoB stands for country of birth; BAME stands for Black, Asian and minority ethnic. For parental CoB, the UK is measured as abroad. 'Other' is not available for mothers' CoB as the sample (N) is too small.

Figure 5.2 also shows that children with both parents (or lone parent) from an ethnic minority background (Black, Asian or Other/mixed ethnicity) have lower mean English vocabulary scores at age five years than those where one or both parents are White.

The models below test whether there is any independent influence of ethnic background when linguistic background and socio-economic status (SES) are controlled. They find a significant gap in the mean vocabulary scores at five years between children where both parents are non-native English speakers (41) and

children with two English-speaking parents (57). Where one of two parents is not a native English speaker, the mean score is 54.⁴²

5.3.2 Overall English reading achievement at age 9

At age nine, the study children completed the Drumcondra Reading Test, which is adjusted for the school stage (for more details see Chapter 3). The measure has been rescaled to have a mean score of 100 and a standard deviation of 15. Figure 5.3 replicates the groups presented in Figure 5.2 to present children from different origin-country, linguistic and ethnic backgrounds.

FIGURE 5.3 MEAN ENGLISH READING SCORES OF 9 YEAR OLDS BY COUNTRY OF ORIGIN, ETHNICITY AND LINGUISTIC BACKGROUND OF PARENTS



Source: GUI Cohort '08, Wave 5, weighted by longitudinal weight.

Notes: 'Other' is not available for mother's CoB as the sample (N) is too small. CoB stands for country of birth; BAME for 'Black, Asian and minority ethnic'. For parental CoB, the UK is measured as abroad. 'Both' also includes single parent families.

The patterns for maternal country of birth are similar to those observed at age five, albeit with a different scale. Reading scores of children whose mother was born in the US/Canada, western Europe and the UK now have slightly higher mean scores than those born in Ireland, while children of African and eastern European origin have slightly lower mean scores, though the difference is not as great as at age five. Mean reading scores of Asian-origin children do not differ from those of children

⁴² Significance tests are available in the model so are not calculated here.

with Irish parents at age nine. Some of these factors may be related to socioeconomic background and other family and school characteristics. We investigate this using regression modelling in Section 5.5.

Figure 5.3 shows that children with parents who were both born abroad, as well as children with both parents of minority ethnicity and where both parents are nonnative English speakers, all have somewhat lower scores than the mean score (100). However, once again, the difference is not as marked as it is at age five (see Figure 5.2). The mean score for children without a native English-speaking parent is five points lower than the mean for those with two native English-speaking parents, which is only one-third of a standard deviation on the scale (SD = 15). It is also interesting to note that children who have one parent born abroad and one Irish parent, as well as those with one native English-speaking parent and one non-native English-speaking parent, actually perform slightly higher on English-speaking parents.⁴³ This suggests that having one Irish parent or one native English-speaking parent significantly facilitates children's English language development. In the next section, using models of reading levels at age nine, we investigate whether any other characteristics of these children and their families play a role.

5.4 TRAJECTORIES OF SKILL DEVELOPMENT AMONG MIGRANT CHILDREN (AGE 3 TO 9)

Another way of considering achievement and how it varies by groups is to focus not on the mean scores, but rather on those with low scores relative to their peers. For this, we divide children into five groups or quintiles based on their scores at each wave. This section focuses on the composition of the lowest quintile of achievement. These children are performing poorly in naming vocabulary in English/English reading, which may adversely affect their peer relationships, as well as their learning.

Are second-generation migrant children over-represented in this lowest quintile? Table 5.4 presents the proportion of children in the lowest quintile of English vocabulary/reading by parental country of birth. At age three, there are very clear differences between children with either both or one parent born in Ireland and children with both parents born abroad. At age three, three-fifths of those in the latter group are in the bottom quintile of English reading. This falls to just over half (52 per cent) at age five. What is remarkable is that four years later, at age nine, only one-quarter of these same children, where both parents were born abroad, are in the lowest quintile of achievement. Given the children would have been at school for all (or at least most) of these four years, it does suggest considerable

⁴³ These groups overlap, but not completely. For example, a child with a parent from the US and another from Turkey might count as having one non-native English-speaking parent, but with both parents born abroad.

progress is made in terms of achievement in English reading by second-generation migrants in Irish schools.

It is also of note that by age nine, while one-fifth of the children of Irish parents are in the lowest quintile in English reading, a lower proportion (14 per cent) of those with one Irish parent are in the lowest quintile. This suggests that this group of children are performing better than children with two Irish parents. This is shown in Figure 5.3, which presents mean reading scores. That said, these families may be advantaged in terms of having higher parental education and higher incomes, which is the focus of Section 5.6.





Source: GUI Cohort '08, Wave 2, Wave 3 and Wave 5, weighted by longitudinal weight.

Notes: *Age 3 scores include those who were not able to complete the test at age 3 in the bottom quintile, which is why 23 per cent of the total sample is in the bottom quintile at age 3, rather than 20 per cent. 'Both' includes single parent families.

Figure 5.5 presents children's linguistic background, rather than parental country of birth. The figure presents the proportion of those in the lowest quintile of English vocabulary/reading among children with two native English-speaking parents, one native English-speaking parent and two non-native English-speaking parents. For children with two non-native English-speaking parents. For children with two non-native English-speaking parents, the change over time is even more dramatic than in Figure 5.4: almost three-quarters (73 per cent) of this group were in the lowest quintile of vocabulary at age three, two-thirds (65 per cent) at age five but only 28 per cent by age nine. That said, even at age nine, this group were more likely to be in a lower achievement quintile than children with at least one native English-speaking parent.

Figure 5.5 also shows how, by age nine, children with one native English-speaking parent are less likely to be in the lowest quintile in English reading than children with two native English-speaking parents (11 per cent versus 20 per cent). As with children with one parent born abroad, with whom there is likely to be significant overlap, this may be an advantaged group in other ways, something we investigate using regression modelling.





Source: GUI Cohort '08, Wave 2, Wave 3 and Wave 5, weighted by longitudinal weight.

Notes: Age 3 scores include those who were not able to complete the test at age 3 in the bottom quintile, as in the model above (Table 5.2). 'Both' includes lone parent families.

5.5 MODELS OF VOCABULARY AT AGE 5

5.5.1 Mean vocabulary scores at age 5

To what extent are these differences in achievement by parental region of origin related to linguistic and ethnic background, other parental resources or characteristics of schools attended? Some international research argues that much of the gap between immigrant-origin and native-origin children is related to socioeconomic background (see Chapter 1). Many immigrant parents are engaged in lower paid jobs, hence the disadvantage is related to lack of resources (Heath and Brinbaum, 2007; Crosnoe, 2007). Multivariate models presented below investigate the relationship between parental region of origin, ethnicity, language and naming vocabulary in English at age five. These are presented as nested models, which successively add parental resources (their level of education and financial resources); child characteristics, including gender, special educational needs (SEN) and exposure to centre-based childcare and primary school; and key relevant school characteristics (whether the school is a designated disadvantaged (Delivering Equality of Opportunity in Schools, DEIS) school and the proportion of migrants in the school). A final model adds the child's score on naming vocabulary at age three as a way of measuring the association between these factors and change over time in vocabulary score. A decomposition analysis investigating which explanatory variables account for most of the change in the country of birth coefficients between Model 1 and Model 5 is also undertaken.⁴⁴

In the absence of other controls (Model 1, Table 5.1) both maternal and paternal country of birth are significantly related to English vocabulary scores at age five. All groups score significantly lower than those with an Irish-born mother, with the exception of those whose mother comes from the UK and the US/Canada/Australia. The lowest scoring group comprises children with mothers from eastern Europe: they score almost 15 points lower on average than children with Irish mothers. When the child's father is born outside Ireland/UK, there is an additional negative association with vocabulary scores.⁴⁵ As we want to retain those in lone-parent families, children with no father present are included as a separate category. Children from lone parent families also have lower scores on English vocabulary at age five, consistent with previous research (Nixon and Swords, 2016). The same is found for those whose father was present in the home but did not complete the survey. Further analysis (available from the authors) shows a more negative effect of lone parenthood on children's vocabulary at five years for children with migrant mothers. Due to the relatively small size of the migrant lone parent group, we did not estimate separate effects for each country of birth.

Model 2 introduces language and ethnicity. Here we see that children with two non-native English speaking parents have lower English vocabulary scores than those children with two native English-speaking parents. Linguistic background is closely associated with parental origin country: once we account for linguistic background, the influence of parental country of birth is much reduced. For example, Asian-origin children no longer differ from children whose parents are Irish, while the effect of being of eastern European or African origin is also reduced. Being from an ethnic minority background is not associated with vocabulary at age five, at least when parental origin country and linguistic background are accounted for. The decomposition analysis shows that linguistic and ethnicity factors controlled for in the models play the biggest role in reducing the country-of-birth effect for all maternal country-of-birth categories except regarding those from the UK and US/Australia: the impact is largest for mothers born in eastern Europe, Africa, Asia and 'other' country of birth. Similarly, linguistic and ethnic background

⁴⁴ This is available on request from the authors.

⁴⁵ To reduce the number of categories in the model, paternal country of birth is combined.

factors play the largest role in reducing the effect of fathers born outside Ireland and the UK, but play no role in accounting for the 'father born in UK' coefficient.

Model 3 shows that while parental education and income are clearly associated with vocabulary at age five (whereby children disadvantaged on either income or parental education have lower vocabulary scores), this does very little to explain differences by maternal country of origin or linguistic background. The decomposition analysis shows that socio-economic position plays a small role in the negative effect seen among those whose mother was born in Africa and those whose father was born outside Ireland or the UK. Socio-economic status also accounts for a significant increase in the coefficient among those whose mother was born in western Europe and the US/Australia.⁴⁶ The decomposition shows the socio-economic position of the family plays the largest role in accounting for the effect of the father being non-resident. Therefore, the lower vocabulary scores experienced by second-generation migrant children in Ireland are not primarily accounted for by their socio-economic background, as is the case in some other European countries (OECD, 2018).

⁴⁶ In the case of the US/Australia, socio-economic status variables have the greatest influence on the maternal countryof-birth effect.
		Model 1	Model 2	Model 3	Model 4	Model 5	Model 6
Parental CoB	Mother – UK	0.29	0.3	0.51	0.53	0.41	0.34
(Ref. Ireland)	Mother – eastern Europe	-14.75 ***	-10.61 ***	-10.9 ***	-10.98 ***	-10.56 ***	-5.89 ***
	Mother – western Europe	-3.8 ***	-3.09 *	-3.75 **	-3.8 **	-3.58 **	-0.4
	Mother – Africa	-5.85 ***	-2.11 ^	-2.1 ^	-2.14 ^	-1.81	-1.25
	Mother – Asia	-4.45 ***	-0.07	-0.41	-0.53	-0.18	0.34
	Mother – US	0.21	0.13	-0.68	-0.39	-0.56	-0.62
	Mother – 'other'	-5.06 *	-1.97	-2.04	-1.81	-1.48	2.04
(Ref. Ireland)	Father – UK	0.92 ^	0.93 ^	0.93 ^	0.8	0.73	0.55
	Father – born outside IRL/UK	-4.62 ***	-2.93 ***	-2.74 ***	-2.76 ***	-2.57 ***	-1.71 **
	No father/LP	-1.31 **	-0.84 ^	-0.35	-0.28	-0.23	0.22
	Father info missing	-3.34 ***	-2.98 ***	-1.19 **	-1.3 ***	-1.08 **	-0.53
Parental language	One native Eng		0.44	0.35	0.24	0.08	0
(Ref. Both native Eng)	Both non-native Eng		-6.43 ***	-6 ***	-6.3 ***	-6.51 ***	-2.88 **
Parental ethnicity	One BAME		1.87	1.95 ^	1.58	1.58	0.87
(Ref. Both White)	Both BAME		-1.15	-1.04	-1.52	-1.33	-0.37
Mat. educ.	Upper secondary			1.82 ***	1.5 ***	1.37 ***	0.34
Ref. Lower	Non-degree			3.02 ***	2.72 ***	2.53 ***	0.96 *
secondary)	Degree or higher			3.88 ***	3.62 ***	3.43 ***	1.43 ***
Family income	Low income* (Wave 1)			-1.58 ***	-1.42 ***	-1.41 ***	-0.55 *
	Income missing			-0.38	-0.3	-0.31	0.03
Child characteristics	Female				0.82 ***	0.83 ***	-0.32
(Ref. No SEN)	Has SEN				-4.39 ***	-4.31 ***	-2.09 ***
	SEN missing				-1.07 *	-1.74 **	-0.91 ^
(Ref. not in school by age 5)	In school by age 5				2.05 ***	2.18 ***	1.58 ***
	Centre-based care age 3				-0.16	-0.05	-0.03
DEIS status	DEIS Urban 1					-1.74 ***	-1.23 **
(Ref: non-disadvantaged)	DEIS Urban 2					0.22	0.06

TABLE 5.1OLS REGRESSION OF VOCABULARY AT AGE 5 AND CHANGE OVER TIME (3-5 YEARS)

TABLE 5.1 (CONTD.) OLS REGRESSION OF VOCABULARY AT AGE 5 AND CHANGE OVER TIME (3-5 YEARS)

		Model 1	Model 2	Model 3	Model 4	Model 5	Model 6
	DEIS Rural					1.16 *	1.1 *
	DEIS missing					0.77	0.97 ^
Migrant prop. In school	< 5%					-0.67 ^	-0.87 **
(Ref: no	5-9%					-0.35	-0.17
migrant students)	10-19%					-0.78 ^	-1.01 **
	> 20%					-1.18 **	-0.73 ^
	Prop. Missing					0.3	-0.41
T score for naming vocabulary (a	ge 3)						0.39 ***
Constant		57.27 ***	57.14 ***	54.93 ***	54.04 ***	54.49 ***	35.81
	Number	8,514	8,514	8,514	8,514	8,514	8,193
	Adjusted R square	0.135	0.143	0.161	0.185	0.189	0.282

Source: GUI Cohort '08, Wave 1, Wave 2 and Wave 3. Weighted by longitudinal weight.

Notes: *** P < 0.001, ** p < 0.01, * p < 0.05; ^ p < 0.1. Family low income if in bottom two income quintiles. CoB is 'country of birth'; LP is 'lone parent'; BAME is 'Black, Asian and minority ethnic'; SEN is 'special educational needs'.

In Model 4, we see how a child's own characteristics (gender and SEN status) are clearly associated with vocabulary scores, with girls performing slightly better and children with SEN having lower vocabulary scores. Having started school at age five is also associated with higher vocabulary scores, while having attended centrebased care at age three is not associated with vocabulary scores (see also McGinnity et al., 2015).

Model 5 shows that children in DEIS Urban Band 1 schools have lower achievement scores than children attending non-DEIS schools, even after accounting for family income and the range of other individual and family factors. We further test for differences in attendance at DEIS Urban Band 1 schools between second-generation migrant children and those with Irish-born parents. We find that there is an additional negative effect for children with two migrant parents (-2.5) on top of the negative effect of DEIS Urban Band 1 for children with two Irish-born parents (-1.5) compared to all children in non-DEIS schools. The effect for children with one migrant and one Irish parent is the same as for those with two Irish-born parents.⁴⁷

In addition, children in schools with a higher proportion of migrants (20 per cent or more) have slightly lower scores on vocabulary at age five. The direction of the relationship is the same for schools with a lower proportion of migrant students but is not significant at the five per cent level. This is consistent with some international studies that find that children in schools with higher proportions of migrants tend to have somewhat lower achievement levels (Bottia, 2019). This finding is for the whole sample, meaning that the effect is present for children with Irish-born parents as well as those with migrant parents.⁴⁸ These effects, combined with the fact that migrant children are more likely to be both in DEIS Urban Band 1 schools and in schools with a higher proportion of migrants, serves to slightly reduce country-of-origin and linguistic differences, though the effect is very small.

It is the introduction of the child's own vocabulary score at age three (Model 6) that really reduces the country-of-origin and linguistic background differences. There are still differences between children with eastern European or African mothers and those with two non-native English-speaking parents. This suggests that these children make less progress between ages three and five than others. For other regional groups, there is no statistically significant difference in English vocabulary development over time.

⁴⁷ The interaction models are available on request from the authors. In the model, we do not distinguish between parental country of birth as the groups would become very small. The interaction between parental native language and DEIS Urban Band 1 is not significant; this may be due to the smaller numbers in the 'two non-native English-speaking parents' category.

⁴⁸ The interaction between the proportion of migrant students (measured on a continuous scale) and the migrant background of the children themselves was tested and found to be non-significant (not shown). This indicates that the effect of the proportion of migrant students within a school is the same for both migrant and Irish-origin children.

5.5.2 Low vocabulary at age 5

As noted in Section 5.4, an alternative way of considering English vocabulary achievement and how it varies across different groups is to focus on children with lower levels of achievement, rather than mean scores. As before, we consider the scores of all children at age five, and then identify those in the lowest quintile of achievement. These children are struggling in naming vocabulary in English, which may adversely affect their peer relationships as well as their learning and motivation to learn.

Table 5.2 presents a model investigating factors associated with being in the lowest quintile of vocabulary at age five. Groups with an odds ratio greater than one are more likely to be in the bottom quintile of achievement than the reference category, while those with an odds ratio less than one are less likely to be in the bottom quintile of achievement. Not all controls are presented, but factors that have been accounted for are outlined in notes beneath the table.

Children whose mother was born in the UK, Asia, the US or another foreign country do not differ significantly from the children of Irish mothers, at least when other factors are accounted for. But the children of eastern European, western European and African mothers are more likely to be in the lowest quintile of vocabulary scores than the children of Irish mothers. The children of eastern European mothers are six times more likely to be in the low English vocabulary quintile at age five. Having a father born abroad is also associated with a greater likelihood of being in the lowest quintile, as is having two non-native English-speaking parents. Parental ethnicity is not associated with achievement, once origin and linguistic background are controlled. Adding information on whether the child was in the lowest quintile of vocabulary at age three considerably reduces these differences, though the effect of maternal region of origin and linguistic background remain.

		Model 1		Model 2	
Parental CoB	Mother – UK	0.96		0.97	
(Ref. Ireland)	Mother – eastern Europe	6.20	***	4.34	***
	Mother – western Europe	2.77	***	2.20	**
	Mother – Africa	1.74	*	1.83	*
	Mother – Asia	1.36		1.41	
	Mother – US	0.95		0.88	
	Mother – 'other'	1.72		1.54	
Father's CoB	Father – UK	0.76	۸	0.77	
(Ref. Ireland)	Father – born outside IRL/UK	1.47	**	1.35	*
	Father – not resident	1.06		1.06	
	Father – info missing	1.19	۸	1.14	
Parental language	One native Eng	0.83		0.84	
(Ref. Both native Eng)	Both non-native Eng	2.42	***	1.78	**
Parental ethnicity	One BAME	0.61	۸	0.58	٨
(Ref. Both White)	Both BAME	1.28		1.02	
Maternal education	Upper secondary	0.77	**	0.86	٨
(Ref. Lower secondary)	Non-degree	0.65	***	0.76	*
	Degree	0.58	***	0.72	**
Income Wave 1 (top 60%)	Bottom 40%	1.34	***	1.25	**
	Child female	0.88	*	1.00	
	Child SEN Wave 3	2.48	***	1.97	***
	Inschool at Wave 3	0.61	***	0.64	***
	Centre care Wave 2	0.83	**	0.90	
DEIS school Wave 3	Urban Band 1	1.19		1.22	۸
(ref no)	Urban Band 2	1.01		0.93	
	Rural DEIS	0.82		0.85	
% migrant students in	Less than 5%	1.22	*	1.20	٨
Wave 3	5%–9%	1.19		1.15	
	10%–19%	1.30	*	1.26	*
	20% or more	1.30	*	1.20	
Bottom 20% vocabulary at age 3				4.42	***
Constant		0.25	***	0.16	***
	Ν	8,437		8,437	
	Chi-square	1231.340		1689.060	
	Nagelkerke R-square	0.212		0.284	

TABLE 5.2LOGISTIC REGRESSION MODELS OF LOWEST QUINTILE OF VOCABULARY AT 5 YEARS
(BOTTOM QUINTILE) AND CHANGE OVER TIME (AGE 3–5) (ODDS RATIOS)

Source: GUI Cohort '08, Wave 3. Weighted by longitudinal weight.

Notes: p < 0.001, ** p < 0.01, * p < 0.05, ^ p < 0.1. Models 1 and 2 include controls for mother's education, family income, attributes of children (sex; SEN; whether in school by age 5; whether in centre-based care by age 3), and school characteristics (DEIS status; migrant prop). CoB is 'country of birth'; LP is 'lone parent'; BAME is 'Black, Asian and minority ethnic'.

5.6 MODELS OF ENGLISH READING AT AGE 9

5.6.1 Mean scores on English reading at age 9

Table 5.3 presents multiple regression models that investigate factors associated with English reading at age nine. As in Table 5.1, the model allows us to both estimate the effect of maternal region of birth, while also accounting for the child's linguistic and ethnic background, as well as other child, family and school characteristics that may be relevant for English reading scores and which may partly explain the differences observed in reading scores by maternal region of birth.

Model 1, which accounts for no other factors, shows that children whose mothers are from the UK, the US or other Anglophone countries have higher reading scores than the children of Irish mothers. But children whose mothers are from eastern Europe or Africa have slightly lower scores. Having a father from outside the UK or Ireland is also associated with lower reading scores. Having no secondary caregiver – that is, having just one parent – is associated with much lower scores. Further analysis of interactions (not shown) found that the effect of lone parenthood remained the same whether the mother was born in Ireland or abroad.⁴⁹

When we account for linguistic background in Model 2, we find that when all parents present are not native English speakers, children at nine years score five points lower on average than children where all parents are native English speakers. Once we account for language, the disadvantage for the children of eastern European and African mothers is much reduced and becomes not significant. This implies that linguistic background explains much of the difference in scores.⁵⁰ It is in sharp contrast to the results at age five, where even controlling for linguistic background, the children of eastern European mothers in particular still had much lower vocabulary scores.

Similar to the pattern at age five, Model 2 in Table 5.3 shows that children who have one native English-speaking parent do not differ from those with two native English-speaking parents. Model 2 also shows that children's ethnic background has no association with English reading at nine years when language and region of origin are accounted for.

⁴⁹ However, the effect of having a father present who did not complete the secondary caregiver survey (in Wave 1) was more negative for migrant mothers, suggesting non-completion may be tapping into poorer English-language skills of fathers.

⁵⁰ The decomposition analysis confirms that linguistic background and ethnicity accounts for the majority of the difference in scores of birth for mothers from eastern Europe, Asia, Africa and 'other' countries. These two variables also account for most of the negative effect of the father being born outside Ireland or the UK.

		Model 1	Model 2	Model 3	Model 4	Model 5	Model 6
Mother'S CoB	Mother – UK	1.2 ^	1.23 ^	1.54 *	1.62 *	1.58 *	1.44 *
(Ref. Ireland)	Mother – eastern Europe	-3.68 ***	-0.42	-1.38	-1.65	-1.35	2.7 *
	Mother – western Europe	0.37	1.12	-0.24	-0.81	-0.78	0.54
	Mother – Africa	-2.11 ^	0.28	0.06	-0.23	0.15	0.87
	Mother – Asia	0.06	3.34 ^	2.47	2.16	2.47	1.74
	Mother – US	6.24 **	6.25 **	4.10 *	4.01 *	3.88 *	3.92 *
	Mother – 'other'	2.73	5.33	5.58 ^	5.02	5.23 ^	4.99 ^
Father'S CoB (Ref. Ireland)	Father – UK	-0.19	-0.16	-0.48	-0.71	-0.7	-0.89
	Father – born outside IRL/UK	-2.27 **	-0.83	-065	-0.69	-0.61	0.49
	No father/Lone parent	-7.19 ***	-6.95 ***	-2.79 ***	-3.48 ***	-3.26 ***	-2.84 ***
	Father – info missing	-3.4 ***	-3.02 ***	-1.61 **	-1.49 *	-1.45 *	-1.22 *
Parental language	One native English		0.23	-0.19	-0.09	-0.09	0.04
(Ref. Both native Eng)	Both non-native English		-5.12 ***	-4.08 **	-4.13 **	-4.28 **	-0.95
Parental ethnicity	One BAME		-0.94	-0.65	-1.06	-1.09	-1.44
(Ref. Both White)	Both BAME		-0.06	0.03	-0.68	-0.62	0.32
Mat. educ.	Upper secondary			4.14 ***	3.58 ***	3.48 ***	2.98 ***
(Ref. Lower secondary of less)	Non-degree			6.46 ***	5.81 ***	5.81 ***	4.69 ***
	Degree			9.37 ***	8.69 ***	8.64 ***	7.24 ***
Family income	Low income* (Wave 1)			-3.14 ***	-2.88***	-2.62 ***	-1.99 ***
	Income missing			-2.03 **	-2.00 **	-1.66 **	-1.54 *
Child characteristics	Female				0.59	0.31	-0.04

TABLE 5.3 OLS REGRESSION MODELS OF READING SCORES AT AGE 9 AND CHANGE OVER TIME BETWEEN AGES 5 AND 9

TABLE 5.3	(CONTD.) OLS REGRESSION MODELS OF READING SCORES AT AGE 9 AND CHANGE OVER TIME BETWEEN AGE 5 AND AGE 9

		Model 1	Model 2	Model 3	Model 4	Model 5	Model 6
(Ref. No SEN)	Has SEN				-7.10 ***	-6.89 ***	-5.17 ***
(Ref. not in school by age 5)	In school by age 5					4.77 ***	4.14 ***
	Centre-based care age 3					0.84 *	0.84 *
DEIS status Wave 5	DEIS Urban 1					-2.06 **	-1.33 *
(Ref: non-disadvantaged)	DEIS Urban 2					0.96	1.03
	DEIS Rural					-0.25	-0.62
	DEIS status missing					0.26	0.47
Migrant prop. In school Wave 5	0.1< 5%					0.12	0.23
(Ref: no migrants in school*)	5-9%					0.8	0.6
	10-19%					-0.05	0.22
	> 20%					-0.87	-0.79
	Prop. Missing					-0.58	-0.26
T score for naming vocabulary (a	ge 5)						0.43 ***
Constant		101.76 ***	101.67 ***	96.71 ***	94.38 ***	94.52 ***	70.73 ***
	Ν	7,104	7,104	7,104	7,104	7,104	7,104
	Adjusted R square	0.040	0.042	0.104	0.150	0.153	0.248

Source: GUI Cohort '08, Wave 5. Weighted by longitudinal weight.

Notes: *** P < 0.001, ** p < 0.01, * p < 0.05, ^ p < 0.1. Family low income if in bottom two income quintiles. CoB is 'country of birth'; LP is 'lone parent'; BAME is 'Black, Asian and minority ethnic'; SEN is 'special educational needs'.

Model 3 shows how the children of mothers with a third-level degree have English reading scores over nine points higher than the children of mothers who left school early. Children from low-income backgrounds also have lower scores. Both of the findings are as expected, though do little to affect the differences between migrant groups, except that the reading score advantage for the children of US/Australian/Canadian mothers is somewhat reduced.

Model 4 shows that children with SEN have lower scores, as expected, and also that those who had started school at age five have higher scores in English reading, presumably as they have been exposed to school for a longer period. There is a small positive effect associated with having been in centre-based care on English reading at age nine. The fact that children with both parents born abroad are more likely to have started school by age five may be an advantage for them (see Figure 4.6). However, the fact that they are less likely to be in centre-based care at age three (Figure 4.4) is likely to be a disadvantage, particularly as previous research has found that children from a non-English-speaking background benefit more from centre-based care than children from an English-speaking background, at least in terms of vocabulary (McGinnity et al., 2015).

Model 5 shows that being in a disadvantaged school (Urban DEIS Band 1) is associated with lower English reading scores, even accounting for a range of other background characteristics. However, other school characteristics included in the model are not associated with English reading levels at age nine. In particular, the number of migrants in primary school does not affect English reading at nine years, though higher concentrations of migrant pupils in a school were found to be associated with vocabulary scores at age five (see Table 5.1). This may be because, as this model shows, differences in reading scores between migrant-origin and Irish-origin children are much reduced by age nine.

Finally, Model 6 adds scores at age five, to permit analysis of change over time. In general, adding the previous score reduces the effect of other characteristics or makes them statistically insignificant. An important exception here is the children of eastern European mothers. Here we find these children make more progress in English achievement than the children of Irish mothers. This is interesting and consistent with the overall finding that differences in English language achievement at age nine are much smaller than at age five.

5.6.2 Low reading scores at age 9

Table 5.4 presents the findings of a model estimating the chances of being in the lowest quintile for English reading at age nine. The groups of children and the estimation and interpretation of this model are the same as those in Table 5.2, but the findings are quite different. At age nine, none of the children whose mothers were born abroad are more likely to be in the lowest quintile of achievement in English reading than children with Irish mothers. In fact, the children of Asian

mothers are now less likely to be in the lowest quintile (less than half as likely). Having a father born outside Ireland or the UK is also not associated with being in this low vocabulary quintile; neither is having two non-native English-speaking parents. After adding the score at age five, we find that children with eastern European mothers are actually less likely to be in the lowest quintile than children with Irish mothers, suggesting they progress more over time.

TABLE 5.4LOGISTIC REGRESSION MODELS OF BEING IN THE LOWEST QUINTILE IN READING
SCORES AT 9 YEARS (BOTTOM QUINTILE) AND CHANGE OVER TIME (5–9) (ODDS
RATIOS)

		Model 1 at 9 years	Model 2: add score at 5
Maternal CoB	Mother – UK	0.87	0.88
(Ref. Ireland)	Mother – eastern Europe	0.88	0.56 *
	Mother – western Europe	1.16	0.92
	Mother – Africa	0.66	0.54
	Mother – Asia	0.43 *	0.42 ^
	Mother – US	0.61	0.69
	Mother – 'other'	0.68	0.61
Paternal CoB (Ref. Ireland)	Father – UK	1.28 ^	1.36 *
	Father – born outside IRL/UK	1.1	1
	No father/LP	1.34 **	1.31 *
	Father – info missing	1.67 ***	1.65 ***
Parental language	One native Eng	0.83	0.87
(Ref. Both native Eng)	Both non-native Eng	1.87	1.43
Parental ethnicity	One BAME	0.76	0.8
(Ref. Both White)	Both BAME	1.12	1.01
Bottom quintile vocabulary at age 5			3.66 ***
Constant		0.43 ***	0.32 ***
	Ν	7,104	7,104
	Chi-square	740.566	1023.141
	Nagelkerke R-square	0.157	0.212

Source: Notes: GUI Cohort '08, Wave 5. Weighted by longitudinal weight.

*** P < 0.001, ** p < 0.01, * p < 0.05, ^ p < 0.1. Models 1 and 2 include controls for mother's education, family income, attributes of children (sex; SEN; whether in school by age 5; whether in centre-based care by age 3), and school characteristics (DEIS status; migrant prop). CoB is 'country of birth'; LP is 'lone parent'; BAME is 'Black, Asian and minority ethnic'.

5.7 SUMMARY

Language proficiency is a key driver of immigrant integration, as discussed in Chapter 2. Cultural and linguistic distance plays a role in the acquisition of the language of the host country, one that may impact on the outcomes of migrantorigin children (Melkonian, et al., 2019; Isphording, 2014). Furthermore, language development is shaped by a range of background and school-level factors (De Feyter et al., 2009; Dustmann et al., 2012; Baysu et al., 2018).

The primary focus of this chapter was on English language achievement at ages five and nine. It examined how this factor differed depending on maternal country of birth, and linguistic and ethnic background, and how it changed over time. As noted at the outset, these are not measures of overall linguistic ability, but of performance in English expressive vocabulary and in English reading.

At age five, there are some clear disadvantages for some second-generation migrant children, with lower scores found particularly among children whose mothers are from eastern Europe, but also western Europe, Africa and Asia. These differences are considerably reduced when we account for children's linguistic background in statistical models, though the lower scores for children from eastern and western Europe remain even after controlling for many background characteristics, and these children are more likely to be in the lowest vocabulary quintile at age five than Irish children. Children whose mothers come from the UK or other Anglophone countries (Australia, Canada, New Zealand and the US) do not differ from the children of Irish mothers in expressive vocabulary scores in English at age five, as expected.

A key finding of this chapter is that by age nine, the penalty in English language achievement for these same migrant-origin children is much reduced. We observe dramatic falls in the proportion of children with both parents born abroad or nonnative English-speaking who are in the lowest quintile of English language achievement (Figures 5.4 and 5.5). In the models (Tables 5.3 and 5.4), we find children with two non-native English-speaking parents have somewhat lower reading scores at age nine, but once we control for this there is no evidence of differences by country of birth (except an advantage for the children of mothers from Anglophone countries). The children of eastern European mothers actually make somewhat more progress between ages five and nine than the children of Irish mothers. As these are the same children in all three waves of the survey, we know this is not due to compositional differences between the samples. The tests of English language administered at age five and age nine are age appropriate if somewhat different from each other, but as English vocabulary is a prerequisite for reading comprehension, it seems appropriate to compare them. Note that all through this chapter, the children of Irish mothers are used as a 'benchmark' as the key question in this report is whether the development of second-generation migrant children differs from that of the children of Irish parents.

Children without a native English-speaking parent are clearly at a disadvantage in terms of English reading, as shown in both descriptive charts and models presented in this chapter, though the gap is lower at age nine than at age five. However, another important finding is that children who have one native English-speaking parent and one non-native English-speaking parent do not differ from children where both parents are native English speakers. A similar pattern is found for those who have one Irish parent and one foreign-born parent. It seems that having one English-speaking parent is enough to facilitate English language development, at ages five and nine.

Children who have two parents from an ethnic minority background have lower scores at age five and somewhat lower scores at age nine to children from a White background. However, when we account for maternal region of origin and language background, there is no additional penalty.

One point to note is that the focus of this chapter is on expressive vocabulary and English reading, because this is very important for subsequent educational achievement and peer relationships. It is also where we might expect the greatest differences between the second generation and the children of Irish parents. However, it is only one element of cognitive skills and achievement. Indeed, Programme for International Student Assessment (PISA) scores at age 15 consistently show that the mathematics and science achievement of children from non-English-speaking backgrounds do not tend to differ from those for children from English-speaking backgrounds. (A difference is found between the two groups regarding English reading scores at 15 years.) Similarly, McGinnity et al. (2015) find a large disadvantage in vocabulary for five-year-old children from non-English-speaking backgrounds but a much smaller difference in non-verbal reasoning scores between children from an English-speaking and a non-English-speaking background (*ibid*.).⁵¹

Cognitive development and achievement in school are important aspects of childhood development and integration, but they are not the only ones. The chapter that follows considers the socio-emotional development of second-generation migrant children in order to give a more comprehensive picture of their development and how they are integrating into Irish society.

⁵¹ The non-verbal reasoning test used pictures and only required English to understand the instructions. It is a measure of reasoning ability (see McGinnity et al., 2015, Chapter 1 for further details of this test). This study does not account for maternal country of birth or ethnic background.

CHAPTER 6

Child self-concept

6.1 INTRODUCTION

In this chapter, we broaden the examination of second-generation migrant children's outcomes to encompass their social and psychological wellbeing. We draw on a measure of child self-concept (also termed self-image) as reported by the study children themselves when they were nine years of age. This covers aspects such as their concept of their intellectual and physical selves, as well as their happiness and popularity. As noted in Chapter 1, previous studies have used wellbeing as an indicator of migrant integration in the host country (see, for example, Safi, 2010; Johnsson and Mood, 2018). Self-concept as part of a sense of belonging has also been considered important in migrant integration (Wu et al., 2012).

While the challenges of adaptation and acculturation for migrant families may lead us to expect wellbeing in general, including self-concept, to be lower for migrantorigin children, findings to date on the topic are mixed (Belhadj-Kouider et al., 2014, 2015). Indeed, some studies even show migrant children have a higher sense of mental wellbeing than the majority population (Jonsson and Mood, 2018).

Social interaction is an important correlate of migrant integration and wellbeing, fostering a sense of belonging in the wider society (Gsir, 2014). Participation in sports and social activities (Darmody and Smyth, 2017) is an important means of forming friendships for first-generation migrants. Therefore, we are interested in the role of participation in sports and social/cultural activity as predictors of positive self-concept among second-generation migrant children.

Previous research in Ireland and elsewhere has shown that migrant-origin children engage with sports and cultural activities in different ways (Hertting and Karlefors, 2013), with somewhat lower participation levels in organised sports or structured cultural activities than those found for their native counterparts (Darmody and Smyth, 2017). Participation can also vary by national groups: research in Ireland, based on the *Growing Up in Ireland* (GUI) Cohort '98, found children from eastern Europe, Asia and Africa are less likely to participate in structured sporting activities and that this was only partly explained by economic resources (Darmody et al., 2016).

The characteristics of schools have previously been found to be important for the wellbeing and self-concept of children in middle childhood (Smyth, 2015). It has also been shown that this factor can influence the sense of belonging and inclusion among first-generation migrant children (Devine, 2013; Harte, et al., 2016).

Wellbeing of migrant-origin students is found to be influenced not only by their region of origin, but also by how well schools in the receiving country support them in overcoming various difficulties associated with integrating and succeeding at school (OECD, 2015). In line with earlier research in Ireland (see Smyth et al., 2009), we saw in Chapter 4 that there is a greater concentration of students with a migrant background in disadvantaged schools, which may have a negative effect on their social outcomes. Previous research has also found that principals of Delivering Equality of Opportunity in Schools (DEIS) schools, at both primary and post-primary level, report a higher prevalence of emotional and behavioural difficulties among students (Smyth et al., 2015).

It may have a positive effect on the wellbeing of individual migrant students to attend a school with a higher proportion of migrant students, if this enhances their sense of belonging in the school community. For example, in schools with high levels of migrant concentration, migrant-origin children may experience an enhanced sense of belonging due to greater opportunities to form positive relationships that may mitigate against the risk of bullying (Vitoroulis and Georgiades, 2017). The Programme for International Student Assessment (PISA, 2018) shows that in some OECD countries, namely Australia, Norway and the UK, both migrant and non-migrant students' sense of belonging was stronger in schools with a higher concentration of migrant students. Previous research findings regarding school principals in Ireland (Smyth et al., 2009) suggest that the presence of migrant children in a school has a positive spill-over effect for non-migrant children in terms of motivation and behaviour.

6.2 SELF-CONCEPT SCORES

We examine the total self-concept scores among children with migrant backgrounds and those with Irish-born parents. The scale ranges from three to 31. Taking the maternal country of birth, we see that children of mothers from eastern Europe, western Europe, the US/Canada/Australia and Asia have marginally lower total self-concept scores than those with mothers from Ireland. Children of mothers from the UK and Africa have the same scores as children with Irish-born mothers.

Taking the background of both parents into account, we see that where both parents are born abroad children have a somewhat lower self-concept score than when one or both parents are born in Ireland (p<.001).

The raw difference in self-concept between children where both parents are from an ethnic minority background and those where both are White is just over one point on the scale, which is one-quarter of a standard deviation (see Figure 6.1). By contrast, those with a mixed ethnic background (one White and one Black, Asian or minority ethnic (BAME) parent) have a somewhat higher score than families where both parents are White.⁵² The same pattern is observed based on the linguistic background of parents.



FIGURE 6.1 MEAN TOTAL SELF-CONCEPT SCORES OF 9 YEAR OLDS BY COUNTRY OF BIRTH, ETHNICITY AND LANGUAGE BACKGROUND OF PARENTS

Source: GUI Cohort '08 Wave 5. Weighted by longitudinal weight.

Notes: Self-concept measured by Piers-Harris Scale. Scale ranges from three to 31. 'Both' includes lone parent families. CoB stands for 'country of birth'; BAME stands for 'Black, Asian and minority ethnic'.

As noted in Chapter 3, the self-concept score is made up of six subscales: popularity, freedom from physical happiness, anxiety, self-image, intellectual/educational self-image and behavioural self-image (see Chapter 3 for further details). All subscales are scored in a way that means higher scores indicate a more positive self-concept. The subscales have a different number of items, so the focus here is on the relativities within each subscale rather than differences between scales. Figure 6.2 shows that the differences in subscale scores between migrant-background children and those from an Irish background are relatively minor. This illustrates that the total score is not disguising wider differences in particular subscales between second-generation migrant and Irish-origin children.

⁵² P<.05 (t-test). Some of the advantage in raw scores of the mixed group may arise because every child in this group is in a two-parent family, whereas the other two groups include lone parent families.



FIGURE 6.2 SELF-CONCEPT SUBSCALE SCORES OF 9 YEAR OLDS BY PARENTAL COUNTRY OF BIRTH

Source: GUI Cohort '08 Wave 5. Weighted by longitudinal weight.

Notes: Self-concept measured by Piers-Harris Scale. 'Both' includes lone parent families. 'Born abroad' includes UK-born.

6.3 PARTICIPATION IN SPORT, SOCIAL AND CULTURAL ACTIVITIES

Participation in team sports is significantly lower among children with a migrant background (48 per cent) compared to children with Irish parents (69 per cent), a finding that is in line with earlier studies (Darmody and Smyth, 2017; Coughlan et al., 2014).⁵³ Participation among children with mixed-migrant and Irish parentage (65 per cent) is more similar to those with only Irish parents.⁵⁴ Where both parents are migrants, children are also more unlikely to participate in individual-level sports.

⁵³ The question refers to activities outside of school hours, though it does not specify if the activity takes place at the school. Respondents are also asked if the activity has to be paid for but we did not use this information. Those at DEIS schools were less likely to participate in individual and team sport than those in non-DEIS schools. Further analysis would be necessary to distinguish the influence of school-level and family-level influences.

⁵⁴ This group includes those with a lone Irish parent or two Irish parents in couple households.



FIGURE 6.3 PARTICIPATION IN SPORTS AT AGE 9 BY MIGRANT BACKGROUND (PARENTS BORN ABROAD)

Source: GUI Cohort '08 Wave 5. Weighted by longitudinal weight.

Note: Self-concept measured by Piers-Harris Scale. 'Both' includes lone parent families. 'Born abroad' includes UK-born.

Children's participation in eight social and cultural activities is also assessed: drama classes; arts/crafts, computer/technology, youth clubs, religious clubs or groups, music or dance, scouts, and language classes.⁵⁵ We count the number of activities but we do not have information on frequency of attendance. Children with a migrant background (both parents) have a somewhat lower level of participation than those with Irish-born parents; however, those with one parent born in Ireland and one born abroad have the highest level of participation.⁵⁶

⁵⁵ We exclude homework classes as these are judged to be an educational activity rather than social or cultural, though the results are not affected if we include this activity.

⁵⁶ T-tests show that both groups are significantly different to those with two Irish-born parents.



FIGURE 6.4 MEAN NUMBER OF SOCIAL/CULTURAL ACTIVITIES AT AGE 9 BY MATERNAL COUNTRY OF BIRTH AND PARENTS BORN ABROAD

Source:GUI Cohort '08 Wave 5. Weighted by longitudinal weight.Notes:Self-concept measured by Piers-Harris Scale. 'Both' includes lone parent families. 'Born abroad' includes UK-born.

6.4 MODELS

As with cognitive outcomes, self-concept is analysed through a series of nested models. Model 1 presents differences across second-generation groups based on parental country of origin. Model 2 and Model 3 examine whether these initial differences are related to language background or ethnicity. We then investigate whether any remaining differences are explained by socio-economic conditions, individual-level characteristics, such as gender, special educational need (SEN) and participation in sport and social/cultural activities. Finally, Model 5 assesses the influence of school characteristics, namely DEIS status, which indicates whether or not a school is socially disadvantaged, and the proportion of migrants in the student population.

In Model 1, children with a mother from eastern Europe, western Europe or Asia have significantly lower self-concept scores. Paternal country of birth does not have an additional effect. However, where the father is not co-resident, children have a significantly lower score. We checked whether the effect of lone parenthood differed by whether the mother was born abroad (not shown), and found that, without controls, the effect of a non-resident father is slightly lower for migrant mothers, but that when the full set of controls were included there is no significant interaction. This shows that the effect is the same for each group. Model 2 indicates that where one parent is a native English speaker, there is no effect on self-concept, but where both parents are not native English speakers, child self-concept is reduced. Once parental language is controlled for, those with a mother from Asia or eastern Europe are no longer significantly different from those with an Irish-born mother. However, the decomposition analysis (available on request from authors) shows that the contribution of the linguistic factor, in terms of explaining mother and father country-of-birth coefficients, is not significant.

Ethnic minority background does not have a significant influence on child selfconcept (Model 3), but when it is added to the model, children with an African mother have a more positive self-concept compared to the Irish reference group. The decomposition analysis confirms that the ethnic background controls are not significant in accounting for country-of-birth differences. The lower self-concept of children with a western European background persists despite language and ethnicity controls.

Socio-economic conditions are significantly related to self-concept (Model 4) but do not alter the parental country-of-birth effects. However, the 'both non-native speakers' effect is now insignificant, suggesting that this effect partially works through lower socio-economic resources among these families. This is consistent with findings from McGinnity et al. (2020b) that show among adult migrants, lower self-rated English language skills are associated with higher unemployment rates and lower occupational attainment when in work in Ireland. The decomposition analysis further shows that socio-economic status (SES) factors contribute positively to the self-concept scores of children whose mother is from western Europe but that this is not enough to counteract other (mostly unobserved) influences that reduce self-concept.

Model 5 shows that girls have a more positive self-concept than boys at age nine, but that children with SEN have a significantly more negative self-concept. In this study, participation in team sport has a significant positive association with child self-concept, while involvement in individual sports or social and cultural activities has no such effect.⁵⁷ The decomposition analysis shows that child characteristics – gender, SEN and participation in sports and social/cultural activities – contribute most to explaining the lower scores of the eastern European mother group and the Asian mother group. Indeed, these characteristics are also associated with a significant lowering of the score for children with a mother from Africa, the UK and western Europea.

⁵⁷ An earlier study by Smyth (2015) based on the GUI Cohort '98 indicates that participation in sport appears to enhance a sense of belonging and fulfilment.

Model 6 adds a set of institutional variables to the picture. It finds that age when starting school and participation in formal childcare at age three do not influence child self-concept, and that neither do they influence the country-of-birth effect. Similarly, attending a disadvantaged school does not influence self-concept; neither does the proportion of migrants in a school's student population.⁵⁸ Given that the sample is not clustered within school, the analysis cannot rule out the possibility that such school-level processes would be discernible with an alternative research design. Overall, the model explains a relatively low proportion of variance (6 per cent), suggesting that there are other unobserved factors at play; for example, personality type.

⁵⁸ There is some indicative evidence that a higher proportion of migrants in a school population has a positive effect on the self-concept of second-generation students; however, because of the relatively small number of respondents in schools with a high proportion of migrant students, the results are not statistically significant.

TABLE 6.3 OLS REGRESSION OF SELF-CONCEPT SCORES AT AGE	9
--	---

		Model 1 – CoB	Model 2 – +language	Model 3 – +BAME	Model 4 – +SES	Model 5 – +child	Model 6 – +school
Mother's	UK	-0.10	-0.09	-0.09	-0.07	-0.05	-0.05
CoB Ref. Ireland)	Eastern Europe	-0.69 *	-0.13	-0.27	-0.41	-0.22	-0.22
	Western Europe	-1.48 **	-1.45 **	-1.49 **	-1.62 **	-1.47 **	-1.48 **
	Africa	0.40	0.79 *	1.23 *	1.15 *	1.26 *	1.25 *
	Asia	-0.92 *	-0.41	0	-0.18	0.06	0.05
	US, Aus etc.	-0.85	-0.84	-0.84	-1.00 ^	-1.03 ^	-1.05 ^
	'Other'	-1.19	-0.79	-0.71	-0.78	-0.47	-0.42
Fathers CoB	UK	-0.06	-0.05	-0.06	-0.04	-0.01	-0.01
(Ref. Ireland)	Born outside IRL/UK	-0.11	0.15	0.18	0.23	0.23	0.22
	Not-resident/Lone parent	-1.58 ***	-1.53 ***	-1.52 ***	-1.03 ***	-0.83 ***	-0.86 ***
	Info. missing	-0.01	0.08	0.08	0.21	0.18	0.17
Parental language	One native English		0.28	0.28	0.3	0.26	0.26
(Ref. Both Eng)	Both non-native English		-0.94 *	-0.79 ^	-0.63	-0.64	-0.64
Parental ethnicity	One BAME			-0.08	-0.06	-0.01	-0.03
(Ref. Both White)	Both BAME			-0.72	-0.61	-0.72	-0.71
Mother's educ.	Upper secondary				0.85 ***	0.67 ***	0.69 ***
(Ref. lower second)	Non-degree				0.81 ***	0.57 **	0.61 ***
	Degree				1.04 ***	0.78 ***	0.82 ***
Family income	Low income* (9 months)				-0.13	-0.07	-0.07
	Low income* (9 years)				-0.45 ***	-0.37 **	-0.39 **
Child characteristics							
Male (ref)	Female					0.50 ***	0.51 ***

		Model 1 – CoB	Model 2 – +language	Model 3 – +BAME	Model 4 – +SES	Model 5 – +child	Model 6 – +school
Ref. No SEN	Has SEN					-1.53 ***	-1.54 ***
Ref. =No	Team sport (9 years)					0.90 ***	0.89 ***
Ref. =No	Individual sport (9 years)					-0.06	-0.09
Scale	N social and cultural activities (9 years)					0.06	0.04
Institutional							
Ref. =No	In school by 5 years						-0.16
Ref. =No	Centre-based care at 3 years						-0.10
School DEIS status	DEIS Urban 1						0.30
(Ref: Non-	DEIS Urban 2						0.04
disadvantaged)	DEIS Rural						0.23
% Migrant students	0.1< 5%						0.03
(Ref: No migrants in	5%-9%						-0.25
school*)	10%-19%						0.15
	> 20%						-0.03
Constant		27.3 ***	27.27 ***	27.27 ***	26.61 ***	26.06***	26.16 ***
	Ν	6680	6680	6680	6680	6680	6680
	Adjusted R-square	0.019	0.021	0.021	0.032	0.061	0.060

TABLE 6.3 (CONTD.) OLS REGRESSION OF SELF-CONCEPT SCORES AT AGE 9

Source: GUI Cohort '08 Wave 5. Weighted by longitudinal weight.

Notes: *** P < 0.001, ** p < 0.01, * p < 0.01, * p < 0.05; ^ p < 0.1. Self-concept measured by Piers-Harris Scale. Family low income if in bottom two income quintiles. CoB is 'country of birth'; LP is 'lone parent'; BAME is 'Black, Asian and minority ethnic'; SEN is 'special educational needs'.

6.5 SUMMARY

As discussed in Chapter 2, migrant integration is influenced by a range of factors, including country of origin, individual characteristics, experiences in the host country and structural characteristics. The level of integration of migrant-origin children varies across OECD countries, ranging from low integration (for example, France and Belgium) to high integration (for example, Australia) (OECD, 2012). This study has shown that in Ireland the children of migrants have self-concept scores, which incorporates a child's sense of their intellectual, educational, physical, emotional and social selves, that are generally on a par with those of children whose parents were born in Ireland. The differences in self-concept between those with mothers and/or fathers born outside Ireland are relatively small, and children with an African mother fare better than those with Irish-born parents.

Lower self-concept is observed for some groups. We find that, of the explanatory factors investigated, the child characteristics of gender, SEN and participation in sports and social/cultural activities play the biggest role in accounting for parental country-of-birth characteristics. However, the lower self-concept levels found among children whose mothers are from western Europe and the US/Canada/Australia are not explained by the models and are puzzling because these are socially and economically advantaged groups. This indicates that unobservable characteristics, such as abilities, ambition and resiliency (Rooth, 2007; Feliciano, 2020) may also play a role. A meta-analysis of cross-cultural differences in self-concept among children (Wastlund et al., 2001), albeit using a different measure, found that children from Africa had the highest self-concept, those from Asia had the lowest self-concept on non-academic dimensions and those from America/Australia had the lowest academic self-concept. Our results suggest that cultural differences persist for the second generation who are living in the same country but with parents from different backgrounds. In addition, we find that both parents being non-native English speakers has a negative influence on self-concept, and that this in turn is related to the lower economic resources of these households.

According to integration context theory, migrant integration is strongly influenced by contextual differences in terms of where integration takes place, such as institutional arrangements in schools (Crul and Schneider, 2010). No significant school effects were found for self-concept; however, further research focusing on different outcomes or on samples clustered within schools might identify such differences. International research suggests that there are positive ways of promoting the value of diversity through the curriculum and learning materials, by promoting cultural competence among teachers and other students as well as by creating opportunities for common activities that encourage links between students (Darmody and Smyth, 2015). Participation in team sports in particular is associated with higher self-concept among children, regardless of their national, ethnic or cultural background; however, children in families where both parents were born abroad are less likely to participate. Therefore, greater supports for participation both inside and outside the school setting are likely to be beneficial. However, it should be noted that the direction of causality in the relationship between sports participation and self-concept cannot be established in the current analysis and it is possible that those with higher self-concept scores select into these activities. Disentangling these processes is possible with the GUI Cohort '98 as self-concept is measured in multiple waves.

While self-concept is a well-validated and multi-dimensional measure of how children are faring, it is nevertheless just one measure of child wellbeing and there may be group-level differences in terms of other aspects of social integration, such as peer relationship and bullying, which could be explored in future research, using data from the GUI.

CHAPTER 7

Conclusion

7.1 SUMMARY OF FINDINGS

In Ireland, it is unclear whether differences in outcomes between native and migrant-origin children exist at a very young age, growing more stable thereafter, or whether they increase or decrease over time. It is also unclear whether such patterns are the same across different types of outcomes (such as cognitive and socio-emotional outcomes) and different national groups. Recent years have seen a growing interest in these topics, however; for a recent contribution to the evidence, see Sprong and Skopek (2021). This is the context for the present study, which seeks to address this gap in Irish research by exploring a range of cognitive and socio-emotional outcomes of migrant-origin children at ages three, five and nine years, which span an important period for both skill development and laying the foundations of future learning.

Language development is measured using English expressive vocabulary tests, which were carried out at ages three and five, and reading ability tests at nine years. Child self-concept is measured at nine years using a shortened version of the Piers-Harris Self-Concept Scale, which the children completed themselves. English language ability is important for school achievement and integration into the labour market: it is also important for peer relationships. Self-concept is a useful indicator of child socio-emotional wellbeing. Previous research has indicated that both domains are essential to the learning processes of a child (Suárez-Orozco, 2018), and thus should be considered together.

We consider second-generation migrant-origin children to be children of parents born outside Ireland, including parents born in the UK. Using this definition, we find that 14 per cent of children have one parent born abroad and one born in Ireland while a further 19 per cent have both parents (or lone parent) born abroad.

On the basis of previous research and theories concerning migrant integration, cultural distance and linguistic difference, we present children's English-language and self-concept scores, distinguished by the country of origin, linguistic background and ethnic background of parents. We then use statistical modelling to explore the role of these three factors in understanding child outcomes. First-generation migrants may often be economically disadvantaged (Jonsson et al., 2018); therefore, the economic and social resources of parents are also considered as a possible mediating factor (Suárez-Orozco, 2018; OECD, 2017). Participation in early care and education and certain characteristics of schools attended by migrant-origin children have also been shown to influence the development of this

group; for this reason, these factors are also considered in the statistical models of English language ability and self-concept at age nine.

A consistent finding across both the cognitive (language/reading) and self-concept outcomes is that children with one parent born in Ireland and one born abroad do not differ from children whose parents were both born in Ireland. Indeed, on some intermediary measures the latter group fares slightly better than the Irish-origin children; for example, these children are less likely to be in the low-income group and more likely to participate in social and cultural activities.

There is a strong upward trajectory in the English-language attainment of migrantorigin children over time. The proportion of children with both parents born abroad who are reported to have English as their first language increases from 53 per cent at age three to 72 per cent at age nine. At age three, 60 per cent of children with both parents born abroad are in the bottom quintile of English language/reading achievement, compared to 20 per cent of the whole sample. Yet by age nine, only 28 per cent of children with both parents born abroad are in this lowest quintile, compared to 20 per cent of the full sample. This indicates a positive integration in early to middle childhood for second-generation migrant-origin children in terms of English language achievement. It is likely due to a combination of factors - from learning in school, including additional language learning, to interactions with peers and others outside school, to encouragement from their parents. This may be allied with migrant 'language adoption', whereby migrant parents may encourage the learning of the dominant language. Previous research has identified a tendency among migrants to focus on the acquisition of the (official/high status) languages spoken in the receiving country, at the expense of heritage languages (Chaudhry and Zeeshan, 2019). Whatever factors underlie this finding, the acquisition of English language skills by this cohort is likely to facilitate both current and future learning in the Irish education system by secondgeneration migrant children, as well as social interaction with their peers.

Chapter 2 has outlined the challenges encountered by migrants in receiving countries as well as the background and structural factors that impact the social and academic integration of children of migrants. Notwithstanding the progress made, this report shows that there are some clear and persisting disadvantages for some second-generation migrant children in Ireland. Children of a mother from eastern Europe, western Europe, Africa and Asia were found to have lower vocabulary scores at five years and somewhat lower reading scores at nine years, compared to children of Irish-born mothers. These differences are considerably reduced when we account for child linguistic background in statistical models, though the lower scores remain, particularly for children from eastern but also western Europe, even after controlling for many background characteristics. These children are more likely to be in the lowest vocabulary quintile at age five compared to their native counterparts. Children in families without a native

Conclusion 85

English-speaking parent continue to be disadvantaged in their reading scores at age nine. The effect size is not large (-4.3 points on a 100 point scale, just one-third of a standard deviation), comparing to an effect size of +8.6 for those whose mother has a degree. The finding indicates the importance of language background for academic outcomes, in line with earlier work by Darmody and Smyth (2018). The model of reading at age nine that included the chances of being in the lowest quintile of vocabulary at age five suggests that migrant-origin children, particularly those from eastern Europe, are much less likely to be in this lowest quintile of reading at age nine, suggesting greater progression over time for these groups.

Turning to child self-concept, we find that linguistic and economic resources play a role in explaining some differences in the self-concept of migrant-origin and Irishorigin children, consistent with previous research in the US and Europe (Rumbault 1994; Crul, 2007). Overall, the self-concept scores of second-generation migrant children were somewhat lower than Irish-origin children. However, the selfconcept scores of children with mothers from eastern Europe and Asia were no longer different to those with Irish mothers when parental linguistic background and socio-economic status (SES) were taken into account. Children whose mothers were from western Europe and the US, however, continued to have lower scores even when a wide range of individual, family and school characteristics were taken into account. This may be due to unobservable characteristics of these groups. Children with African mothers had somewhat higher self-concept scores than those with Irish-born mothers. This may be due to a number of factors. One of these is the migrant selection effect as discussed in Chapter 2, whereby highly motivated migrants are more likely to migrate, compared to others from the same country of origin. Religiosity, the nature of family ties and social networks may also play a role: some authors have suggested that migrant children and young people might also compare their situation with peers or family in the country of origin, who did not migrate, rather than peers in their host country (Jonsson and Mood, 2018).

Ethnic background did not play a significant role in self-concept when parental country of birth and language were held constant. Other research has found that the experience of racism has a significantly negative impact on self-concept among other age groups, as well as wellbeing and mental health among adults, including young adults (Paradies et al., 2015; Fibbi et al., 2021). It is therefore important that ethnic differences in self-concept continue to be monitored over time, and that the future trajectories for these groups are observed as they reach adolescence. Previous research suggests that children's recognition of discriminatory acts is evident by age ten (McKown and Weinstein, 2003). However, child awareness of prejudices and in-group biases may increase as they further develop their social skills and increase independent contact with a wider group of people outside of family, neighbours and classmates. Cohort '98 of the *Growing Up in Ireland* (GUI) study are asked about their experience of bullying related to their race, nationality

or ethnicity, as well as their experience of discrimination. It is important that such issues are captured in future waves of Cohort '08.

7.2 POLICY IMPLICATIONS

There is evidence of good progression in English language proficiency among young children who were born in Ireland to migrant, non-native English-speaking parents; however, gaps remain at nine years of age. What steps might be taken to narrow gaps in the English language/reading achievement of children? Several studies have suggested targeted measures should be implemented to assist this group of young people, ranging from early education programmes and cognitive and language training to parent and teacher training. A review undertaken by Beelmann et al. (2020) has revealed strong effects of child cognitive and language training programmes on child academic and language outcomes and relatively low effects of all programmes on child socio-emotional outcomes. The authors argue that individualised and culturally tailored programmes seem to be more effective. They also question the potential of individual psycho-social and educational programmes to counterbalance the multifaceted risks of immigration.

Cavallo and Russo (2020) highlight the need to promote language integration at a young age and to invest in the language acquisition of first-generation migrants. Rodriguez and Darmody (2017) note how migrant parents who lack English proficiency may experience considerable difficulties negotiating their way through the education system and establishing themselves as partners in the home–school interface. The lack of a coordinated approach to English language provision for adult learners in Ireland has been raised as a policy issue for a number of years, with need for language learning far outstripping supply (McGinnity et al., 2020a). Parental English language proficiency is likely to have positive spill-over effects for children (Driessen, 2017).

The results of this study have implications for early education and care policy. We find that while there was a high level of take up of the free preschool year (the ECCE scheme) for children with two parents who were born abroad, these children were somewhat less likely to have participated in the programme than Irish-origin children. We also found that, while participation in centre-based childcare at age three (before ECCE) was not associated with vocabulary levels at age five for all children, it was associated with higher English reading skills at age nine, even when vocabulary scores at age five are included. Previous research identified a 'compensatory effect', in that the language benefit of centre-based care was greater for children from a non-English-speaking background (McGinnity et al., 2015). Taken together, this evidence suggests that early education investments benefit children of both migrant and non-migrant backgrounds, and are likely to benefit migrant-origin children even more. Since this cohort of children were of pre-school age (in the period 2008–2013), there has been a major reform of pre-school childcare policy with the introduction of the National Childcare Scheme

(Russell et al., 2018). Recent estimates using the ESRI's tax-benefit model (SWITCH) indicate that families in the bottom third of the income distribution will benefit most from the childcare subsidy introduced, with this group experiencing the greatest fall in childcare costs (Callan et al., 2021). This may facilitate more low-income migrant families to access formal pre-school childcare than has been the case for this cohort, and thereby enable more migrant mothers to return to work after childbirth, if they wish, even if their earnings are low. This is important given that maternal employment can act as a protective factor against poverty and economic vulnerability, particularly among low-income families (Maître et al., 2021). The streamlined nature of the National Childcare Scheme, designed to make the system clearer and more straightforward, may also be an advantage for migrant parents in navigating the complex system of childcare provision in Ireland.

As noted in Chapter 1, some countries provide language screening as well as, where necessary, language support through pre-school (OECD, 2021), and some evaluations have indicated that this approach reaps large benefits in terms of language proficiency (Leseman et al., 2017). Consideration could be given to additional targeted language supports at pre-school drawing on lessons from other jurisdictions.

According to context theory, migrant integration is strongly influenced by differences in the contexts in which integration takes place, such as schools. School effects in the current study are relatively modest, though it should be noted that there was typically only a small number of pupils from our sample in each school.

We find that a higher concentration of migrant students in the school is associated with a lower score on the English vocabulary test at age five, independent of the child's own and family characteristics. This effect was the same for children with a migrant background and those without. The size of the effect is small and disappears by age nine. Nevertheless, it suggests there may be a need for additional supports for schools with a high number of students with migrant backgrounds, especially for the younger classes.

We find a small negative effect on English language achievement, at ages five and nine, for those attending the most disadvantaged schools (Delivering Equality of Opportunity in Schools (DEIS) Urban Band 1) over and above the children's own economic or social disadvantage – a reduction of minus two on a 100-point scale at age nine. Moreover, progression is lower between five and nine years for pupils in these schools. At age five, we find that this negative effect is greater for students with migrant parents than those with Irish-born parents. By age nine, however, no difference is found in the 'DEIS Urban Band 1 effect' between migrant-origin and Irish-origin children. Greater investment in schools with the most disadvantaged student populations would be of benefit for all children attending. However, given that there is also a higher concentration of migrant-origin children in DEIS schools

(in line with the earlier study by Smyth et al., 2009), further support for the most disadvantaged schools would assist with the narrowing of the achievement gap. Regarding self-concept, we found that children in DEIS schools fare as well as those in non-DEIS schools.

School admissions policies may not intentionally disadvantage migrant families, but a number of factors lead to migrant children in this cohort being concentrated in disadvantaged schools. The impact, if any, of the introduction of the Education (Admission to Schools) Act 2018 on migrant children's allocation to schools will be interesting in this regard. Cultural diversity of a school population will of course also be determined by the diversity of the local population/school catchment area, which cannot be manipulated by school policy.

While significant progress is made between ages three and nine in terms of the English language skills of children with two parents born abroad, these children still have lower language skills at age nine, on average, than Irish-origin children. As discussed in Chapter 1, there is no standardised system of English language assessment in place in Irish primary schools and there is likely to be considerable variation in how schools assign resources to English language tuition and special educational needs (SEN), though the funding model means it is not possible to assess this. Are learning supports for English as an additional language (EAL) being effectively directed towards children who need it? Monitoring supports offered at school level and their effectiveness via English-language assessments of participating children could usefully inform policy in the area. For children in Cohort '08, further research could investigate reported participation in language support classes and whether this is linked to English language development between ages five and nine.⁵⁹

A recurring theme in educational policy is the importance of parental engagement. Yet if parents lack the requisite language skills, home–school liaison may be negatively affected. In addition, parents with poor English language skills may not be able to support their children with their learning or homework. Facilitating adult acquisition of English language skills is recognised as an action in the Migrant Integration Strategy (2017–2021). However, there have been some implementation issues (Department of Justice and Equality, 2019), and it is important that this is prioritised in the successor strategy. Supporting English language skills through effective policy provision will reap benefits, not just for the integration of adult migrants (see McGinnity et al., 2020a), but also the integration of their children into Irish society.

⁵⁹ In the teacher surveys at Wave 3 and Wave 5, respondents are asked whether the study child has received support for English as an additional language (if the teacher has answered yes to the child having any difficulty that limits their learning).

7.3 LIMITATIONS OF STUDY AND FUTURE RESEARCH

As noted above, the integration of children from a migrant background not only occurs in family, childcare centres and schools, but also in local neighbourhoods and communities. Embeddedness in a local neighbourhood that is welcoming and accepting of migrant families (see Fahey et al., 2019) can further support the socioemotional development of migrant-origin children and young people. Matching information on local areas to the GUI would allow much greater insights into how the composition of a local community influences the experiences of children and young people from migrant backgrounds.

Children of parents that came to Ireland as refugees, and who may have spent time in direct provision, are likely to require additional supports (Ombudsman for Children's Office, 2020). It is not possible to identify children from a refugee background in the GUI study, but future research could focus on the wellbeing and educational needs of this group.

An interesting finding from this report is that, in terms of both English language proficiency and self-concept, children with one migrant parent or one English-speaking parent do not typically differ from those with two Irish-origin or native-English-speaking parents, even after accounting for background characteristics. We know relatively little in Ireland about recent patterns of intermarriage by migrant status or linguistic background.⁶⁰ Future research could consider the nature and patterns of these 'mixed' partnerships in Ireland.

In the current study, we focus on self-concept as reported by the children themselves, as this provides an important insight into their self-esteem, happiness and sense of belonging in school and among peers. However, while this is a multidimensional measure, it cannot capture all the dimensions of young people's experience. There is scope for extending the current research to consider other outcomes, such as socio-emotional development as measured by the Strengths and Difficulties Questionnaire (SDQ). Previous research on socio-emotional outcomes of children in Ireland at five years of age has found that migrant children experience fewer socio-emotional difficulties and have higher pro-social scores, although teacher ratings identify lower pro-social scores among this group (see Russell et al., 2016), conflicting findings that highlight the need for further research in this area. Further, the measure of self-concept is collected for the first time at nine years of age, so the analysis could not incorporate change over time, which means that causal conclusions cannot be drawn. Future waves of the GUI will allow this initial study to be further developed.

The role of religious affiliation and religiosity has not been considered in this report, though it is potentially important, both in terms of children's experience of

⁶⁰ See Lunn and Fahey (2011) for an analysis of marital homogamy using the Census.

school in Ireland and their wellbeing more generally (Smyth, et al., 2013). Particularly in the context of the dominant role of the Catholic Church in Irish schools, and the fact that many of those from minority religions come from a migrant background, investigating this topic in greater depth using data from the GUI Cohort '08 could provide useful insights, as could exploring the role of minority religious affiliation in the lives of 17 and 20 year olds using later waves of the GUI. The current study focuses on English language and reading as core elements of the school curriculum and strong predictors of achievement within the Irish school system, as well as a key measure of migrant integration. However, this is only one indicator of cognitive development; as noted in Chapter 5, other non-verbal skills are less likely to differ between migrant and non-migrant groups. Moreover, there is evidence that bilingualism is an advantage in other cognitive outcomes. The GUI provides the opportunity to examine other cognitive outcomes such as the selective attention 'map' test at age nine and the influence of bilingualism and other cognitive/learning outcomes as the children age. Mathematics skills are an important indicator of cognitive skills, as well as being important for many occupational trajectories. Investigating how these skills differ in Cohort '08 could usefully add to what we know from other sources, such as the Programme for International Student Assessment (PISA) study or the national assessments of mathematics administered at primary school (Kavanagh et al., 2016; Kavanagh and Weir, 2018).

The GUI data provide the opportunity to explore many of the issues raised here in greater detail, by following the same group of children over time, allowing researchers to analyse not only the 'direction of travel' in terms of development and integration, but also the impact of past experiences (positive or negative) on children's lives. For example, analysis of racially motivated bullying and experience of discrimination is possible using Cohort '98, which collected data on both children who were born in Ireland and those who migrated as children. A further wave of data will be collected from Cohort '08 when they reach 13 years. This continued collection of data on the progression of those from a migrant background – both people who came to Ireland as children (who make up 11 per cent of Cohort '98), and those born in Ireland to migrant parents – makes for an invaluable resource in terms of examining the experiences of these groups in Irish society and informing the development of policies that promote integration and minimise the risks of segregation and exclusion that have emerged in other countries.

Of course, migration is a dynamic phenomenon: Chapter 1 shows how the scale and composition of immigration flows to Ireland have varied considerably over the past 30 years. The children in the GUI Cohort '08 were born just after a rapid rise in immigration, particularly from eastern Europe, and this study does show parental region of origin and language background play an important role in child outcomes. The GUI study, both the '08 and the earlier '98 cohorts, offers tremendous potential as a resource for investigating the integration of migrantorigin children into Irish society, as children and into young adulthood. Any new child cohort study could also reveal valuable insights into subsequent cohorts of migrant-origin children. In any new cohort study, consideration could usefully be given to surveying children who were born in Ireland to migrant parents, and those who came to Ireland as children.
REFERENCES

- Aboud, F.E. and M. Amato (2001). 'Developmental and socialization influences on intergroup bias', in F. Brown and S.L. Gaertner (eds.), *Blackwell handbook of social psychology, Vol. 4, Intergroup Relations*, Blackwell, Oxford, pp. 65-85.
- Amato, P.R., S. Patterson and B. Beattie (2015). 'Single-parent households and children's educational achievement: A state-level analysis', *Social Science Research*, Vol. 53, pp. 191-202, doi: 10.1016/j.ssresearch.2015.05.012.
- Anderson, B. and S. Blinder (2019). *Who counts as migrant? Definitions and their consequences*, The Migration Observatory, Oxford.
- Anger, S. and G. Heineck (2010). 'Do smart parents raise smart children? The intergenerational transmission of cognitive abilities', *Journal of Population Economics*, Vol. 23, 1105-1132, https://doi.org/10.1007/s00148-009-0298-8.
- Anjeh, D., J. Caputo and S. Armani (2006). 'The implications high stakes, state-mandated academic tests on the educational future of LEP students', Foundation of Instruction in English, Grand Canyon University, Phoenix-Arizona, available at: https://files.eric.ed.gov/fulltext/ED503817.pdf.
- Anti-Racism Committee (2021). Interim report to the Minister for Children, Equality, Disability, Integration and Youth, Department of Children, Equality, Disability, Integration and Youth, Dublin.
- Arnold, S., E. Quinn, S. Groarke, F. McGinnity and C. Durst (2019). Policy and practice targeting the labour market integration of non-EU nationals in Ireland, ESRI/EMN, Dublin, https://doi.org/10.26504/rs89.
- Asher, C. (2011). 'The progressive past: How history can help us serve generation 1.5', *Reference & User Service Quarterly*, Vol. 51, No. 1, pp. 43-48.
- Bauer, P. and R.T. Riphahn (2007). 'Heterogeneity in the intergenerational transmission of educational attainment: Evidence from Switzerland on natives and secondgeneration Immigrants', *Journal of Population Economics*, Vol. 20, No. 1, pp. 121-148.
- Baysu, G., A. Alanya and H.A.G. de Valk (2018). 'School trajectories of the secondgeneration of Turkish immigrants in Sweden, Belgium, Netherlands, Austria and Germany: The role of school systems', *International Journal of Comparative Sociology*, Vol. 59, No. 5-6, 451, https://doi.org/10.1177/0020715218818638.
- Becker, B., O. Klein and N. Biedinger (2013). 'The development of cognitive, language, and cultural skills from age 3 to 6: A comparison between children of Turkish origin and children of native-born German parents and the role of immigrant parents' acculturation to the receiving society', *American Educational Research Journal*, Vol. 50, No. 3, pp. 616-649, https://doi.org/10.3102/0002831213480825.
- Becker, B. (2013). 'The impact of familial activities on the verbal and non-verbal skills of children of Turkish immigrants in Germany', *International Research in Early Childhood Education*, Vol. 4, No. 1, pp. 91-104.
- Beelmann, A., L. Arnold and S. Schultz (2020). 'Buffering negative effects of immigration on cognitive, social, and educational development: A multinational meta-analysis of child and adolescent prevention programmes', *International Journal of Psychology*, Vol. 56, No. 3, pp. 478-490, https://doi.org/10.1002/ijop.12725.

- Belhadj-Kouider E., U. Koglin and F. Petermann (2014). 'Emotional and behavioral problems in migrant children and adolescents in Europe: A systematic review', *European Child & Adolescent Psychiatry*, Vol. 23, No. 6, pp. 373-391, doi: 10.1007/s00787-013-0485-8.
- Belhadj-Kouider, E.B., U. Koglin and F. Petermann (2015). 'Emotional and behavioral problems in migrant children and adolescents in American countries: A systematic review', Journal of Immigrant and Minority Health, Vol. 17, pp. 1240-1258, https://doi.org/10.1007/s10903-014-0039-2.
- Berger, L.M., L. Panico and A. Solaz (2021). 'The impact of center-based childcare attendance on early child development: Evidence from the French Elfe Cohort', *Demography*, Vol. 58, No. 2, pp. 419-450, https://doi.org/10.1215/00703370-8977274.
- Bernstein, B. (1975). Sources of consensus and disaffection in education Class, codes and control, Routledge and Kegan Paul, London.
- Borjas, G.J., S.G. Bronars and S.J. Trejo (1992). 'Self-selection and internal migration in the United States', *Journal of Urban Economics*, Vol. 32, No. 2, pp. 159-185.
- Bottia, M.C. (2019). The relationship between school and housing segregation and immigrants' futures in the US, Poverty and Race Research Action Council, available at: https://www.immigrationresearch.org/system/files/PRRAC-ImmigrantIntegrationandSegregation.pdf.
- Bourdieu, P. (1979). *Distinctions. A social critique of the judgment of taste*, Harvard University Press.
- Bourdieu, P. (1998). Practical reason: On theory of action, Stanford University Press.
- Bradley, R.H., R.F. Corwyn, H.P. McAdoo and C.G. Coll (2001). 'The home environments of children in the United States Part I: Variations by age, ethnicity, and poverty status', *Child Development*, Vol. 72, pp. 1844-1867, https://doi.org/10.1111/1467-8624.t01-1-00382.
- Brunello, G. and L. Rocco (2011). *The effect of immigration on the school performance of natives: Cross country evidence using PISA test scores*, Institute for the Study of Labor Discussion Paper No. 5479, IZA, Bonn.
- Byrne, D., F. McGinnity, E. Smyth and M. Darmody (2010). 'Immigration and school composition in Ireland', *Irish Educational Studies*, Vol. 29, No. 3, pp. 271-288.
- Callan, T., C. Keane and M. Regan (2021). *Microsimulation modelling of the National Childcare Scheme: Updated cost estimates using SWITCH*, Working paper No. 695, Economic and Social Research Institute, Dublin.
- Campbell, R. (1983). 'Status attainment research: End of the beginning or beginning of the end?', *Sociology of Education*, Vol. 56, pp. 47-62.
- Cavallo, M. and G. Russo (2020). *Lost in translation: Reading performance and math performance of second-generation children in Italy*, Working Paper No. 554, Centre for Studies in Economics and Finance, Naples.
- Cebolla-Boado, H. and L.G. Medina (2011). 'The impact of immigrant concentration in Spanish schools: School, class, and composition effects', *European Sociological Review*, Vol. 27, No. 5, pp. 606-623, https://doi.org/10.1093/esr/jcq024.

- Cheah, C.S.L. and C.Y.Y. Leung (2011). 'The social development of immigrant children: A focus on Asian and Hispanic children in the United States', *The Wiley-Blackwell Handbook of Childhood Social Development*, pp. 161-180, https://doi:10.1002/9781444390933.ch9.
- Cheah, C.S.L., C.Y.Y. Leung, M. Tahseen and D. Schultz (2009). 'Authoritative parenting among immigrant Chinese mothers of preschoolers', *Journal of Family Psychology*, Vol. 23, No. 3, pp. 311-320.
- Chao, R.K. (2006). 'The prevalence and consequences of adolescents' language brokering for their immigrant parents', in M.H. Bornstein and L.R. Cote (eds.), *Acculturation and parent–child relationships: Measurement and development*, Erlbaum, Mahwah, New Jersey, pp. 271-296.
- Chaudry, A.G. and M. Zeeshan (2019). 'Migration's impacts on diminishing lingual heritage', *Global Language Review*, Vol. 4, No. I (spring), pp. 60-65, https://doi:10.31703/glr.2019(IV-I).08.
- Chen, F. and H. Ragnarsdóttir (2014). 'Single-parent immigrant families in Iceland: Lives and educational experiences of their children', *Netla Online Journal of Pedagogy and Education*, pp. 1-20.
- Citizensinformation.ie. (2021). 'Admissions policies in primary and secondary schools' [online] available at:

https://www.citizensinformation.ie/en/education/the_irish_education_system/a dmissions_policies_in_primary_and_secondary_schools.html, accessed 14 April.

- Citizensinformation.ie (2021) 'Early start programme' [online], available at https://www.citizensinformation.ie/en/education/pre_school_education_and_ch ildcare/early_start.html, accessed 14 April 2021.
- Conger, R.D., K.J. Conger and M.J. Martin (2010). 'Socioeconomic status, family processes and individual development', *Journal of Marriage and Family*, Vol. 72, No. 3, pp. 685-704.
- Costigan, C.L. and C.M. Koryzma (2011). 'Acculturation and adjustment among immigrant Chinese parents: Mediating role of parenting efficacy', *Journal of Counseling Psychology*, Vol. 58, No. 2, pp. 183-196, https://doi.org/10.1037/a0021696.
- Coughlan, B., E. Doherty, C. O'Neill and B. McGuire (2014). 'Minority status, social welfare status and their association with child participation in sporting, cultural and community activities', *The Economic and Social Review*, Vol. 45, No. 1, pp. 65-85.
- Crosnoe, R. (2007). 'Early childcare and the school readiness of children from Mexican immigrant families', *International Migration Review*, Vol. 41, No. 1, pp. 152-181.
- Crosnoe, R. and R. Turley (2011). 'K–12 educational outcomes of immigrant youth', *The future of children*, Vol. 21, No. 1, pp. 129-152, available at https://files.eric.ed.gov/fulltext/EJ920370.pdf.
- Crosnoe, R. (2013). Preparing the children of immigrants for early academic success, Migration Policy Institute, available at: https://www.migrationpolicy.org/sites/default/files/publications/Crosnoe-FINAL.pdf.
- Crul, M. and H. Vermeulen (2003). 'The second generation in Europe', *International Migration Review*, Vol. 37, pp. 965-986.

- Crul, M. (2007). 'Pathways to success for the second generation in Europe' [online], availabe at https://www.migrationpolicy.org/article/pathways-success-secondgeneration-europe/.
- Crul, M. and J. Schneider (2010). 'Comparative integration context theory: Participation and belonging in new diverse European cities', *Ethnic and Racial Studies*, Vol. 33, pp. 1249-1268.
- Curdt-Christiansen, X.L. (2016). 'Conflicting language ideologies and contradictory language practices in Singaporean multilingual families', *Journal of Multilingual and Multicultural Development*, Vol. 37, No. 7, pp. 694-709, https://doi.org:10.1080/01434632.2015.1127926.
- Darmody, M. and E. Smyth (2018). 'Immigrant student achievement and educational policy in Ireland', in L. Volante, D. Klinger and O. Bilgili (eds.), *Immigrant student achievement and educational policy in Ireland*, Springer, pp. 119-135.
- Darmody, M. and F. McGinnity (2019). 'Immigrant-origin children and the education system', in B. Fanning and L. Michael (eds.), *Immigrants as outsiders in the two Irelands*, Manchester University Press, Manchester.
- Darmody, M. and E. Smyth (2017). 'Out-of-school social activities among immigrant-origin children living in Ireland', *Economic and Social Studies*, Vol. 48, No. 4, pp. 419-439.
- Darmody, M. and E. Smyth (2016). *Entry to programmes of initial teacher education*, Economic and Social Research Institute/The Teaching Council, Dublin.
- Darmody, M., F. McGinnity and G. Kingston (2016). 'The experiences of migrant children in Ireland', in J. Williams, E. Nixon, E. Smyth and D. Watson (eds.), *Cherishing all children equally? Ireland 100 years on from the Easter Rising*, Oak Tree Press, Dublin.
- Darmody, M. and E. Smyth (2015), "When you actually talk to them…" Recognising and respecting cultural and religious diversity in Irish schools', in I. Honohan and N. Rougier (eds.) in *Tolerance in Ireland, north and south*, Manchester University Press, Manchester.
- Darmody, M., D. Byrne and F. McGinnity (2014). 'Cumulative disadvantage? Educational careers of migrant students in Irish secondary schools', *Race Ethnicity and Education*, Vol. 17, No. 1, pp.129-151.
- Darmody, M. (2011). 'Power, education and migration into Ireland', *Power and Education*, *Special issue: Migration and Education*, Vol. 3, No. 3.
- Dennis, J.M., A.L. Fonseca, G. Gutierrez, J. Shen and S. Salazar (2016). 'Bicultural competence and the Latino 2.5 generation: The acculturative advantages and challenges of having one foreign-born and one U.S.-born parent', *Hispanic Journal of Behavioral Sciences*, Vol. 38, No. 3, pp. 341-359.
- Department of Education and Skills (2010). *Intercultural Education Strategy 2010–2015*, Department of Education and Skills, Dublin.
- Department of Education and Skills (2017). *Interesting facts First look at data from POD, 2016/2017*, Department of Education and Skills, Dublin.
- Department of Education and Skills (2019). *Education indicators for Ireland 2019*, Department of Education and Skills, Dublin, available at

https://www.education.ie/en/Publications/Statistics/Key-Statistics/educationindicators-for-ireland.pdf.

- Department of Justice and Equality (2019a). The Migrant Integration Strategy: 2017–2020 progress report to Government, Department of Justice and Equality, Dublin.
- Department of Justice and Equality (2017). The Migrant Integration Strategy A Blueprint for the Future, Department of Justice and Equality, Dublin, available at www.justice.ie/en/JELR/Pages/Migrant_Integration_Strategy_-A Blueprint for the Future.
- Devine, D. (2009). 'Mobilising capitals? Migrant children's negotiation of their everyday lives in the primary school', British Journal of Sociology of Education, Vol. 30, No. 5.
- Devine, D. (2011). 'Securing migrant children's educational well-being: Policy and practice in Irish schools', in D. Devine (ed.), Ethnic minority children and youth in Ireland, Springer, Amsterdam, pp. 73-87.
- Devine, D. (2013). 'Value'ing children differently? Migrant children in education', Children & Society, Vol. 27, pp. 282-294.
- Dimitrova, R., A. Chasiotis and F. van de Vijver (2016). 'Adjustment outcomes of immigrant children and youth in Europe', European Psychologist, Vol. 21, pp. 150-162. https://doi.org/10.1027/1016-9040/a000246.
- Doepke, M. and F. Zilibotti (2015). Parenting with style: Altruism and paternalism in intergenerational preference transmission, working paper, Human Capital and Economic Opportunity Global Working Group, Economics Research Center, University of Chicago.
- Driessen, G. (2017). 'Language choice and language proficiency of immigrant parents and children', Journal of Education Research, Vol. 10, No. 2.
- Dronkers, J. and M. Kalmijn (2013). Single-parenthood among migrant children: Determinants and consequences for educational performance, CReAM discussion CDP No paper series, 09/13, available at: https://www.creammigration.org/publ_uploads/CDP_09_13.pdf.
- Drouhot, L. and V. Nee (2019). 'Assimilation and the second generation in Europe and America: Blending and segregating social dynamics between immigrants and natives', Annual Review of Sociology, Vol. 45, pp. 177-199.
- Duncan, G.J., C.J. Dowsett, A. Claessens, K. Magnuson, A.C. Huston, P. Klebanov, L.S. Pagani, L. Feinstein, M. Engel, J. Brooks-Gunn, H. Sexton, K. Duckworth and C. Japel (2007). 'School readiness and later achievement', *Developmental Psychology*, Vol. 43, No. 6, pp. 1428-1446, https://doi.org/10.1037/0012-1649.43.6.1428.
- Dustmann, C., T. Frattini and G. Lanzara (2011). Educational achievement of second generation immigrants: An international comparison, Discussion Paper No. 2011-25, Norface Migration, available at:

https://www.dagliano.unimi.it/media/WP2011 314.pdf.

Dunn, J. (2004). Children's friendships: The beginnings of intimacy, Blackwell, Malden, MA.

- Eckstein, S. (2002). 'On deconstructing and reconstructing the meaning of immigrant generations', in P. Levitt and M.C. Waters (eds.), *The changing face of home: The transnational lives of the second generation*, Sage, New York, pp. 211-215.
- Erkut, S. and A. Tracy (2002). 'Predicting adolescent self-esteem from participation in school sports among Latino subgroups', *Hispanic Journal of Behaviour Sciences*, Vol. 24, No. 4, pp. 409-429, available at www.ncbi.nlm.nih.gov/pmc/articles/PMC3048356.
- Eurostat (2014). First and second-generation immigrants A statistical overview, available at https://ec.europa.eu/eurostat/statisticsexplained/index.php/First_and_second-generation_immigrants_-_a_statistical_overview.
- Evans, G.W. (2004). 'The environment of childhood poverty', *American Psychologist*, Vol. 59, No. 2, pp. 77-92, https://doi.org/10.1037/0003-066X.59.2.77.
- Faas, D., M. Darmody and H. Foley (2019). 'Post-school transitions in Ireland: A case study of Russian-speaking students', *Comparative Education*, Vol. 55, No. 3, pp. 326-346, https://doi.org/10.1080/03050068.2019.1619329.
- Fahey, É., H. Russell, F. McGinnity and R. Grotti (2019). *Diverse neighbourhoods: An analysis* of the residential distribution of immigrants in Ireland, ESRI/Department of Justice and Equality, Dublin.
- Felfe C., M. Lechner and A. Steinmayr (2016). 'Sports and child development', *PLoS ONE*, Vol. 11, No. 5, https://doi.org/10.1371/journal.pone.0151729.
- Feliciano, C. and Y.R. Lanuza (2015). 'The immigrant advantage in adolescent educational expectations', *International Migration Review*, Vol. 50, No. 3, pp. 758-792, doi:10.1111/imre.12183.
- Feliciano, C. and Y.R. Lanuza (2017). 'An immigrant paradox? Contextual attainment and intergenerational educational mobility', *American Sociological Review*, Vol. 82, No. 1, pp. 211-241, doi:10.1177/0003122416684777.
- Feliciano, C. (2020). 'Immigrant selectivity effects on health, labor market and educational outcomes', Annual Review of Sociology, Vol. 46, pp. 315-334, https://doi.org/10.1146/annurev-soc-121919-054639.
- Fernández-Reino, M. (2016). 'Immigrant optimism or anticipated discrimination? Explaining the first educational transition of ethnic minorities in England', *Research in Social Stratification and Mobility*, Vol. 46, Part B, pp. 141-156.
- Fibbi, R., H.A. Midtbøen and P. Simon (2021). *Migration and discrimination: IMISCOE short reader*, Springer Nature.
- Flisi, S., E.C. Meroni and E. Vera-Toscano (2016). Educational outcomes and immigrant background, Publications Office of the European Studies, Luxembourg, https://doi.org/10.2791/026577.
- Ford, R., T. King, N. Priest and A. Kavanagh (2017). 'Bullying and mental health and suicidal behaviour among 14- to 15-year-olds in a representative sample of Australian children', Australian & New Zealand Journal of Psychiatry, Vol. 51, No. 9, pp. 897-908.

- Fortuny, K., D.J. Hernandez and A. Chaudry (2010). 'Young children of immigrants: The leading rdge of America's future', Children of Immigrants Research Brief 3, The Urban Institute, Washington, DC.
- Frese, C., A. Röder and M. Ward (2015). *New Irish families: Successes and challenges in heritage language acquisition for second generation migrant children*, Trinity College Dublin, Dublin.
- Gambaro, L., K. Stewart and J. Waldfogel (2014). 'Introduction' in L. Gambaro, K. Stewart and J. Waldfogel (eds.), *An equal start? Providing quality early education and care for disadvantaged children*, Policy Press, Bristol.
- García-Hermoso, A., I. Hormazábal-Aguayo, O. Fernández-Vergara, N. González-Calderón, J. Russell-Guzmán, F. Vicencio-Rojas, C. Chacana-Cañas and R. Ramírez-Vélez (2019). 'A before-school physical activity intervention to improve cognitive parameters in children: The Active-Start study', *Scandinavian Journal of Medicine and Science in Sports*, Vol. 30, No. 1, pp. 108-116.
- Geay, C., S. McNally and S. Telhaj (2012), 'Non-native speakers of English in the classroom: What are the effects on pupil performance?', Centre for the Economics of Education Discussion Paper No 137, CEE, London.
- Gelbach, J.B. (2016). 'When do covariates matter, and which ones, and when?', *Journal of Labor Economics*, Vol. 34, No. 2.
- Gibson, M.A. and J.U. Ogbu (1991). *Minority status and schooling: A comparative study of immigrant and involuntary minorities*, Routledge, New York, London.
- Gilmartin, M. and J. Dagg (2020). 'Spatializing immigrant integration outcomes', *Population, Space and Place*, Vol. 27, No. 2.
- Gsir, S. (2014). 'Social interactions between immigrants and host country populations: A country-of-origin perspective', Migration Policy Centre, INTERACT 2014/02 Position Paper, retrieved from Cadmus, European University Institute Research Repository, available at http://hdl.handle.net/1814/31243.
- Harte, E., F. Herrera and M. Stepanek (2016). Education of EU migrant children in EU Member States, RAND Corporation, Santa Monica and Cambridge, available at https://www.rand.org/content/dam/rand/pubs/research_reports/RR1700/RR171 5/RAND_RR1715.pdf.
- Heath, A. and Y. Brinbaum (2007). 'Explaining ethnic inequalities in educational attainment', *Ethnicities*, Vol. 7, No. 3, pp. 291-305.
- Hertting, K. and I. Karlefors (2013). 'Sport as a context for integration: Newly arrived immigrant children in Sweden drawing sporting experiences', *International Journal of Humanities and Social Science*, Vol. 3, No. 18, pp. 35-44.
- Hoff, E., C. Core, S. Place, R. Rumiche, M. Señor and M. Parra. 2012. 'Dual language exposure and early bilingual development', *Journal of Child Language*, Vol. 39, No. 1, pp. 1-27, https://doi.org/10.1017/S0305000910000759.
- Hoffmann, N. (2021). 'A "win-win exercise"? Eastern European children in western Europe', Department of Sociology, UCLA, Los Angeles.
- Hollebeke, I., E. Struys and O. Agirdag (2020). 'Can family language policy predict linguistic, socio-emotional and cognitive child and family outcomes? A systematic

review', *Journal of Multilingual and Multicultural Development*, https://doi.org/10.1080/01434632.2020.1858302.

- Holmes, R.M., B. Gardner, K. Kohm, C. Bant, A. Ciminello, K. Moedt and L. Romeo (2017). 'The relationship between young children's language abilities, creativity, play, and storytelling', *Early Child Development and Care*, Vol. 189, No. 2, https://doi.org/10.1080/03004430.2017.1314274.
- Honohan, I. and N. Rougiern (2016). *Tolerance and diversity in Ireland, north and south,* Manchester University Press, Manchester.
- Huynh, Q.-L., T. Devos and L. Smalarz (2011). 'Perpetual foreigner in one's own land: Potential implications for identity and psychological adjustment', *Journal of Social* and Clinical Psychology, Vol. 30, pp. 133-162, https://doi.org/10.1521/jscp.2011.30.2.133.
- Ismail, A.A. (2019). 'Immigrant children, educational performance and public policy: A capability approach', Journal of International Migration and Integration, Vol. 20, pp. 717-734, https://doi.org/10.1007/s12134-018-0630-9.
- Isphording, I.E., M. Piopiunik and N. Rodriguez-Planas (2016). 'Speaking in numbers: The effect of reading performance on math performance among immigrants', *Economics Letters*, Vol. 139, pp. 52-56.
- Isphording, I. (2014). 'Disadvantages of linguistic origin Evidence from immigrant literacy scores', *Economics Letters*, Vol. 123, No. 2, pp. 236-239.
- Izzo, C., L. Weiss, T. Shanahan and F. Rodriguez-Brown (2000). 'Parental self-efficacy and social support as predictors of parenting practices and children's socioemotional adjustment in Mexican immigrant families', *Journal of Prevention & Intervention in the Community*, Vol. 20, No. 1-2, pp. 197-213.
- Jensen, P. and A. Wurtz-Rasmussen (2011), 'The effect of immigrant concentration in schools on native and immigrant children's reading and math skills', *Economics of Education Review*, Vol. 30, No. 6, pp. 1503-1515.
- Johnson De Feyter, J. and A. Winsler (2009). 'The early developmental competencies and school readiness of low-income, immigrant children: Influences of generation, race/ethnicity, and national origins', *Early Childhood Research Quarterly*, Vol. 24, No. 4, pp. 411-431, https://doi.org/10.1016/j.ecresq.2009.07.004.
- Jonsson, J., F. Kalter and F. Van Tubergen, F. (2018). *Growing up in diverse societies: The integration of the children of immigrants in England, Germany, the Netherlands and Sweden*, Oxford University Press, Oxford.
- Jonsson, J. and K. Mood (2018). 'Mental well-being in boys and girls of immigrant background: The balance between vulnerability and resilience', in J. Jonsson, F. Kalter and F. Van Tubergen (eds.), *Growing up in diverse societies: The integration* of the children of immigrants in England, Germany, the Netherlands and Sweden, Oxford University Press, Oxford.
- Kalmijn, M. (2018) 'Family structure and father absence among immigrant children: The role of migration, religion and inequality', in Jonsson, J., F. Kalter and F. Van Tubergen (eds.), Growing up in diverse societies: The integration of the children of immigrants in England, Germany, the Netherlands and Sweden, Oxford University Press, Oxford.

- Karoly, L.A. and G.C. Gonzalez (2011). 'Early care and education for children in immigrant families', *Future of Children*, Vol. 21, No. 1, pp. 71-101.
- Kavanagh, L. and S. Weir (2018). The evaluation of DEIS: The lives and learning of urban primary school pupils, 2007–2016, Educational Research Council, Dublin, available at: https://www.erc.ie/wp-content/uploads/2018/09/DEIS_2018_Online.pdf.
- Kavanagh, L., G. Sheil, L. Gilleece and J. Kiniry (2016). *The 2014 national assessments of English reading and mathematics, Volume II: Context report,* ERC, Dublin.
- Kelly, E., S. McGuinness, P. O'Connell, A. González Pandiella and D. Haugh (2016). How did immigrants fare in the Irish labour market over the Great Recession?, OECD Economics Department Working Papers No. 1284, OECD Publishing, Paris.
- King, R., M. Thomson, T. Fielding and T. Warnes (2006). 'Time, generations and gender in migration and settlement', in R. Penninx, M. Berger and M. Kraal (eds.), *The dynamics of international migration and settlement in Europe*, Amsterdam University Press, Amsterdam, pp. 233-268.
- Kingdon, G. and R. Cassen (2010). 'Ethnicity and low achievement in English schools', *British Educational Research Journal*, Vol. 36, No. 3, pp. 403-431.
- Kitching, K. (2011). 'Interrogating the changing inequalities constituting 'popular', 'deviant' and 'ordinary' subjects of school/subculture in Ireland: New migrant student recognition, resistance and recuperation', *Race Ethnicity and Education*, Vol. 14, No. 3, pp. 293-311.
- Kirpitchenko, L. and F. Mansouri (2014). 'Social engagement among migrant youth: Attitudes and meanings', *Social Inclusion*, Vol. 2, No. 2, pp. 17-27.
- Krause, A., U. Rinne and S. Schuller (2011). Kick it like Ozil? Second generation migrants' initial disadvantage and the failure of the education system, IZA Working Paper series, available at http://conference.iza.org/conference_files/amm2011/krause_a5647.pdf.
- Kristen, C., P. Mühlau and D. Schacht (2016). 'Language acquisition of recently arrived immigrants in England, Germany, Ireland and the Netherlands', *Ethnicities*, Vol. 16, No. 2, pp. 180-212.
- Kulic, N., J. Skopek, M. Triventi and H.-P. Blossfeld (2019). 'Social background and children's cognitive skills: The role of early childhood education and care in a cross-national perspective', Annual Review of Sociology, Vol. 45, No. 1, pp. 557-579, https://doi.org/10.1146/annurev-soc-073018-022401.
- Lareau, A. and E.M. Horvat (1999). 'Moments of social inclusion and exclusion: Race, class, and cultural capital in family–school relationships', *Sociology of Education*, Vol. 72, pp. 37-53.
- Leseman, P., H. Mulder, J. Verhagen, M.L. Broekhuizen, S.D.M. van Schaik and P.L. Slot (2017). 'Effectiveness of Dutch targeted preschool education policy for disadvantaged children', in H.-P. Blossfeld, N. Kulic, J. Skopek and M. Triventi (eds.), *Childcare, early education, and social inequality – An international perspective,* Edward Elgar, Cheltenham, pp. 173-193.

- Levels, M. and J. Dronkers (2008). 'Educational performance of native and immigrant children from various countries of origin', *Ethnic and Racial Studies*, Vol. 31, No. 8, pp. 1404-1425.
- Lewis, V., J. Boucher, L. Lupton and S. Watson (2000). 'Relationships between symbolic play, functional play, verbal and non-verbal ability in young children', *International Journal of Language & Communication Disorders*, Vol. 35, No. 1, pp. 117-127, https://doi.org/10.1080/136828200247287.
- Liebig, T. and G. Spielvogel (2021). 'Residential segregation of immigrants: Patterns, drivers, effects and policy responses', Chapter 5, *OECD International Migration Outlook* 2021; OECD: Paris.
- Liu, J. (2013). Integration of ethnic Chinese children into Irish primary schools, unpublished PhD thesis, DCU, available at http://doras.dcu.ie/18076/1/Vol_1_for_library.pdf.
- Lunn, P. and T. Fahey (2011). *Households and family structures in Ireland: A detailed statistical analysis of Census 2006*, Economic and Social Research Institute and the Family Support Agency, Dublin.
- Luthra, R. and T. Soehl (2015). 'From parent to child? Transmission of educational attainment within immigrant families: Methodological considerations', *Demography*, Vol. 52, pp. 543-567.
- Luthr, R.R. and R. Waldinger (2013). 'Intergenerational mobility among immigrants and their descendants', in D. Card and S. Raphael (eds.), *Immigration, poverty, and socioeconomic inequality*, Russell Sage, New York, pp. 169-205.
- Lüdemann. E. and G. Schwerdt (2010). *Migration background and educational tracking: Is there a double disadvantage for second-generation immigrants?* CESIFO Working Paper No. 3256.
- Magnuson, K.A., C. Lahaie and J. Waldfogel (2006). 'Preschool and school readiness of children of immigrants', *Social Science Quarterly*, Vol. 87, No. 5, pp. 1241-1262.
- Maître, B., H. Russell and E. Smyth (2021). *The dynamics of child poverty in Ireland: Evidence from the Growing Up in Ireland survey*, ESRI Research Series 121, Economic and Social Research Institute, Dublin.
- Makarova, V., N. Terekhova, and A. Mousavi (2019). 'Children's language exposure and parental language attitudes in Russian-as-a-heritage-language acquisition by bilingual and multilingual children in Canada', *International Journal of Bilingualism*, Vol. 23, No. 2, pp. 457-485, https://doi.org/10.1177/1367006917740058.
- Masten, A.S., K. Liebkind and D.J. Hernandez (eds.) (2012). *Realizing the potential of immigrant youth*, Cambridge University Press, Cambridge.
- McCoy, S., A. Quail and E. Smyth (2014). 'The effects of school social mix: Unpacking the differences', *Irish Educational Studies*, Vol. 33, No. 3, pp. 307-330, https://doi.org/10.1080/03323315.2014.955746.
- McGinnity, F., I. Privalko, H. Russell, S. Curristan, A. Stapleton and J. Laurence (forthcoming 2022). *Origin and integration: Housing and family among migrants in the 2016 Irish Census*, ESRI Research Series, Economic and Social Research Institute, Dublin.

- McGinnity, F., S. Enright, E. Quinn, B. Maître, I. Privalko, M. Darmody and M. Polakowski (2020a). *Monitoring report on integration 2020*, Economic and Social Research Institute/Department of Children, Equality, Disability, Integration and Youth, Dublin.
- McGinnity, F., I. Privalko, E. Fahey, D. O'Brien and S. Enright (2020b). *Origin and integration: A study of migrants in the 2016 Irish Census,* Economic and Social Research Institute/Department of Justice and Equality, Dublin.
- McGinnity, F. and M. Darmody (2019). 'Immigrant-origin children and the education system' in B. Fanning and L. Michael (eds.), *Immigrants as outsiders in the two Irelands*, Manchester University Press, Manchester.
- McGinnity, F., H. Russell and A. Murray (2015a). *Non-parental childcare and child cognitive outcomes at age 5*, The Stationery Office, Dublin.
- McGinnity, F., M. Darmody and A. Murray (2015b). *Academic achievement among immigrant children in Irish primary schools*, Economic and Social Research Institute, Dublin.
- McGinnity, F., E. Quinn, G. Kingston and P.J. O'Connell (2014). *Annual monitoring report on integration 2013*, Economic and Social Research Institute/Integration Centre, Dublin.
- McGinnity, F., A. Murray, S. McNally (2013). *Mothers' return to work and childcare choices* for infants in Ireland, Growing Up in Ireland Report 2, The Stationery Office, Dublin.
- McGinnity, F. E. Quinn, G. Kingston and P.J. O'Connell (2012). *Annual monitoring report on integration 2011,* Economic and Social Research Institute/Integration Centre, Dublin.
- McKown, C. and R.S. Weinstein (2003). 'The development and consequences of stereotype consciousness in middle childhood', *Child Development*, Vol. 74, No. 2, pp. 498-515.
- McNally, S. M. Darmody and J. Quigley (2019). 'The socio-emotional development of language-minority children entering primary school in Ireland', *Irish Educational Studies*, Vol. 38, No. 4, pp. 519-534, https://doi.org/10.1080/03323315.2019.1663550.
- Meade, S. and M. O'Connell (2009). 'Complex and contradictory accounts: The social representations of immigrants and ethnic minorities held by Irish teenagers', *Translocations: Migration and Social Change*, Vol. 4, No. 1, pp. 51-66.
- Melkonian, M., S. Areepattamannil, L. Menano and P. Fildago (2019). 'Examining acculturation orientations and perceived cultural distance among immigrant adolescents in Portugal: Links to performance in reading, mathematics, and science', Social Psychology of Education, Vol. 22, pp. 969-989, https://doi.org/10.1007/s11218-019-09506-5.
- Menipaz, E. and A. Menipaz (2011). *International business: Theory and practice*, SAGE, London.
- Minello, A. and N. Barban (2012). 'The educational expectations of children of immigrants in Italy', *The ANNALS of the American Academy of Political and Social Science*, Vol. 643, pp. 78-103.

- Molcho, M., T. Bjarnason, F. Cristini, M. Gaspar de Matos, T. Koller, C. Moreno, S. Nic Gabhainn and M. Santinello (2006). *Foreign-born children in Europe: An overview from the health behaviour in school-aged children*, International Organization for Migration.
- Murray, A., F. McGinnity and H. Russell (2016). 'Inequalities in access to early care and education in Ireland', in J. Williams, E. Nixon, E. Smyth, E. and D. Watson (eds.), *Cherishing all the children equally? Ireland 100 years on from the Easter Rising*, Oaktree Press, Dublin.
- NCES (2015). School composition and the Black–White achievement gap, available at https://nces.ed.gov/nationsreportcard/subject/studies/pdf/school_composition_ and_the_bw_achievement_gap_2015.pdf.
- OECD (2021). Young people with migrant parents, OECD Publishing, Paris.
- OECD (2018a). Settling in. Indicators of immigrant integration, OECD Publishing, Paris.
- OECD (2018b). *The resilience of students with an immigrant background*, OECD Publishing, Paris.
- OECD (2015). Helping immigrant students to succeed at school and beyond, OECD Publishing, Paris.
- OECD (2017). Catching up? Intergenerational mobility and children of immigrants, OECD Publishing, Paris, http://dx.doi.org/10.1787/9789264288041-en.
- OECD (2018). PISA 2018 results (Volume III): What school life means for students' lives, available at https://www.oecd-ilibrary.org/sites/d69dc209en/index.html?itemId=/content/component/d69dc209-en.
- Ogbu, J. (1983). 'Minority generation and schooling in plural societies', *Comparative Education Review*, Vol. 27, pp. 168-190.
- Ohinata, A. and J.C. VanOurs (2012). *How immigrant children affect the academic achievement of native Dutch children*, Norface Migration Discussion Paper No. 2012-12, available at https://www.norface-migration.org/publ_uploads/NDP_12_12.pdf.
- Oppedal, B., S. Keles, C. Cheah and C. Røysamb (2020). 'Culture competence and mental health across different immigrant and refugee groups', *BMC Public Health*, Vol. 20, No. 292, https://doi.org/10.1186/s12889-020-8398-1.
- Paradies, Y., J. Ben, N. Denson, A. Elias, N. Priest, A. Pieterse, A. Gupta, M. Kelaher and G. Gee (2015). 'Racism as a determinant of health: A systematic review and metaanalysis', *PLoS One*, Vol. 10, No. 9, https://doi.org/10.1371/journal.pone.0138511.
- Park-Taylor, J., V. Ng, A.B. Ventura, A.E. Kang, C.R. Morris, T. Gilbert and R.A. Androsiglio (2007). 'What it means to be and feel like a "true" American: Perceptions and experiences of second-generation Americans', *Cultural Diversity and Ethnic Minority Psychology*, Vol. 14, pp. 128-137, https://doi.org/10.1037/1099-9809.14.2.128.
- Piers, E.V. and D.S. Herzberg (2002). *Piers-Harris 2: Piers-Harris Children's Self-Concept Scale* (2nd ed.), Western Psychological Services, Los Angeles.

- Pong S., J. Dronkers and G. Hampden-Thompson (2003). 'Family policies and children's school achievement in single- versus two-parent families', *Journal of Marriage and Family*, Vol. 65, pp. 681-699.
- Poulin-Dubois, D., E. Bialystok, A. Blaye, A. Polonia and J. Yott (2012). 'Lexical access and vocabulary development in very young bilinguals', *International Journal of Bilingualism*, Vol. 17, No. 1, pp. 57-70, https://doi.org/10.1177/1367006911431198.
- Portes, A. and M. Zhou (1993). 'The new second generation: Segmented assimilation and its variants', *Annals of the American Academy of Political and Social Science*, Vol. 530, pp. 74-97.
- Portes, A., R. Aparicio and W. Haller (2016). *Theories of second-generation adaptation*, California Scholarship Online, https://doi.org/10.1525/california/9780520286290.001.0001.
- Priest, N., Kavanagh, A., Bécares, L., King, T. (2019). 'Cumulative effects of bullying and racial discrimination on adolescent health in Australia', *Journal of Health and Social Behavior*, Vol. 60, No. 3, pp. 344-361.
- Rivas-Drake, D., D. Hughes and N. Way (2009). 'A preliminary analysis of associations among ethnic racial socialization, ethnic discrimination, and ethnic identity among urban sixth graders', *Journal of Research on Adolescence*, Vol. 19, No. 3, pp. 558-584.
- Rooth, D.-O. and J. Saarela (2007). 'Selection in migration and return migration: Evidence from micro data', *Economics Letters*, Vol. 94, No. 1, pp. 90-95, https://doi.org/10.1016/j.econlet.2006.08.006.
- Röder, A., M. Ward and C. Frese (2017). 'From labour migrant to stay-at-home mother? Childcare and return to work among migrant mothers from the EU accession countries in Ireland', *Work, Employment and Society*, Vol. 32, No. 5, pp. 1-18.
- Röder, A., M. Ward, C. Frese and E. Sánchez (2014). *New Irish families: A profile of second-generation children and their families*, Trinity College Dublin, Dublin.
- Rodríguez-Izquierdo, R.M. and M. Darmody (2019). 'Policy and practice in language support for newly arrived migrant children in Ireland and Spain', *British Journal of Educational Studies*, Vol. 67, No. 1, pp. 41-57, https://doi.org/10.1080/00071005.2017.1417973.
- Rumbaut, R.G. (2004). 'Ages, life stages, and generational cohorts: Decomposing the immigrant first and second generations in the United States', *The International Migration Review*, Vol. 38, No. 3, pp. 1160-1205.
- Rumbaut, R.G. (1994). 'The crucible within: Ethnic identity, self-esteem, and the segmented assimilation among children of immigrants', *International Migration Review*, Vol. 28, pp. 748-794.
- Rumbaut, R.G. (2005). 'Children of immigrants and their achievement: The roles of family, acculturation, social class, gender, ethnicity and school context', in R.D. Taylor (ed.), *Addressing the achievement gap: Theory informing practice*, Information Age Publishing, pp. 23-59.

- Russell, H., F. McGinnity, É. Fahey and O. Kenny (2018). *Maternal employment and the cost of childcare in Ireland*, Research Series No. 73, Economic and Social Research Institute, Dublin.
- Russell, H., O. Kenny and F. McGinnity (2016). *Childcare, early education and socioemotional outcomes at age 5: Evidence from the Growing Up in Ireland study,* Economic and Social Research Institute, Dublin.
- Sam, D. (2006). 'Adaptation of children and adolescents with immigrant background: Acculturation or development?', in M.H. Bornstein and L.R. Cote (eds.), Acculturation and parent-child relationships: Measurement and development, Erlba, Mahwah, New Jersey, pp. 97-111.
- Schleicher, A. and J. McLaughlin (2018). *Why social emotional learning matters for migrant students and how schools can help*, OECD, Paris.
- Schneeweis, N. (2015). Immigrant concentration in schools: Consequences for native and migrant students, IZA DP No. 7230, available at http://ftp.iza.org/dp7230.pdf.
- Schurer, S. (2008). Labour market outcomes of second-generation immigrants: How heterogeneous are they really? Ruhr Economic Papers, Department of Economics, Ruhr-Universität Bochum (RUB).
- Schüller, S. (2015). 'Parental ethnic identity and educational attainment of secondgeneration immigrants', *Journal of Population Economics*, Vol. 28, No. 4, 965-1004, https://doi.org/10.1007/s00148-015-0559-7.
- Sharp, C. (2002). 'School starting age: European policy and recent research', paper presented at the LGA seminar, 'When should our children start school?', London, 1 November 2002..
- Shin, J., S. Bayram Özdemir, J. Lee and C.S.L. Cheah (2010). 'The parenting practices and child outcomes of Korean immigrant mothers with different acculturative strategies', *Korean Journal of Child Studies*, Vol. 31, pp. 1-17.
- Sierens, S. and P. van Avermaet (2015). *Inequality, inequity and language in education: There are no simple recipes*! Background paper for the 6th Meeting of the Transatlantic Forum on Inclusive Early Years, July 8-10, 2015, Washington, DC.
- Sirin, S. and P. Ryce and M. Mir (2009). 'How teachers' values affect their evaluation of children of immigrants: Findings from Islamic and public schools', *Early Childhood Research Quarterly*, Vol. 24, No. 4, pp. 463-473.
- Smyth, E. (2016). 'Inequalities from the start? Children's integration into primary school', in J. Williams, E. Nixon, E. Smyth and D. Watson (eds.), *Cherishing all the children* equally? Ireland 100 years on from the Easter Rising, Oaktree Press, Dublin.
- Smyth, E., M. Darmody, F. McGinnity and D. Byrne (2009). *Adapting to diversity: Irish schools and newcomer students*, ESRI/Department of Education, Dublin.
- Smyth, E., S. McCoy and G. Kingston (2015). *Learning from the evaluation of DEIS*, Economic and Social Research Institute, Dublin.
- Smyth, E., M. Lyons and M. Darmody (2013). *Religious education in a multicultural Europe: Children, parents and schools*, Palgrave Macmillan.

- Smyth, E. (2015). Wellbeing and school experiences among 9- and 13-year-olds: Insights from the Growing Up in Ireland study, Economic and Social Research Institute, Dublin.
- Smyth, E (2018). The transition to primary education: Insights from the Growing Up in Ireland study, National Council for Curriculum and Assessment, Economic and Social Research Institute, Dublin.
- Sørensen, R., H. Bonesrønning and J. Iversen (2016). *Culture and school performance: Evidence from second generation immigrants to Norway*. Working Paper.
- Spieker, B., J. Steutel and D. Ruyter (2004). 'Self-concept and social integration: The Dutch case as an example', *Theory and Research in Education*, Vol. 2, No. 2, https://doi.org/10.1177/1477878504043442.
- Sprong, J. and Skopek, J. (2021). *Examining the 'migrant gap' at school starting age: The case of Ireland*, Trinity College Dublin, Dublin.
- Stevens, C., B. Lauinger and H. Neville (2009). 'Differences in the neural mechanisms of selective attention in children from different socioeconomic backgrounds: An event-related brain potential study', *Developmental Science*, Vol. 12, No. 4, pp. 634-646.
- Struys, E., G. Mohades, P. Bosch and M. van den Noort (2015). 'Cognitive control in bilingual children: Disentangling the effects of second-language proficiency and onset age of acquisition', *Swiss Journal of Psychology*, Vol. 74, No. 2, pp. 65-73, https://doi.org/10.1024/1421-0185/a000152.
- Suárez-Orozco, M., M.M. Abo-Zena and A.K. Marks (2015). *Transitions: The development* of children of immigrants, New York University Press, New York.
- Suárez-Orozco, M. (2015). *Children of immigration*, Paper prepared for the policy forum of the Strength through Diversity project – Social emotional learning to foster a sense of belonging for immigrant and refugee learners, available at https://www.oecd.org/education/school/Children-of%20-Immigration.pdf.
- Suárez-Orozco, C., J. Rhodes and M. Milburn (2009). 'Unraveling the immigrant paradox: Academic engagement and disengagement among recently arrived immigrant youth', *Youth and Society*, Vol. 41, pp. 151-185.
- Thomson, M. and M. Crul (2007). 'The second generation in Europe and the United States: How is the transatlantic debate relevant for further research on the European second generation?', *Journal of Ethnic and Migration Studies*, Vol. 33, pp. 1025-1041.
- Tormey, R. and J. Gleeson (2012). 'Irish post-primary students' attitudes towards ethnic minorities', *Irish Educational Studies*, Vol. 31, No. 2, pp. 157-173.
- Triandis, H.C. (1998). 'Introduction to diversity in clinical psychology', in A.S. Bellack and M. Hersen (eds.), *Comprehensive Clinical Psychology*, Pergamon, Oxford, pp. 1-33.
- Turney, K. and G. Kao (2009). 'Barriers to school involvement: Are immigrant parents disadvantaged?', *The Journal of Educational Research*, Vol. 102, No. 4, pp. 257-271.
- Van Driel, B., M. Darmody, and J. Kerzil (2016). 'Education policies and practices to foster tolerance, respect for diversity and civic responsibility in children and young people

in the EU', *NESET II report*, Publications Office of the European Union, Luxembourg, https://doi.org/10.2766/46172.

- Vathi, Z. (2015). *Migrating and settling in a mobile world. Albanian migrants and their children in Europe*, Springer.
- Vitoroulis, I. and K. Georgiades (2017). 'Bullying among immigrant and non-immigrant early adolescents: School- and student-level effects', *Journal of Adolescence*, Vol. 61, pp. 141-151, https://doi.org/10.1016/j.adolescence.2017.10.008.
- Volante, L., D. Klinger and O. Bilgili (2018). *Immigrant student achievement and education policy. Cross-cultural approaches*, Springer.
- Waldinger, R. and J. Perlmann (1998). 'Second generations: Past, present, future', *Journal* of Ethnic and Migration Studies, Vol. 24, pp. 5-24.
- Wästlund, E., T. Norlander and T. Archer (2001). 'Exploring cross-cultural differences in selfconcept: A meta-analysis of the self-description questionnaire-1', *Cross-cultural Research*, Vol. 35, No. 3, pp. 280-302.
- Waters, M.C. (1990). *Ethnic options: Choosing identities in America*, University of California Press, Los Angeles.
- Watson, D., B. Maître, R. Grotti and C.T. Whelan (2018). *Poverty dynamics of social risk groups in the EU: An analysis of the EU Statistics on Income and Living Conditions, 2005 to 2014*, ESRI, Dublin.
- Weine, S.M., N. Ware and A. Klebic (2004). 'Converting cultural capital among teen regugees and their families from Bosnia-Herzegovina', *Psychiatric Services*, Vol. 55, pp. 923-927.
- Weir, S. and L. Kavanagh (2018). *The evaluation of DEIS at post-primary level: Closing the achievement and attainment gaps*, ERC, Dublin, available at: https://www.erc.ie/2019/01/07/new-report-on-the-evaluation-of-deis-at-postprimary-level/.
- White, J. (2018). *Children's social circumstances and educational outcomes*, NHS Health Scotland, Edinburgh.
- Williams, J., E. Nixon, E. Smyth and D. Watson (2016). *Cherishing all the children equally?* Oaktree Press, Dublin.
- Wolke, D. and S.T. Lereya (2015). 'Long-term effects of bullying', *Archives of Disease in Childhood*, Vol. 100, No. 9, pp. 879-885.
- Zhou, M. (1997). 'Segmented assimilation: Issues, controversies, and recent research on the new second generation', *International Migration Review*, Vol. 31, pp. 975-1008.
- Zhou, Z., S.T. Peverly, T. Xin, A.S. Huang and W. Wang (2003). 'School adjustment of first generation Chinese–American adolescents', *Psychology in the Schools*, Vol. 40, pp. 71-84.
- Zuccotti, C.V., H.B.G. Ganzeboom and A. Guveli (2017). 'Has migration been beneficial for migrants and their children?' *International Migration Review*, Vol. 51, No. 1, pp. 97-12

APPENDIX 1



FIGURE A1 PROPORTION OF MIGRANT CHILDREN IN SCHOOL AT 9 YEARS BY PARENTAL COUNTRY OF BIRTH

Note: GUI Wave 5, weighted by cross sectional weight.

Whitaker Square, Sir John Rogerson's Quay, Dublin 2 Telephone **+353 1 863 2000** Email **admin@esri.ie** Web **www.esri.ie** Twitter **@ESRIDublin**

